

Technological University Dublin ARROW@TU Dublin

Other Resources

School of Surveying and Construction Management (Former DIT)

2011-1

Strategic Planning in Irish Quantity Surveying Pracitces

Roisin Murphy Technological University Dublin, roisin.murphy@tudublin.ie

Follow this and additional works at: https://arrow.tudublin.ie/beschreoth

Part of the Business Commons

Recommended Citation

Murphy, R.: Strategic Planning in Irish Quantity Surveying Pracitces. Doctoral Thesis. Edinburgh Business School, Heriot Watt University, 2011.

This Theses, Ph.D is brought to you for free and open access by the School of Surveying and Construction Management (Former DIT) at ARROW@TU Dublin. It has been accepted for inclusion in Other Resources by an authorized administrator of ARROW@TU Dublin. For more information, please contact arrow.admin@tudublin.ie, aisling.coyne@tudublin.ie, vera.kilshaw@tudublin.ie.



Dublin Institute of Technology ARROW@DIT

Other Resources

School of Real Estate and Construction Economics

2011-01-01

Strategic Planning in Irish Quantity Surveying Pracitces

Roisin Murphy

Follow this and additional works at: http://arrow.dit.ie/beschreoth

Part of the <u>Business Commons</u>

This Theses, Ph.D is brought to you for free and open access by the School of Real Estate and Construction Economics at ARROW@DIT. It has been accepted for inclusion in Other Resources by an authorized administrator of ARROW@DIT. For more information, please contact yvonne.desmond@dit.ie, arrow.admin@dit.ie.



Strategic Planning in Irish Quantity Surveying Practices

Róisín Murphy BA(Hons); PDES; MBS; MSc.

Submitted for the Degree of Doctor of Business Administration Heriot-Watt University Edinburgh Business School

JANUARY 2011

The copyright of this thesis is owned by the author. Any quotation from the thesis or use of any of the information contained in it must acknowledge this thesis as the source of the quotation or information.

ABSTRACT

The role and usefulness of strategic planning has been well documented over several decades of strategic management research. Despite the significant body of existing knowledge in the field of strategic planning, there remains a paucity of investigation into the construction sector, specifically in Professional Service Firms (PSF's) operating within it. The aim of this research was to ascertain the type, scope and extent of strategic planning within Irish Quantity Surveying (QS) practices and to ascertain the extent to which such processes correspond to strategic planning literature.

This research was an exploratory study, undertaken in two phases in line with mixed methods employed for undertaking the study. The first, qualitative, phase involved indepth semi-structured interviews with the principals of eleven QS practices of varying size. The second, quantitative phase, involved a widespread survey of every QS practice registered with the Society of Chartered Surveyors (SCS) in Ireland, for which a response rate of over 40% was achieved.

The findings discover that the type, scope and extent of strategic planning within Irish QS practices vary with practice size and ownership structure. Distinct groups of practices are evident based on a number of strategic planning process characteristics, including formality, approach, participation, flow and planning horizon. Despite the absence of a systematic or formal process within smaller QS practices, it is clear that principals are thinking and acting strategically. These practices broadly follow strategic planning processes advocated in the literature, are mostly unaware that this is the case; however confirm that a more systematic strategic planning process is beginning to emerge, particularly in light of the severity of the current economic and construction industry downturn in Ireland.

The findings of the research provide an important contribution towards addressing the significant gap in existing knowledge in this regard. The conclusions drawn are specific to the QS profession, the research has been designed such that it has potential to be applied to other PSF's within and outside the construction industry.

DEDICATION

This work is dedicated to Ian, Fionn, Donncha and our impending new arrival.

ACKNOWLEDGEMENTS

Many family, friends and colleagues in Dublin Institute of Technology provided considerable support throughout the course of undertaking my DBA, too many to thank individually. There are a number of people I do wish to specifically acknowledge who provided invaluable and much appreciated support.

Firstly I wish to thank my supervisor, Dr. Iain Henderson of Edinburgh Business School, for his guidance, advice and direction in undertaking this thesis.

To all the Quantity Surveyors who gave so generously of their time in participating in the research. The study would not have been possible without their involvement thus it is very much appreciated.

On a personal note, I wish to thank my mum and dad, who have always encouraged me in whatever I have done.

I would particularly like to thank my husband Ian for his love, support, encouragement and for listening when I need to let off steam!

Finally, to Fionn and Donncha, who are the two most beautiful little boys in the whole world and the best sons a mother could wish for.

DECLARATION STATEMENT

ACADEMIC REGISTRY

Research Thesis Submission



Name:	Róisín Murp	bhy	
School/PGI:	Edinburgh E	Business School	
Version: (i.e. First, Resubmission, Final)	First	Degree Sought (Award and Subject area)	Doctor of Business Administration

Declaration

In accordance with the appropriate regulations I hereby submit my thesis and I declare that:

- 1) the thesis embodies the results of my own work and has been composed by myself
- 2) where appropriate, I have made acknowledgement of the work of others and have made reference to work carried out in collaboration with other persons
- the thesis is the correct version of the thesis for submission and is the same version as any electronic versions submitted*.
- 4) my thesis for the award referred to, deposited in the Heriot-Watt University Library, should be made available for loan or photocopying and be available via the Institutional Repository, subject to such conditions as the Librarian may require
- 5) I understand that as a student of the University I am required to abide by the Regulations of the University and to conform to its discipline.
- * Please note that it is the responsibility of the candidate to ensure that the correct version of the thesis is submitted.

Signature of Candidate:	Date:	

Submission

Submitted By (name in capitals):	
	ROISIN MURPHY
Signature of Individual Submitting:	
Date Submitted:	

For Completion in Academic Registry

Received in the Academic Registry by (name in capitals):	
Method of Submission (Handed in to Academic Registry; posted through internal/external mail):	
E-thesis Submitted (mandatory for final theses from January 2009)	

TABLE OF CONTENTS

ABSTRACT	i
DEDICATION	ii
ACKNOWLEDGEMENTS	iii
DECLARATION STATEMENT	iv
TABLE OF CONTENTS	V
LIST OF TABLES	xi
LIST OF FIGURES	xiii
LIST OF ABBREVIATIONS	xiv
1. INTRODUCTION	1
1.1 Background to the Study	1
1.2 Research Question, Aims and Objectives	3
1.3 Research Methodology	4
1.4 Contribution of the Study	4
1.5 Outline of Thesis	5
2. STRATEGIC PLANNING	9
2.1 Introduction	9
2.2 Strategy Management and Planning Theory	10
2.2.1 Defining Strategic Planning	13
2.2.2 Strategic Planning and Organisational Performance	16
2.2.3 Strategic Planning in an Uncertain Environment	19
2.2.4 Strategic Types	22
2.2.5 Summary	24
2.3 Strategic Planning Characteristics	24
2.3.1 Comprehensiveness	25
2.3.2 Strategic Planning Flow and Participation	27
2.3.3 Strategic Planning Tools	29

	2.3	.4	Strategic Plan Duration	30
	2.3	5.5 Si	ummary	31
-	2.4	Mo	delling the Strategic Planning Process	32
	2.4	.1	The Modelling Approach	32
	2.4	.2	Edinburgh Business School Strategic Planning Process Model	35
,	2.5	Co	nclusion	39
3.	TH	IE C	ONSTRUCTION INDUSTRY IN IRELAND	41
-	3.1	Intr	roduction	41
-	3.2	Cha	aracteristics of the Construction Industry and Role of QS	41
-	3.3	Co	nstruction Industry and the Irish Economy	43
	3.3	.1	GDP, Employment and Inflation	43
-	3.4	The	e Quantity Surveying Profession in Ireland	47
	3.4	.1	Evolution of the Quantity Surveying Profession	47
	3.4	.2	Education, Training and Employment	49
	3.4	.3	Role of the Professional Body – The Society of Chartered Surveyors	50
-	3.5	Qua	antity Surveying Practice Organisation and Management	51
	3.5	5.1	The Professional Service Firm (PSF)	51
	3.5	5.2	Human Resources	52
	3.5	5.3	Marketing	53
	3.5	5.4	Quality Assurance	54
	3.5	5.5	Finance	55
-	3.6	Coi	nclusions	56
4.	ST	RAT	EGIC PLANNING IN QUANTITY SURVEYING PRACTICES	57
2	4.1	Intr	roduction	57
2	4.2	Stra	ategic Planning in the Quantity Surveying Profession	58
	4.2	2.1	Construction Industry Environment and Strategic Planning	59
	4.2	2.2	Application of Strategic Planning Theory in Construction	62
2	4.3	Stra	ategic Planning Characteristics in Quantity Surveying Practices	64

4.	3.1 Strategic Planning Formality and Appr	oach 64
4.	3.2 Strategic Planning Flow and Participat	ion 66
4.	3.3 Strategic Planning Tools	67
4.	3.4 Strategic Planning Time Horizon	68
4.4	Strategic Choice for Quantity Surveying Pr	actices 69
4.4	4.1 Corporate Level Strategy	70
4.4	4.2 Business Level Strategy	70
4.5	Monitoring and Implementation	75
4.6	Conclusion	76
5. LI	ITERATURE SYNTHESIS	77
5.1	Introduction	77
5.2	Merging Two Strands of Investigation	79
5.3	Defining Measurement Criteria	80
5.4	Conclusion	82
6. RI	ESEARCH METHODOLOGY	83
6.1	Introduction	83
6.2	Research Aim	83
6.3	Research Paradigm and Theoretical Framew	vork 84
6.4	Research Design	86
6.4	4.1 Phase 1: Semi-Structured Interview	86
6.4	4.2 Phase 2: Questionnaire	89
6.5	Sample Design	92
6.:	5.1 Phase 1: Qualitative Data	92
6.:	5.2 Phase 2: Quantitative Data	93
6.6	Conclusion	96
7. PI	ILOT STUDY	97
7.1	Introduction	97
7.2	Pilot Testing Phase 1: Semi-Structured Inte	rviews 97

	7.2	2.1	Findings from Phase 1 Pilot Study	101
	7.2	2.2	Phase 1 Pilot Study Conclusions and Adjustments for Main Study	102
	7.3	Pil	ot Testing Phase 2: Online Survey	103
	7.4	Co	nclusion	107
8	. DA	ATA	COLLECTION AND ANALYSIS	108
	8.1	Dat	ta Collection	108
	8.1	.1	Phase 1 Data Collection: Semi-Structured Interview	108
	8.1	.2	Phase 2 Data Collection- Online Survey	109
	8.2	Dat	ta Analysis	112
	8.2	2.1	Phase 1: Qualitative Data Analysis	112
	8.2	2.2	Phase 2: Quantitative Data Analysis	114
	8.3	Va	lidity and Reliability	114
	8.4	Res	search Ethics	115
	8.5	Lin	nitations of Methodology	116
	8.6	Co	nclusion	117
9	. RE	ESEA	ARCH FINDINGS AND ANALYSIS	118
	9.1	Pha	ase 1 Findings – Semi-structured Interviews	118
	9.1	.1	General Company Background	118
	9.1	.2	Strategic Planning in Irish QS Practices	123
	9.1	.3	Summary	151
	9.2	Str	ategic Planning Characteristics in Irish QS Practices	153
	9.2	2.1	Approach to Strategic Planning	153
	9.2	2.2	Strategic Planning and Process Formality	156
	9.2	2.3	Strategic Planning Comprehensiveness	158
	9.2	2.4	Strategic Planning Flow	159
	9.2	2.5	Strategic Planning Tools	159
	9.2	2.6	Constraints to the Strategic Planning Process	160
	9.2	2.7	Summary	161

9.2.8	Quantity Surveying Practice Groups	162
9.2.9	Summary of Phase One – Qualitative Data	175
9.3 Pha	ase 2 Findings - Survey	176
9.3.1	General Company Information	177
9.3.2	Strategic Management Practices	182
9.3.3	Strategic Planning	189
9.3.4	Miscellaneous Comments Received	199
9.4 Cor	mparative Analysis of Phase One and Phase Two Findings	201
9.4.1	Triangulation, Validation and Generalisation	201
9.4.2	Key Developments between Phase One and Phase Two	202
9.4.3	QS Practice Groups	206
9.5 Stra	ategic Planning Guidelines for Irish QS Practices	214
9.6 Co	nclusion	217
10. CON	ICLUSIONS AND RECOMMENDATIONS	218
10.1 I	ntroduction	218
10.2 C	Contribution to knowledge	218
10.2.1	Quantity Surveying Profession in Ireland	218
10.2.2	Strategic Planning Process in QS Practices	219
10.2.3	Groups within QS Profession in Ireland	220
10.2.4	Strategic Choice – Past, Present and Future	221
10.2.5	Strategic Planning Guidelines for Professional Practices	222
10.2.6	Tried and Tested Methodology	222
10.2.7	Application to other Professional Service Firm's	222
10.3 I	Limitations of Study	223
10.4 F	Recommendations for Future Research	224
10.4.1	QS Practice Groups	224
10.4.2	Performance Relationships	224
10.4.3	Construction Professional Practices	225

REFERENCES		226
Appendix A	Edinburgh Business School Strategic Planning Process Model	248
Appendix B	Augmented Process Model	250
Appendix C	Construction Industry Sectors	252
Appendix D	Strategic Planning in New Zealand QS Practices	254
Appendix E	Interview Prompt Sheet	257
Appendix F	Article for the SCS Magazine	264
Appendix G	Pilot Questionnaire Feedback Form	269
Appendix H	Questionnaire	272
Appendix I	Correspondence with Survey Respondents	283
Appendix J	Excerpt from Interview Transcript Exemplar Company A	288
Appendix K	Pertinent Points & Observations for Exemplar	300
Appendix L	NVivo 8 Coding Strategy	306
Appendix M	Examples of Exemplar Case Coding in NVivo8	313

LIST OF TABLES

Table 2.1	Relationship between planning formality and firm performance	18
Table 3.1	Key Construction Industry Statistics	44
Table 3.2	Construction Employment	45
Table 3.3	Construction Professionals	46
Table 4.1	Effect of environmental change over time	60
Table 5.1	Strategic planning characteristic and relevant author(s)	81
Table 6.1	Evolution of primary research methods in strategic management	85
Table 6.2	Phase one practices and number of employees	93
Table 6.3	Survey Population	95
Table 6.4	Sample Size and Response Rate	95
Table 7.1	Pilot study participant practice size and ownership structure	98
Table 8.1	Overall response rate to questionnaire	110
Table 9.1	Number of people employed in Phase 1 QS practices	119
Table 9.2	Change in Corporate Strategy	145
Table 9.3	Strategic type and approach to strategy (Phase 1)	154
Table 9.4	Phase 1 QS Practice Groups	163
Table 9.5	Number of people currently employed (2009)	178
Table 9.6	Number of people employed at peak (2007)	179
Table 9.7	Services currently provided	180
Table 9.8	Strategy type and approach to strategy (Phase 2)	184
Table 9.9	Formality of strategic planning across strategic approach	184
Table 9.10	Risk attitude	185
Table 9.11	Importance of Analysis	186
Table 9.12	Communication of Strategy	188
Table 9.13	Lifestyle choices	189
Table 9.14	Company size and formality of strategic plan	191
Table 9.15	Strategic planning comprehensiveness	192
Table 9.16	Experience with strategic planning	193
Table 9.17	Strategic plan content	195
Table 9.18	Obstacles to Formulating Strategic Plan	197
Table 9.19	Use of strategic plan	198
Table 9.20	Difficulties in strategic plan implementation	199
Table 9.21	Additional comments	200

Table 9.22	Changes in strategic type between phases	203
Table 9.23	Corporate strategy	204
Table 9.24	Business level strategy	205
Table 9.25	Comparative Table Group A	208
Table 9.26	Comparative Table Group B	210
Table 9.27	Comparative Table Group C	212

LIST OF FIGURES

Number of employees at peak (2007)		
Number of employees – July 2009	120	
Sectors serviced during peak (2007)		
Sectors serviced – July 2009		
Strategic type	124	
Business Level Strategy	147	
Approach to Strategy	154	
Approach to strategic planning versus risk attitude	156	
Links between groups	166	
Causal Map: Group A	170	
Causal Map: Group B	172	
Causal Map: Group C	174	
Position held by respondent	177	
Proportion of work currently undertaken within each sector	181	
Proportion of work undertaken within key sectors at peak	181	
Strategic Types (Phase 2)	182	
Approach to strategy	183	
Strategic plan time horizon	196	
	Number of employees at peak (2007) Number of employees – July 2009 Sectors serviced during peak (2007) Sectors serviced – July 2009 Strategic type Business Level Strategy Approach to Strategy Approach to strategic planning versus risk attitude Links between groups Causal Map: Group A Causal Map: Group B Causal Map: Group C Position held by respondent Proportion of work currently undertaken within each sector Proportion of work undertaken within key sectors at peak Strategic Types (Phase 2) Approach to strategy Strategic plan time horizon	

LIST OF ABBREVIATIONS

APC	Assessment of Professional Competence			
ARM	Agreed Rules of Measurement			
B&C	Building and Construction			
BER	Building Energy Rating			
BREEAM	Building Research Establishment Environmental Assessment			
	Method			
CAD	Computer Aided Design			
CAQDAS	Computer-assisted qualitative data analysis software			
CEEC	Conseil Européan des Economistes de la Construction			
	(The European Council of Construction Economists)			
CEO	Chief Executive Officer			
CIF	Construction Industry Federation			
CPD	Continuous Professional Development			
CSO	Central Statistics Office			
CWMF	Capital Works Management Framework			
DIT	Dublin Institute of Technology			
DKM	DKM Economic Consultants			
EBS	Edinburgh Business School			
EBS SPPM	Edinburgh Business School Strategic Planning Process Model			
ECB	European Central Bank			
EMU	European Monetary Union			
ESRI	Economic and Social Research Institute			
EU	European Union			
FOI	Freedom of Information			
GDP	Gross Domestic Product			
GNP	Gross National Product			
HR	Human Resources			
HRM	Human Resource Management			
IAVI	Irish Auctioneers and Valuers Institute			
IEI	Institute of Engineers in Ireland			
IFSC	International Financial Services Centre			
ΙΟ	Industrial Organisation			

IPT	Integrated Project Team			
ISO	International Standardization Organization			
IT	Information Technology			
MDP	Multidisciplinary Practice			
MS	MicroSoft			
NAMA	National Asset Management Agency			
NDP	National Development Plan			
NQAI	National Qualifications Authority of Ireland			
OJEC	Official Journal of the European Communities			
PASW	Predictive Analytics SoftWare			
РСР	Public Capital Programme			
P/E	Price Earnings			
PII	Professional Indemnity Insurance			
PSF	Professional Service Firm			
PQS	Professional Quantity Surveyor			
QNHS	Quarterly National Household Survey			
QS	Quantity Surveyor			
RBV	Resource Based View			
RIAI	Royal Institute of Architects in Ireland			
RIBA	Royal Institute of British Architects			
RICS	Royal Institution of Chartered Surveyors			
SCS	Society of Chartered Surveyors			
SCSI	Society of Chartered Surveyors Ireland			
SME	Small and Medium-sized Enterprise			
SPSS	Statistical Package for the Social Sciences			
SWOT`	Strengths Weaknesses Opportunities and Threats			
VAT	Value Added Tax			
VRIN	Valuable, Rare, Inimitable and Nonsubstitutable			

1. INTRODUCTION

1.1 Background to the Study

Strategic planning is a long-established component of strategic management theory (Ansoff, 1957, 1970; Pearce et al. 1987; Porter, 1979; Steiner, 1979). The field of strategic planning has attracted significant attention by academics and practitioners alike, and the benefits of engaging in systematic strategic planning are well documented (Anderson 2000; Baker et al 1993; Brews & Purohit 2007; Rudd, 2005; White 1986). The field may be broadly separated between strategic planning process and strategic planning content, the former being the primary concern of this investigation.

A substantial proportion of the existing empirical studies relating to strategic planning have been undertaken in the context of manufacturing sectors, while considerably fewer have focused on construction (Hillebrandt and Cannon, 1994; Green et al., 2008). The construction industry is a hugely important industry to the global economy, and in particular to the Irish economy when at peak in 2007, it contributed an estimated 24% of Gross National Product (DKM, 2010). However, the construction industry is undergoing a sharp contraction brought about by a combination of the global recession, domestic banking crises, overreliance on the industry and deteriorating exchequer balances. The net effect of which is that it is estimated the industry accounted for approximately 9.2% of GNP by the end of 2010 (DKM, 2010).

The construction industry is complex due to the unique nature of output, the number of professionals involved, investment intensity, pricing structure, the influence of government in the industry and the multiplied effect it has on other sectors (Ashworth and Hogg, 2002; Hillebrandt and Cannon, 1994). The limited research undertaken within the sector has been heavily biased towards large contracting firms, and there remains a dearth of information regarding the professional practices operating within the industry.

The Quantity Surveyor (QS) is one of a number of professionals involved in the construction process and has specific responsibility for project cost control not only

through the construction phase but for the whole life of the building (Seeley, 1997). The role of the QS has evolved such that they may be employed as construction project manager, risk analyst, investment appraiser or by financing agencies (Ashworth and Hogg, 2002). The shift towards more sustainable methods of construction and concern for the impact on the environment has offered further potential to broaden the role of the QS. A QS may work for a contracting organisation or for a Professional Quantity Surveying (PQS) practice (referred to as QS Practice for the remainder of the thesis), the latter forming the basis of the investigation at hand.

The QS plays a crucial role in a capital intensive industry that is clearly influential to the economy as a whole in Ireland, and thus far the strategic management and planning process of QS practices has not been the subject of empirical investigation. How strategic decisions are made, what informs such decisions, the process by which the organisation makes the decisions and the difficulties therein, remain unknown. The field of strategic planning is well established and facets of the process have been tested in varying industry sectors. The extent to which strategic planning processes are applicable to QS practices is as yet unconfirmed, thus providing the rationale for this exploratory study.

The use of a strategic planning process model provides a framework within which to analyse aspects of strategic planning in a firm. Numerous strategic planning models have been developed over several decades (Shrivastava and Grant, 1985; Rhyne, 1986; Dess, 1993; Scott, 2007; Abraham, 2006) including that developed by Scott (2007) known as the Edinburgh Business School Strategic Planning Process Model (EBS SPPM). The strength of the EBS SPPM lies in its potential application across industry sectors regardless of approach or degree of formality of process within an organisation. Consequently the model is used as a framework to investigate strategic planning processes within QS practices in Ireland.

The framework facilitates consideration of many facets of the strategic planning process from the approach to strategy formulation (Mintzberg, 2003), organisational types (Miles and Snow, 1978) and characteristics of the process. Features such as strategic planning comprehensiveness (Fredrickson and Mitchell, 1984), participation in the process (Dyson and Foster, 1982), flow (Segars and Grover, 1999) and use of strategic planning tools (Frost, 2003) are also addressed in the context of an industry environment which is clearly changing and uncertain (Brews and Purohit, 2007; Covin and Slevin, 1989; Grant, 2003).

The research thus addresses the perceptible gap in existing knowledge on strategic planning in QS practices in Ireland. The substantial body of existing strategic planning literature is thus employed to frame the investigation.

1.2 Research Question, Aims and Objectives

The background to the study outlined in section 1.1 above gives rise to the following research question:

"To what extent are systematic strategic planning processes evident within Irish Quantity Surveying practices?"

The aim of the research is:

"To investigate current and recent strategic planning processes evident within Irish Quantity Surveying practices."

The research has the following objectives:

- to investigate current and recent trends in strategic planning in Irish QS professional practices
- 2. to evaluate existing strategic planning practises within the Irish QS profession against the strategic management literature
- to develop a set of guidelines to assist systematic strategic planning in Irish QS practices

1.3 Research Methodology

The exploratory nature of the study required a flexible design thus mixed methods were employed.

Fieldwork was undertaken over two phases. Qualitative data was collected by means of in-depth semi structured interviews with eleven managing directors/directors of QS practices in Ireland. Interviews were recorded and transcribed and data was analysed using Nvivo 8 software.

Phase two took the form of an online survey of all QS practices registered with the Quantity Surveying Division of the Society of Chartered Surveyors (SCS), for which a response rate of over 40% was achieved. The second phase was designed along similar dimensions to the first, which facilitated triangulation of findings between phases.

Each phase was pilot tested prior to undertaking the main study. Participants from phase one were included in the second phase to validate findings and results from the second phase served to triangulate those from the first increasing generalisability.

1.4 Contribution of the Study

As noted there remains a paucity of research in Ireland on QS practices, and the overriding contribution of the study is in addressing this gap in knowledge.

The importance of the industry to the economy and the QS to the industry has been noted. The aims of the study outlined in section 1.2 serve to provide a significant contribution to knowledge within the field. Not only is the study amongst the first empirical studies of strategic planning practices within Irish QS practices, it has a practical application to those organisations, and possibly to other Professional Service Firms (PSF's) within the industry.

The study provides extensive data regarding the structure, strategies, process and characteristics of strategic planning within Irish QS practices. This is invaluable information for any QS practitioner with responsibility for strategy development and is currently unavailable.

Secondly, a number of guidelines have been developed based upon the findings of the research to aid the strategic planning process. The guidelines provide a useful framework for QS practices and may potentially be used in other professional organisations, particularly within the construction sector.

The timing of the research reflects a period of transition within the industry in Ireland whereby QS practices are adjusting to greater intensity of competition given the reduction in demand brought about by the financial and economic crises. The findings of the study may thus act as a benchmark against which future studies may monitor strategic change within the profession.

Finally, the research instruments used in the study were derived following an extensive investigation of existing strategic planning process literature and were rigorously pilot tested prior to being administered. Therefore another notable contribution of the study lies in that these instruments can now usefully be applied in future studies, both within the QS (and construction professions) as well as other PSF's outside of the construction industry.

1.5 Outline of Thesis

Chapter 1

Chapter one provides an overview of the rationale behind the study. The chapter outlines the research aims and objectives as well as the methodology employed for the investigation. The contribution of the study is overviewed prior to the presentation of an outline of the thesis.

Chapter 2

Chapter two presents a critical review of strategic planning process literature. There are many dimensions to the process including approach (Hamel and Prahalad, 1994; Mintzberg, 2003), tools and techniques (Porter, 2004a) planning formality, comprehensiveness, flow, content, participation (Papke-Shields et al., 2006) which are

considered. The complexity of analysis is greatly facilitated by the use of strategic planning models (Higgins, 1981; Rhyne, 1986; Shrivastava & Grant, 1985; Dess, 1993), hence a number of models are presented and analysed with particular focus placed on the EBS SPPM (Scott, 2007) due to its wide application across industry sectors.

Chapter 3

This chapter analyses the construction industry and the QS profession in Ireland. Following an overview of the macroeconomic contribution of the industry and recent trends in this regard, the QS profession is examined in detail. Consideration is given to the evolution of the profession, the role of the professional body as well as facets of strategic management of a professional practice (Maister, 2003; Løwendahl, 2005).

Chapter 4

Chapter four combines strategic planning and construction to provide an insight into the existing body of literature on the former in the context of the latter (Betts and Ofori, 1992; Hillebrandt and Cannon, 1989 and 1994; Hillebrandt, Cannon and Lansley et al. 1995). It is evident that there is a considerable gap in the literature in this regard, (particularly in an Irish context) hence forming the rational for undertaking the study.

Chapter 5

A synthesis of the literature is provided in chapter five, drawing conclusions from related literature. Themes are drawn from the literature which defines measurement criteria to be used in the main study.

Chapter 6

The methodology for the study is detailed in chapter six. As noted in section 1.3 above, mixed methods were employed for the study, which are outlined in this chapter. The primary research method employed is qualitative and takes the form of semi-structured in-depth interviews with eleven senior managers of QS practices.

The second research method employed was quantitative and took the form of an online survey of all chartered QS practices in Ireland that are registered with the Society of Chartered Surveyors (SCS). The survey achieved a response rate of 41%, and served as

a method of triangulating findings from phase one. Phase one participants were included in the second phase which further served to triangulate findings.

Chapter 7

Chapter seven presents findings from pilot testing both research methods. The pilot test for the qualitative study was undertaken with five QS practices and for the quantitative with seven industry professionals ranging from QS practitioners to academics within the field. Amendments between the pilot and main study are noted.

Chapter 8

Details of data collection and handling techniques are also detailed within this chapter. Qualitative data was analysed using Nvivo 8 software and the survey was administered using <u>www.surveymokey.com</u>. An overview of the software in both instances is provided together with how data was handled with the use of IT. Issues relating to validity and reliability are also discussed.

Confidentiality was ensured to all participants therefore neither individual nor company names are identified throughout the study. Confidentiality agreements were signed with the professional transcription company and the trainer in Nvivo software.

Chapter 9

The backbone of the thesis lies in chapter nine: research findings and analysis. A considerable amount of data was collected over two phases of field work which is presented and analysed in this chapter. The EBS SPPM has been used as a framework for analysis thus discussion of findings broadly follows the main components of the model.

Based on existing literature and findings from fieldwork, a number of guidelines have been developed to assist Irish QS practices in the strategic planning process which are also presented in this chapter.

Chapter 10

The final chapter outlines the conclusions drawn from the research, details the contribution to knowledge and notes limitations in the study prior to outlining recommendations for potential areas of further research in the area.

2. STRATEGIC PLANNING

2.1 Introduction

Strategy and strategic planning theory has evolved over a number of decades to such an extent that a plethora of literature has been published in the field, yet there remains no single definition of what strategic planning entails.

Strategic planning is associated with establishing organisational context, determining a desired future state and mechanisms by which the goal may be reached. Ultimately strategic planning is the process by which organisations seek to gain advantage over competitors (Waalewyn & Seggar, 1993), be it profit, market share, growth or survival (Lavender, 1996). There is considerable evidence to suggest that engaging in the strategic planning process improves firm performance (Anderson 2000; Baker et al 1993; Brews & Purohit 2007; Rudd, 2005; White 1986), hence it is crucial to the success of firms operating in a competitive environment.

This chapter provides a review of the literature pertaining to the core concepts of the strategic planning process. It commences with an overview of the evolution of strategic management theory including varying approaches to the strategic planning process and the role of the strategist in this regard. The relationship between strategic planning and company performance is important and despite it lying outside the scope of the study, an overview of key arguments is presented.

The second section of the chapter scrutinises characteristics of the strategic planning process including: strategic planning comprehensiveness, participation in the process, plan content and duration, and use of strategic planning tools. Issues pertaining to strategic planning within an uncertain environment are presented as well as difficulties associated with planning formulation.

The complex nature of the process may be facilitated by use of strategic planning models. The final section of the chapter reviews literature in this regard. A number of models are reviewed prior to a detailed examination of the EBS SPPM (Edinburgh Business School Strategic Planning Process Model). The EBS SPPM has potential

application across any industry sector regardless of the formality of approach toe the process thus forms the basis from which strategic planning within Irish Quantity Surveying (QS) practices is analysed for the purposes of the investigation.

2.2 Strategy Management and Planning Theory

Strategic management reflects the direction of an organisation in competing for customers, resources and revenue and has developed as a field of inquiry "*firmly grounded in practice*" (Rumelt et al., 1991 pp.6).

The discipline of strategic management (previously referred to as "*business policy*") has been the subject of investigation for several decades. Dating back to the work of Chandler (1962) and Ansoff (1957, 1965, 1970 and 1987), the importance of the field of study has resulted in an extensive array of theoretical and empirical scrutiny. Over time the subject has evolved reflecting changes in the competitive and industry environment facing organisations and the evolution itself has been documented by numerous authors (Armstrong, 1982; Robinson and Pearce, 1984; Shrader et al., 1984; Huff and Reger; 1987; Rajagopalan et al., 1993).

Strategic planning forms an important component of strategic management literature (Kuratko and Audretsch, 2009), and consequently an extensive literature base has developed broadly focusing on strategic planning content (Larson, 2000; Stonehouse and Pemberton, 2002) and process (Fredrickson and Mitchell, 1984; Papke-Shields et al., 2006). The former reflects the relationship between decisions and outcomes and the latter on the actions supporting strategy (Huff and Reger, 1987). Strategic planning is a multi-dimensional, complex process wherein the potential to incorporate related fields of management further broadens the scope of research thus exacerbating the complexity in measuring process variables (Huff and Reger, 1987; Rajagopalan et al. 1993). Understanding the process by which strategic decisions are made is a vital issue within the domain of strategic planning research thus consideration of process characteristics are examined in section 2.3.

Ultimately the approach to strategic planning shapes how decisions are made. Earlier authors advocate a rational, planned and formal process undertaken by a planning

department (Ansoff, 1965). The next wave, largely guided by the work of Porter (1980,1985 and 1994), has its base in industrial economics thus industry analysis and the positioning of the firm within the industry context forms the focal point of diagnostic models developed as well as the "*structure-conduct-performance*" framework of analysis.

Critics of the rational/planned approach posit that it is prohibitively restrictive. The incremental/emergent school claim that complex business environments require a flexible/incremental approach to strategy formation (Grant, 2003) requiring organisations "to function beyond the boxes to encourage the informal learning that produces new perspectives and new combinations." (Mintzberg, 1994; pp.109).

Recent additions to the "*planned versus emergent*" strategy debate emphasise the importance of both the analytical and incremental schools of thought, thus merging the benefits of both in what has been termed "*planned emergence*" (Grant, 2003). This approach advocates a structured yet flexible approach to strategic planning enabling the organisation to adapt the strategy in light of changing circumstances. This view is shared by several prominent authors (Eisenhardt, 1989 and 2000; Hamel and Prahalad, 1996 and Grant, 2003) including Brews and Hunt (1999) who proclaim that:

"...specific plans may represent the "intended' strategy while the inevitable incremental changes that follow as intentions become reality represent the emergent, or "realized' part of the firm's "deliberate' strategy. Both are necessary and neither is sufficient." (Brews and Hunt, 1999, pp. 903)

The conclusion drawn from the above is that, in practice, strategic planning lies somewhere between the two extremes of purely planned and purely emergent approaches. It may be argued that strategy should be planned and developed systematically yet remain sufficiently flexible to react to change within a dynamic environment.

The resource based view (RBV) of the firm provides another dimension for consideration namely that the ability of an organisation to maintain competitive

advantage lies within the internal resources and capabilities of the firm (Barney, 1991). Aspects of the RBV echo the emergent approach put forward by Mintzberg (1994) including the recognition of a firm's history in shaping strategy. It is noted that:

"If a firm obtains valuable and rare resources because of its unique path through history, it will be able to exploit those resources in implementing value-creating strategies that cannot be duplicated for other firms, for firms without that particular path through history cannot obtain the resources necessary to implement the strategy." (Barney, 1991, pp.108)

The concept of a firm obtaining competitive advantage due to its internal resources and company history resources is particularly applicable to professional service firms (see chapters 3 and 4). Kraajienbrink et al. (2010) are, however, critical of the RBV suggesting that sustained competitive advantage is not in fact achievable in the long run at all.

Despite its critics, the RBV has provided a considerable contribution to strategic management theory. Issues such as core competencies, dynamic capabilities, leadership, knowledge, learning organisations have derived from the RBV and consequently have been driven to the forefront of research and practice (Eisenhardt & Martin, 2000; Teece et al. 1997).

The net effect of several decades of strategic management research is a multidimensional field of study. As noted by Rumelt, Schendel and Teece (1991):

"In looking backward over these three decades, what comes into focus is the search for theoretical explanations of very complex phenomena." (Rumelt, Schendel and Teece, 1991, pp.9)

The complexity is exacerbated by the lack of consensus regarding fundamental issues such as the impact of formal strategic planning on performance (Boyd and Reuning-Elliott, 1998) and the degree of formality in the process (Brews and Purohit, 2007). Evidence suggests that environmental uncertainty will impact on the strategic planning process (Koufopoulos et. al, 2000; Grant, 2003), however, planning characteristics will

vary from one organisation to the next (Fredrickson and Mitchell, 1984; Pearce et. al 1987, 1988; Stonehouse and Pemberton, 2002) and from one industry to the next (Langley, 1988; Grant, 2003).

These matters must be considered in the context of a discipline for which no single definition exists.

2.2.1 Defining Strategic Planning

A multitude of studies have been undertaken to determine the extent, type and scope of strategic planning (Gibbons and O'Connor, 2005; Shuman, et. al, 1985; Stonehouse and Pemberton, 2002) across various industries including: manufacturing (Pearce et. al 1987, 1988), public sector (Langley, 1988), oil industries (Grant, 2003), information technology (Eisenhardt, 1989; Segars and Grover, 1999) and construction (Betts et al. 1991; Dansoh, 2005; Waraszowski, 1996). Further investigations have concentrated on company size rather than sector including SME's (Baker et. al, 1993; Frost, 2003; Gibbons and O'Connor, 2005) or Fortune 1000 companies (Higgins, 1981).

Despite the plethora of literature in the field of strategic planning no single definition of strategic planning is universally agreed upon. Terms such as superior performance, competitive advantage over rivals (Grant 2001), growth (Larsen et al., 2000) and formality (O'Regan and Ghobadian, 2007) are often discussed within existing definitions, all of which are valid but none of which are applicable to every business all the time. Growth often does not form part of the strategic plan of a company in a mature industry, similarly the process and outcomes are not necessarily formalised.

Formality is often a defining feature of strategic planning. Armstrong (1982) contends that:

"Formal strategic planning calls for an explicit process for determining the firm's long range objectives, procedures for generating and evaluating alternative strategies, and a system for monitoring the results of the plan when implemented." (Armstrong, 1982, pp. 198)

The emphasis on formality is reiterated by Larsen et al (2000) who describe strategic planning specifically as:

"A formal review of a company's strengths and weaknesses and future prospects leading to decisions aimed at the company's growth" (Larsen et al., 2000, pp. 278)

Caution must be expressed at the use of a prescriptive definition such as the above. Evidence suggests that strategic planning may be undertaken on an informal basis, and sometimes even unbeknown to the strategist (Hillebrandt and Cannon, 1994). Furthermore, as noted previously, organisational growth may not be the overriding objective of the organisation and therefore the definition cannot have general application to all businesses.

Stonehouse and Pemberton (2002) overcome the confines of formality and growth by defining strategic planning as:

"The devising and formulation of organisational level plans which set the broad and flexible objective, strategies and policies of a business, driving the organisation towards its vision of the future." (Stonehouse and Pemberton, 2002, pp.854)

This definition implies the need for a systematic yet flexible approach to achieve organisational goals thus has potential for widespread application across a variety of industries. For this reason, the definition of strategic planning adopted for the purposes of this study is that which is provided by Stonehouse and Pemberton (2002) above. The authors also make an important distinction between strategic planning and business planning based on the time horizons which they address:

"*(business planning)* is concerned with the short-term analysis of a business, the attainment of short-term goals and functional level planning." (Stonehouse and Pemberton, 2002, pp.854)

The distinction between strategic planning, business planning and strategic thinking is also noted by Mintzberg (1994) who contends that "*strategic planning is not strategic thinking…the most successful strategies are visions, not plans*" (pp.107). He argues that strategic planning is about analysis whereas strategic thinking is about synthesis. Both are necessary yet separate facets of the strategic decision making process. Mankins and Steele (2006) go further by proclaiming that

"In most companies, strategic planning isn't about making decisions. It's about documenting choices that have already been made, often haphazardly." (Makins and Steele, 2006, pp. 76)

This view is supported by Grant (2003) who noted in the study of the oil majors that "most strategic decisions appear to be made outside of formal strategic planning systems" (pp. 494).

One may argue, however, that the strategic planning process is not designed to incorporate every strategic decision. The issue is addressed by Hamel and Prahalad (1994) who recognise that strategic decisions are made on an ongoing basis outside of the formal strategic planning process. They coined the phrase "strategic architecture" to represent the scope and purpose of the strategic planning process, providing an overall structure for strategy development rather than the detailed accumulation of every strategic decision made. The highway analogy used goes a long way to explaining the relationship:

"Strategic architecture is a high-level map of interstate highways, not a detailed map of city streets. It is specific enough to provide a general sense of direction, but doesn't detail every side street along the way." (Hamel and Prahalad, 1994, pp. 119)

Strategic planning is thus associated with determining the overriding goals and objectives and the desired future direction of the firm while, allowing the flexibility for making strategic decisions along the way based on circumstances as they arise. In fact Hamel and Prahalad (1994) contend that long term planning is not only impossible but also a "*recipe for inertia*".

As is evident there is no agreement regarding the definition of strategic planning. Whatever the chosen definition of strategic planning or indeed the approach to analysis, an interesting observation is made by Chaffee (1985) when it was noted that:

"Researchers and practitioners have used the term strategy freely – researchers have even measured it – over two decades...No controversy surrounds the question of its existence; no debate has arisen regarding the nature of its anchoring concept. Yet virtually everyone writing on strategy agrees that no consensus on its definition exists." (Chaffee, 1985, pp. 89)

Regardless of the definition of strategic planning one uses there can be no doubt that the process of strategic planning is important for shaping the desired future direction of an organisation. Although many organisational objectives may be pursued they often centre on gaining advantage over competitors and ultimately to improve company performance. The relationship between strategic planning and performance continues to be debated extensively in the literature, and despite it lying outside the scope of the research at hand its importance is such that is considered in the following section.

2.2.2 Strategic Planning and Organisational Performance

The relationship between strategic planning and performance has played a dominant role in strategic planning research. Despite the many uses of a strategic plan (Langley 1988; Baker et al. 1993) the overriding purpose of engaging in the process is to increase organisational performance (Grant, 1991). Empirical findings remain inconclusive as regards the planning-performance relationship, partly due to a lack of agreement as to what constitutes formality, and what performance measures should be used. A comprehensive overview of previous research on the relationship between strategic planning and organisational performance is provided by Boyd and Reuning-Elliott (1998) who note the challenges associated with inconsistencies in describing and operationalising strategic planning (pp.182-183).

An important distinction must be made between strategic *plan* formality and strategic planning *process* formality. The former rests on the existence of a written plan whereas the latter is the process by which it is undertaken:

"A formal strategic planning process is a deliberate means to include factors and techniques in a systematic way to achieve specified tasks" (O'Regan & Ghobadian, 2007, pp. 13)

The definition above is sufficiently broad to be generally applicable across industry sectors, yet recognises the deliberate and systematic process required to achieve organisational goals.

Various mechanism are used to measure organisational performance many of which are based on accounting methods such as by Price/Earnings (P/E) multiple (Welch, 1984), profit margin, return on assets, loan growth and return on equity (Robinson and Pearce, 1983). Alternative performance criteria may include market share, customer service criteria, quality or innovation-related measures (Davig et al., 2004) all of which are equally valid. In fact it may be argued that financial measures of performance are by their nature considering the past rather than the future.

No consensus exists as to the direct relationship between the variables, even taking cognisance of the differences in measurement mechanisms. A strong case has been established in support of the relationship between formal strategic planning and performance (Anderson 2000; Baker et al 1993; Brews & Purohit 2007; Rudd, 2005; White 1986), however, a number of studies show little or no correlation. Robinson and Pearce (1983) noted that the relationship between planning formality and financial performance was weak, while a later study (Pearce, Robbins and Robinson, 1987) concluded that strategic planning formality was positively related to each of the performance indicators used (significantly so in the case of sales growth). Pearce and Robinson are not alone in drawing varying conclusions. The table below outlines a small number of contributors to the planning-performance debate.

Author	Date	Sector	Relationship between formality &
			performance
Robinson and Pearce	1983	Small Banks	No relationship between formality and financial performance
	1004		
Mitchell	1984	Forest products	Negative (in unstable environment)
Fredrickson	1984	Paint & Coatings	Positive (rational model) in a stable environment
Pearce, Robbins and Robinson	1987	Various	Positive relationship
Eisenhardt	1989		Positive relationship
Baker, Addams & Davis	1993	Various	No relationship
Brews and Hunt	1999	Various	Positive relationship
Falshaw & Glaister	2006	Various	No relationship

Table 2.1 Relationship between planning formality and firm performance

Murphy (2010), Unpublished DBA thesis

The table above demonstrates the lack of consensus regarding strategic planningperformance relationship across a range of industry sectors despite being the focus of analysis over a prolonged period of time.

Even if there is a relationship between the two variables, causality is not confirmed in many cases. Does improved organisational performance facilitate comprehensiveness in the strategic planning process, or is it the case that a comprehensive strategic planning process results in improved performance? Does the positive relationship exist because the process brings an improved understanding of the business and its environment hence the firm is better position to compete within it (Welch, 1984; Rhyne, 1986)? A key limitation of the frequently cited Fredrickson and Mitchell (1984) study is just that: causality is not established.
Final judgement has yet to be made as to the definite link between strategic planning and performance (Rhyne, 1986). The relationship is be dependent upon economic and industry variability, as well as managerial competence (Ackelsberg and Arlow, 1985). Shrader et al (1984) reviewed several decades of research concluding there is "*no clear systematic relationship between formal long-range planning and organizational performance*" (pp.154).

2.2.3 Strategic Planning in an Uncertain Environment

An understanding of the industry environment within which an organisation operates is a vital aspect of the strategic planning process, hence it has been widely researched across a vast range of variables (Junnonen, 1998; Grant, 2003; Brews and Purohit, 2007). The environment influences strategy as well as the competitive position held by an organisation, therefore analysis of the macroeconomic, industry, competitive and internal organisational environments are important aspects of the strategic planning process (see also section 2.4.2.2).

The importance of environmental analysis for survival is emphasised in that it "behoves every organisation to be aware of the nature of the environment that it currently faces and those which it may face in the future" (Fahy, 1981; pp.32). Changes in the environment will impact organisational strategy so much so that Dansoh (2005) proclaims that:

"Strategy is defined as a long-term plan aimed at pre-empting and controlling changes in the firm's operating environment." (Danshoh 2005, pp.164)

No mention as to how a firm can pre-empt and actually control the environment is put forward, and one must wonder if it is possible to fully control the environment in the first instance. That is not to say, however, that consideration of the environment is not an important component of the strategic planning process. On the contrary, it is vital. Evidence suggests a positive relationship between environmental uncertainty and the formality of the strategic planning process (Koufopoulos et. al, 2000; Gibbons and O'Connor, 2005; Falshaw et al., 2006). An often cited investigation published by Covin and Slevin (1989) measured performance of small firms in hostile and benign environments concluding that small firms with an organic structure perform best in hostile environments and while structure does not influence firm performance, environmental hostility does. Importantly the "*fit between these variables and the firm*'s *business practices and competitive tactics*" (pp. 87) will ultimately determine firm performance.

This argument is supported by Dess et al. (1997) when in their examination of entrepreneurial strategy making it was concluded that a positive relationship existed between entrepreneurial strategy making and firm performance, when coupled with environmental uncertainty as well as a low cost strategy. A possible explanation for this relationship is provided by Gibbons and O'Connor (2005) who argue that entrepreneurial firms undergo more frequent analysis of where their competitive advantage lies, thus requiring more intensive environmental analysis.

Brews and Purohit (2007) in their multinational survey analysed the relationship between planning and the environment and concluded that as the extent of environmental uncertainty increased so too did the extent of planning. However, factors within managements control were a larger determinant in the increase in planning efforts rather than the external environment. A related discovery is made by Peel and Bridge (1998) who note that:

"...the intensity (detail) of strategic planning in SME's is strongly associated with environmental change, but that planning intensity is not significantly associated with the intensity of market competition in which businesses participating in the survey operated." (Peel and Bridge, 1998, pp. 853)

The suggestion above is that the strategic planning process is influenced to a greater extent by environmental uncertainty than by competition. One shortcoming of the study lies in that it does not draw conclusions regarding the extent, type or scope of competitor analysis, which may impact the deduction above.

The effect of a turbulent environment may influence strategic planning in ways other than performance. Reduced time horizons, less formality, emphasis placed on performance planning and decentralised decision making may be a consequence of operating within a turbulent environment (Grant, 2003). It has also been argued that within *"high velocity"* environments that fast decision makers use more information and identified more alternatives than slower decision makers, ultimately improving performance (Eisenhardt, 1989).

Despite the convincing empirically tested evidence in support of the inter-relationship between the environment and the strategic planning process a minority of authors claim that environmental uncertainty has a detrimental affect on strategic planning. Stubburt (1985) challenges the core of environmental analysis concluding that traditional management conventions cannot be utilised in turbulent environment.

The evidence in support of the relationship between strategic planning and environmental uncertainty is, however considerable (Fahy, 1981; Koufopoulos, 2000; Grant, 2003; Brews and Purohit, 2007). The strength of the relationship requires the organisation to monitor changes in the environment over time. An interesting perspective in this regard is provided by Brews and Hunt (1999) when they proclaimed that:

"Firms emerging from stable environments should be especially cautious. Since one possible outcome of operating in a stable environment is an underdeveloped planning capability, the learning curve faced by such firms when more sophisticated planning becomes necessary may be steep." (Brews and Hunt, 1999, pp. 906)

This raises an important consideration for strategic planning within the construction industry in Ireland (which is further addressed in chapters 3 and 4).

It is impossible to specify a "one size fits all" approach to strategic planning as clearly the process will differ depending not only on the environment within which a firm is operating, but also the attitude and approach employed by the strategist. The following section addresses important considerations in this regard.

2.2.4 Strategic Types

Approach to strategy formation may be planned, emergent or indeed follow a "planned emergence" approach reflecting the necessity for a systematic yet flexible strategic planning system. The strategic type of the organisation is a key influence in determining the approach.

The most prominent contributors in this regard are Miles and Snow (1978) who provide a framework or typology of approaches to the strategic planning process. They contend that although each organisation may have differing strategies based upon their unique characteristics, patterns of behaviour will transpire centring on four organisation types, namely: defenders, prospectors, analysers and reactors. (Miles and Snow, 1978; pp.29)

Of particular importance in the context of the research at hand is the supposition that strategic typology affects the planning process (Miles and Snow, 1978). Defender organisations concentrate on "problem solving rather than problem finding... undertaken prior to organisational action" (pp. 42). Prospectors engage in a planning process that is "broad rather than intensive, oriented toward problem finding, and contingent upon feedback from experimental action" (pp.61). Planning in analyser firms display similar characteristics to both defenders and prospectors whereby planning processes are "intensive and comprehensive" (pp.75). Reactor firms are deemed "an unstable organisation type because it lacks a set of consistent response mechanisms" (pp.81).

The strategic typologies were developed following investigations conducted within a variety of industry settings and its potential application to any industry is a key advantage of the approach. The taxonomy is particularly useful for the empirical examination of an industry/sector where exploratory research is being undertaken.

Consequently, the strategic types (Miles and Snow, 1978) are used for the purposes of classifying organisations in the study at hand.

It has been argued that strategic type may influence organisational performance. Conant et al. (1990) explored this relationship and concluded that there is little difference in the performance of prospector, analyser and defender organisations, however, each outperforms reactors. An extension of the study undertaken by Contant et al. (1990) across other sectors would add weight to the findings, which at present are restricted to a single industry as it is likely that reactors may perform well in certain environments.

The Miles and Snow (1978) typologies have been tested by a number of authors (Shortell and Zajax, 1990; Desarbo et al., 2005) and relatively few criticisms of the model exist. One criticism is, however, presented by Desarbo et al. (2005):

"A business will select a particular strategic type based on its particular internal strengths (capabilities) and external (environment) circumstances, and the strategic types that are actually employed may not, in fact, be cleanly interpretable as the four Miles and Snow (1978) categories" (Desarbo et al., 2005, pp. 51)

In fact in their "revisit" of the typologies they suggest that the "analyser" does not constitute a separate group (pp.62). Perhaps it is the case whereby prospectors and defenders display analyser characteristics, whereas it is unlikely that a reactor would share the same traits. A weakness in their argument is that seldom will an organisations' behaviour be "cleanly interpretable". The typologies act as a framework of analysis rather than a prescriptive guideline as to how to approach strategy.

The organisational type will influence the approach not only to strategic planning but in determining the future goals of the firm, thus form crucial components of the process. As is the case for other dimensions of strategic planning, the approach will differ from one firm to the next and will influence the extent, scope and characteristics of the process.

2.2.5 Summary

Strategic planning has been the focus of strategic management research for several decades. The breadth of the field and the potential to incorporate several aspects of strategic management result in a complex web of inter-relationships, many of which are firm and industry specific. No single definition exists as to what constitutes strategic planning, yet it is accepted as being an important function of business.

There remains conflicting views regarding the relationship between strategic planning and organisational performance, however, considerable evidence supports the notion that a positive relationship between environmental uncertainty and strategic planning exists.

The approach to the process may vary considerably with differing affects on firm performance. The historic "planned versus emergent" approaches to strategic planning have evolved in recognition that the uncertain environment faced by organisations requires a systematic yet flexible approach to strategic planning.

The required flexibility in the approach to strategic planning results in firm's displaying differing characteristics in undertaking the process. The organisation type and approach to strategy clearly influence these characteristics, yet once again there is no "best practice" set of features that ultimately prove the most advantageous. The following section outlines and discusses a number of strategic planning characteristics.

2.3 Strategic Planning Characteristics

The complexity of strategic planning analysis is compounded by numerous variations in the characteristics of the planning process itself. Dissimilarities are apparent in strategic planning characteristics between firms of varying size, experience and industry sector. Characteristics exhibited in manufacturing may not be apparent or indeed appropriate within professional service firms (PSF's) and no consensus as to the optimum array of strategic planning characteristics necessary to achieve competitive advantage is universally agreed upon. Segars and Grover (1999) provide invaluable insight into the diversity of strategic planning characteristics which broadly parallel the planning schools developed by Mintzberg (1990). The potential to group firms along strategic planning characteristics is particularly useful for an exploratory study therefore consideration must be given to such characteristics.

The analysis to follow focuses on strategic planning comprehensiveness (Fredrickson and Mitchell, 1984; Segars and Grover, 1999; Papke-Shields et al. 2006), participation (Dyson and Foster, 1982 and Collier et al. 2004), flow (Baker et al. 1993), use of strategic planning tools (Frost, 2003 and Stonehouse and Pemberton, 2002), content (Boyd and Reuning-Elliott, 1998) and plan duration (Glaister and Falshaw, 1999; Koufopoulous, 2000 and Stonehouse and Pemberton, 2002).

Potential exists to increase the scope of analysis to issues including leadership style, planning departments, motivation theory and corporate governance amongst numerous others; however, for the purposes of the research at hand discussion of characteristics is confined to those outlined above.

2.3.1 Comprehensiveness

An important characteristic of the strategic planning process is the degree to which an organisation gathers and analyses information inform decisions. Strategic planning comprehensiveness is defined by Fredrickson and Mitchell (1984) as follows:

"Comprehensiveness is a measure of rationality and is defined as the extent to which organizations attempt to be exhaustive or inclusive in making and integrating strategic decisions." (Fredrickson and Mitchell, 1984, pp. 399)

Many factors influence the exhaustiveness with which information is gathered and alternatives generated. The environment within which a firm is operating influences the comprehensiveness of the strategic planning process (Eisenhardt, 1989). A single industry investigation undertaken by Fredrickson and Mitchell (1984) initially focused on unstable environments with a subsequent study within a stable environment

ultimately concluding that an incremental approach is best suited in an unstable environment. A limitation of the conclusion is that causality is not determined. It may be the case that in fact increased or decreased company performance (rather than the environment) may result in a firm being more or less comprehensive in their strategic decision making process. For example a firm experiencing a period of poor performance may make future strategic decisions more comprehensively. Another limitation of the study lies in the research methodology. A single scenario presented to CEO's formed the basis of interviews, which clearly narrows the scope of the investigation.

Strategic type may influence the extent of comprehensiveness in the strategic planning process. Comprehensive planning is a feature of prospector and analyser organisations in particular (Miles and Snow, 1978). This is not to suggest that such firms will not make entrepreneurial decisions, it is simply implies a thorough and exhaustive process.

Company size will also have an impact although there is a lack of consensus in this regard. Gibbons and O'Connor (2005) concluded that Irish SME's embark on regular analysis to maintain competitive advantage. Whereas Frost (2003) argues that although company size is an important determinant of comprehensiveness, the strategic planning process in SME's is not comprehensive:

"Whilst large organisations understand the need for in-depth planning, there are indications that small to medium-sized enterprises (SME's) do not necessarily undertake planning to the same depth, nor do they show the same intensity with respect to strategic planning." (Frost, 2003, pp. 50)

Both the Gibbons and O'Connor (2005) and Frost (2003) studies focus on SME's, the former within an Irish context and the latter in Australia. The investigations in each case could be enhanced by expanding the scope to include large firms within a particular industry in order to compare results within a single sector. This would provide definitive results as to the affect of organisation size on strategic planning comprehensiveness.

Arguably the most significant contribution to the discussion of strategic planning comprehensiveness is provided by Segars and Grover (1999). In their investigation of

253 organisations, five profiles of strategic information systems planning became apparent based around the schools of thought developed by Mintzberg (1991). A summary of critical findings for each school is outlined below:

- Design school: moderate levels of comprehensiveness, formality and creativity.
- Planning school: higher levels of comprehensiveness and formalisation
- Positioning school: higher levels of comprehensiveness and formalisation (than design school but lower than planning school)
- Learning school: high levels of comprehensiveness and formalisation.
- Political school: low levels of comprehensiveness and formalisation

The conclusions outlined above provide strong evidence to support the "*planned emergence*" approach; implying that a marrying together of planning and learning schools will result in higher levels of comprehensiveness and formalisation.

Informed by the work of Segars and Grover (1999), Papke-Shields et al (2006) also concluded that the strategic manufacturing process varies systematically with respect to rational and adaptive planning characteristics in medium to large organisaitons. Similarly to the findings of Segars and Grover (1999), clusters of organisations were evident amongst respondents within the particular industry/sector. A limitation of the findings in the context of this study lies on its reliance on manufacturing firms alone. The investigation and its applicability to the service sector, while unconfirmed presents an opportunity for further research.

2.3.2 Strategic Planning Flow and Participation

Strategic planning flow and participation are characteristics of the strategic planning process which will vary between firms. The flow may be either "top-down" or "bottom-up". The former flows from senior management who assume responsibility for the process and lower levels of staff have limited involvement in strategy development. The latter involves several (often lower) levels of staff in the process who may in fact initiate the process.

Strategic planning initiatives frequently flow from the top-down, regardless of the overall approach to strategy development (Baker et al. 1993, Caeldries and vanDierdonck, 1988; Papke-Shields et al., 2006; Segars and Grover, 1999). However, Mintzberg (1994) highlights "the fallacy of detachment" of the planned approach to strategy development. The detachment of top management has important implications in the context of the incremental approach to strategy formation:

"Learning and visionary approaches appear to be superior to planning as a means for creating strategy...(senior managers are) cut off from soft data as a result of being perched atop overgrown hierarchies and over diversified operations...find themselves unable to use these other approaches." (Mintzberg, 1994, pp.270)

Thus the tendency for strategic planning to flow from the top of the organisation down rather than the opposite direction may potentially result in strategy being developed in the absence of first hand knowledge of what is happening on the ground. The resulting disconnect may have significant implications in pursuit of competitive advantage.

A separate but related characteristic lies in the degree of participation in the strategic planning process. The greater the degree of participation from across functional departments and levels of management, the more ownership staff members have of any subsequent strategic plan developed; hence the more likely they will be willing to contribute to its implementation. It has been argued that top-down planning approaches lead to limited participation (Segars and Grover, 1999). However, the role of senior management remains crucial to the success of the strategic planning process (Miller et al. 1988; Collier et al. 2004).

The extent of participation of the strategist is addressed by several authors (Miller et al. 1988; Collier et al., 2004) Following a study of over six thousand managers over a seven year period Collier et al., (2004) discovered that managers who report higher levels of involvement in strategy perceive the process in a more favourable light, resulting in them acting in ways that make the process more effective. The positive effects of increasing the scope of involvement included increased sense of shared ownership and shared vision, while the anticipated political issues arising from

increased participation were not realised. The findings of this study confirm that company wide participation including that of senior management will result in a more effective strategic planning process.

Once again, Segars and Grover (1999) investigated the extent of participation within planning schools outlined by Mintzberg (1990). Those firms associated with the planning school displayed narrow participation with the learning school characterised by broad participation in the strategic planning process.

The flow and participation in the strategic planning process provide insight into staff involvement in the strategic planning process. Senior management support is crucial for the success of any strategic initiative, however, the involvement of business unit level staff is fundamental to ensure key strategic issues are addressed an understood by all to ensure successful implementation. This is particularly important for PSF's whereby many levels of staff have direct contact with the client, thus imparting the vision, ethos and objectives of the firm during such communication. This in turn influences the client's perception of the organisation (see section 4.4.3).

2.3.3 Strategic Planning Tools

The importance of the strategic planning process and the multi-dimensional nature of the discipline have resulted in a number of tools being developed to assist the process.

The range of strategic planning tools potentially encompasses every facet of organisation management and may range from financial performance indicators such as ratios or cost benefit analysis to identifying organisational strengths, weaknesses, opportunities and threats (SWOT). Specific tools have been developed for industry analysis including Porters (1980) five forces model which continues to hold a prominent position within the published literature. Scenario planning is a useful tool within the context of an uncertain environment as well as Balanced Scorecard (BSC) for the continued measurement of performance against pre-defined criteria. A full glossary of strategic planning tools is contained in Frost (2003).

There remains scepticism regarding the extent to which strategic planning tools are used in practice (Koufopoulos and Chryssochoidis, 2000). Stonehouse and Pemberton's (2002) study of 159 SME's confirmed that despite 90% of respondents recognising the importance of strategic planning, the use of strategic planning tools was less apparent. Larger organisations were more likely to use tools for strategic analysis, however, the overall conclusion was that the use of "*tools of analysis largely remain the domain of academics and observers*" (pp.860).

This view is supported by Grant (2003), who revealed that:

"One of the ancillary observations of our study was the limited use by the companies of recently developed strategy concepts and techniques. Despite the rapid diffusion of the tools and techniques of strategic management during the 1980's and 1990's, few of these found application in the strategic planning processes of the oil majors." (Grant, 2003, pp. 513)

The tools and techniques developed to aid the strategic planning process clearly have mixed application in practice (Glaister and Falshaw, 1999). Caeldries and van Dierdonck (1988) eloquently suggest that "*theoretical development thus seems to be far ahead of strategic planning practice*" (pp.45). The reason why strategic planning tools remain in the domain of academia are not noted, leaving one to assume that either practitioners are unaware of their usefulness, are aware of their existence but lack the skills to use them, or believe them to have no practical use. Obviously, a combination of the three reasons is also possible.

2.3.4 Strategic Plan Duration

In the context of a changing environment the content of the strategic plan cannot remain static. The time horizon of the strategic plan thus requires consideration. The planning horizon is important as it is argued that "*a lapse of time is necessary before the fruits of good planning show up in externally oriented performance data...*" (Brews and Hunt, 1999, pp. 905).

The literature presents a general consensus on the strategic planning time horizon. Stonehouse and Pemberton (2002) confirm that for both sectors within their study that the average time horizon for planning is three years. Koufopoulos et. al (2000) in their Greek study discovered that 51% of those surveyed formulate plans for 1-3 years with a further 45% planning from 3-5 years. A similar planning horizon was noted in study of Belgian firms undertaken by Caeldries and van Dierdonck (1988).

A contrasting finding is presented by Glaister and Falshaw (1999) who contend that relatively few firms plan beyond five years in terms of production, sales, HR or financial planning, however, "*the exception to this is the strategic plan where over 55% of firms plan for five or more years ahead*" (pp.109).

It may be concluded that strategic plans are most commonly undertaken for a period of less than five years, and a likely reason for this time horizon may be that

"Creating a detailed plan for a ten-or fifteen-year competitive quest is impossible. Planning assumes a degree of exactitude that is impossible to achieve when one looks out beyond the next two to three years. Insisting on such exactitude before embarking on a new strategic direction is a recipe for inertia and incrementalism." (Hamel and Prahalad, 1994; pp.119)

2.3.5 Summary

With a multi-dimensional discipline like strategic planning, characteristics of the process may be analysed along a number of dimensions as previously outlined.

The importance of identifying the characteristics lies in the possibility for grouping clusters of firms within an industry/sector based on the characteristics displayed. This aspect is particularly useful in the context of an exploratory study such as the investigation at hand.

The complex nature of strategic planning, and varying characteristics displayed across organisations has resulted in a number of models being developed to assist the process

of strategic planning. The benefits of a modelling approach lies primarily in providing a framework for analysing the process. The following section details key aspects of modelling the strategic planning process.

2.4 Modelling the Strategic Planning Process

2.4.1 The Modelling Approach

The use of models to analyse complex phenomena has a long history (Baden-Fuller and Morgan, 2010). The multi-faceted thus complex nature of the strategic planning process warrants the use of a model to aid not only the process itself, but also the analysis thereof. A model may be defined as:

"...a structured method of thinking which enables the component parts of complex process to be identified and related to each other; this does not imply that the process of strategic planning occurs in individual companies exactly as described by a model." (Scott, 2007, pp.2/1)

The purpose of a strategic planning model is to act as a guideline or framework for either the analysis of existing strategic planning practices within an organisation or to aid the development of a strategic plan. A model identifies components to be considered, enables the classification of businesses in a taxonomy or may form an instrument of scientific enquiry (Baden-Fuller and Morgan, 2010).

As with other aspects of strategic management and planning theory, several authors have provided wide-ranging propositions for the strategic planning process ranging from a set of prescriptive steps a firm must follow (Scott, 1965; Dess, 1993; Lynch and Roche, 1997), to a broader process incorporating any action or attempt to provide the organisation with direction towards a future anticipated objective (White 1984; Waalewyn & Segaar 1996; Dansoh, 2005). In practice the process often lies between the two extremes.

Earlier models, associated with the rational or planned approach to the strategy process, tend to be prescriptive in nature. LeBreton and Henning (1960) for example outline a fourteen-step process commencing with becoming aware of the possible need to formulate a strategic plan to finally obtaining approval of the plan. Scott (1965) describes a five stage process (objectives, planning assumptions, seeking facts, evaluating alternatives and selecting a course of action), however like other prescriptive models it lacks the flexibility as well as continuous monitoring necessary as a consequence of operating within a changing environment. The model arguably is therefore more suitably applied in a stable, certain industry environment in organisations that have a planned approach to strategy formulation.

Shrivastava and Grant (1985) noted four models of strategic planning and decision making, namely:

- Managerial autocracy model
- Systemic bureaucracy model
- Adaptive planning model
- Political expediency model

(Shrivastava and Grant, 1985; pp. 103)

The models reflect differing levels of participation, time taken to make decisions as well as the payback period. Consequently the structure required to support each model differs from organisational types such as entrepreneurial, functional, divisional to conglomerate. The strength of this analysis is that numerous facets of strategic decision making are considered in recognition of differing management structures and organisational learning systems. However, it must be noted that the conclusions, although based on several (32) organisations in India, reflect a single strategic decision. How the strategic planning processes emerge when decisions are aggregated over a period of time is not addressed, nor is the monitoring and refinement of the process itself.

Dess (1993) goes some way to addressing the limitation of earlier models with the development of a four-part "cycle of activities" involving the identification of gaps resources needed to fill the gaps, distributing resources and monitoring. The strength the

process model centres on the inclusion of strategic context within which each of the activities must be considered, as well as the level (corporate, business or functional) where the activity takes place. The inclusion of monitoring within the process model is another notable strength.

There remains some scepticism, however, as to the use of strategic planning process models in practice. Mankins & Steel (2006) posit that executives make strategic decisions in an ad hoc fashion rather than following any serious analysis or productive debate, and ultimately:

"The traditional planning model is so cumbersome and out of sync with the way executives want and need to make decisions that top managers all too often sidestep the process when making their biggest strategic choices." (Makins and Steel, 2006, pp.78)

Stubbart (1985) identified the root cause of the problem in strategic planning models usage as follows:

"Strategic planning models have their roots in operations research models....Operations models deal with discrete, routine, highly-structured problems – such as setting optimum levels of inventory stocks based on completed knowledge of criteria...Strategic problems occur in complex social environments, they are periodic, unstructured and wicked." (Stubbart, 1985, pp.69)

The limitations highlighted by Stubbart (1985) are aligned with the incremental school of thought, particularly within an uncertain environment. Elements of the RBV are also evident with reference to the social environment.

Strategic planning models vary from prescriptive (therefore not flexible enough to for a changing environment) to excessively vague (therefore have limited practical use). What is required is a model that provides a framework of analysis of the key components of the strategic planning effort while permitting varying approaches to the

process. The EBS SPPM is one such model that incorporates these characteristics, and for thus has been selected for the purposes of the study at hand.

2.4.2 Edinburgh Business School Strategic Planning Process Model

As noted in the previous section, numerous strategic planning models have been developed over several decades, each with their own strengths and weaknesses. The EBS SPPM is one such model which was developed by Scott (2007), influenced by the work of Schendel (1994), for the purposes of the Edinburgh Business School (EBS) Master of Business Administration (MBA) as a framework within which to integrate and analyse the inter-relationships of core concepts of strategic management and planning (see appendix A). The process model is not intended to explain success or failure of an organisation, rather enable users of the model to apply strategic analysis to ascertain how robust the strategic planning process is within an organisation. The central contribution of the model, which has been used by in excess of 13,000 MBA graduates over two decades, is the conceptual approach which treats strategy as a process in which organisations are involved. The model centres around five key components and mechanisms for investigating and formulating each of the components are provided in the "Augmented Process Model" (see appendix B).

The EBS SPPM model integrates many aspects of several decades of strategic management research ultimately resulting in a systematic approach to the process (Ansoff, 1965) that permits flexibility in its development and implementation (Mintzberg, 1994; Grant, 2003). Despite providing a systematic approach to the process, the model is not prescriptive in its application and thus has potential for use across a wide array of industry sectors with varying levels of environmental uncertainty. This is one of the key advantages of this model over existing alternatives.

Whether an organisation undertakes a formal, structured approach resulting in a written strategic plan or whether the process is informal, unstructured and only considered by senior management, the model may be used as a framework of analysis within any organisation. Given the ongoing debate regarding the appropriateness and effectiveness of a formal strategic planning process in certain circumstances (see section 2.2.2), the

potential for the model to be used to assist either a formal or informal decision making process is a distinct benefit for the use of the model in practice.

The augmented process model draws from a vast body of existing, empirically tested research in the field to outline mechanisms, tools and techniques for addressing each component. The strategist may select relevant tools/techniques appropriate to their own organisation, industry and experience and once again the lack of prescription in this regard reinforces the potential use of the EBS SPPM across industry sectors, strategic types and varying approaches to the strategic planning process.

The emphasis placed on feedback throughout the strategic planning process mirrors the emergent approach championed by a number of prominent authors (Mintzberg, 1994; Hamel and Prahalad, 1996; Grant, 2003). The dynamic nature of the environment necessitates feedback throughout the implementation process therefore feedback must feature as a component of a strategic planning process model (Dyson and Foster, 1982; Shrivastava and Grant, 1985; Miller et al. 1988). Monitoring and feedback are an important part of organisational learning (Benes and Deipeveen, 1985; Mintzberg, 1994) and a component which often receives less commitment than strategic plan formation (Glaister and Falshaw, 1999). Thus their inclusion represents a significant strength of the EBS SPPM model, separating it from others in the field.

The EBS SPPM thus provides a systematic approach to the analysis of each of the core components of strategic planning, as well as a mechanism for emerging strategies to be incorporated through continuous feedback. Scope exists within the model for use of empirically tested techniques of strategic management, regardless of the approach to the strategic planning process, and in this regard does not attempt to reinvent the wheel. However, it's capability for use across any industry sector, within organisations adopting formal or informal strategy processes and with strategists of varying types gives potentially widespread application of the model in practice which is a distinct strength over similar process models.

The EBS SPPM model thus far has not been empirically tested outside of academia and has not been published in a refereed journal to date. While this is a possible limitation of the model it represents an opportunity to contribute to the body of knowledge in the field of strategic planning. The next number of sections provides an overview of the components of the EBS SPPM.

2.4.2.1 Who decides to do what?

This component brings together aspects of strategic management theory such approach to strategic planning (Mintzberg, 1994) as well as strategic type (Miles and Snow, 1978) which influence the entire strategic planning process. Previous sections have noted the relationship between strategic type and approach to plan formality, process formality and characteristics of the process including comprehensiveness, participation and flow (see section 2.3).

2.4.2.2 Analysis and Diagnosis

The second component of the EBS SPPM is the analysis of the business, industry and international environment, internal organisation as well as competitive environment within which the firm is operating. Cognisance of the environment within which the firm operates is of paramount importance to compete successfully, particularly within an uncertain environment (see section 2.2.3).

This component of the EBS SPPM offers potential to draw from a multitude of empirically tested research in the areas of industry analysis (Porter, 1980), environmental analysis (Eisenhardt, 1989; Grant, 2003; Brews and Purohit, 2007) and internal organisation analysis (Barney, 1991; Porter, 1985; Hamel and Prahalad, 1996; Løwendahl, 2005). The analysis and diagnosis of key trends in the variables, including macroeconomic indictors will inform the strategic decision making process, enable the firm to position itself within the industry context and ultimately impact strategic choice.

The need to consider these matters is not in itself a new phenomenon, however, the model provides a framework within which such analysis may take place. Some evidence suggests that competitor analysis, for example, does not receive considerable emphasis in practice (Boyd and Reuning-Elliott, 1998) nor do companies in practice utilise tools outlined in the augmented process model (Glaister and Falshaw, 1999; Stonehouse and

Pemberton, 2002; Frost, 2003). What the model provides is a structure within which these issues may be considered which may signal the need for further action or investigation.

2.4.2.3 Strategic Choice

The objectives of the organisation in light of environmental analysis ultimately inform strategic choice. The most prominent author in this regard is Porter (1980 and 1985) who revealed two tiers of strategy namely corporate level and business level. The choice of generic strategy may lead to a number of strategic variations including merger, acquisition, partnering or seeking opportunities abroad.

Without a doubt the choice of strategy will be influenced by the organisation objectives and analysis undertaken, thus demonstrating the linkages with previous stages of the EBS SPPM. It has also been established that the approach to strategy development (Mintzberg, 1994) and the organisation strategic type (Miles and Snow, 1978) play a particularly influential role in this regard.

The purpose of developing a strategy is to gain an advantage over competitors. Within an industry it is possible that several firms may broadly follow a similar strategy in pursuit of competitive advantage. The existence of strategic groups within an industry is commonplace (Desarbo e al., 2009; Leask & Parker, 2007; Short; 2007) with some research suggesting that performance of strategic groups will vary despite facing similar competitive and economic environmental conditions (Dikmen, 2009). The existence of "mobility barriers" are a contributory factor enabling strategic groups to persist (Leask & Parker, 2007), however, caution must be exercised as the existence and formation of strategic groups is subject to change.

2.4.2.4 Implementation

Implementation is a critical stage in the process which is rightly captured within the EBS SPPM and represents a key strength of the model. Structures, systems and processes must be established to ensure the success of the strategic plan alongside clear

communication of the organisation's strategy and adequate resourcing. If necessary, the strategic plan should be incorporated into performance targets and/or incentive schemes (where appropriate).

A crucial aspect of the above is to monitor the progress of strategic plan implementation and to ensure feedback mechanisms are in place to facilitate such monitoring. No attempt is made to provide a prescriptive methodology for strategy implementation within the EBS SPPM therefore it has potential to be used in any industry context and adapted as market forces demand.

Despite the obvious benefits of engaging in strategic planning many obstacles to the process continue to exist in its implementation. Issues such as time and budget constraints, market uncertainty, lack of experience (Uni, 1981) and resistance to change (Baker et al 1993; Dansoh 2005; Mintzberg, 1994) may negatively impact the implementation process. Dansoh (2005) emphasises the difficulty in predicting events with accuracy which limits the usefulness of strategic plans in management.

Obstacles noted by O'Regan and Ghobadian (2002) were experienced to a similar extent in independently-owned and subsidiary firms. Importantly, it was also concluded that firms without a formal planning process experienced each obstacle to a greater extent than those with a formal process.

2.5 Conclusion

Strategic planning is a well established field of study which has evolved over several decades (Ansoff, 1957; Pearce et al; 1987; Grant, 2003; Brews and Hunt, 2007). Despite its dominance in the strategic management literature, there remains no single definition of strategic planning. The lack of specificity has not stifled the research into either process or content aspects of strategic planning.

The multi-faceted nature of the discipline necessitates consideration of issues relating to the strategic planning approach (Mintzberg 1994; Miles and Snow, 1978), formality (O'Regan and Ghobadian, 2007), process characteristics (Papke-Shields et al., 2006) as well as the investigation of the relationship between strategic planning and performance

(Covin and Slevin, 1989). Consideration of these matters within a constantly changing and increasingly competitive environment augments the complexity of the discipline.

Characteristics of strategic planning will vary across organisations and will be influenced by the approach taken to the process as well as the organisational strategic type. Comparisons may be made along comprehensiveness, flow and participation dimensions resulting in varying levels of formality, use of strategic planning tools and covering various time horizons.

The multi-dimensional nature of the discipline has resulted in many strategic planning models being developed, all with their own strengths and weaknesses. What sets the EBS SPPM apart from other models, is its foundations in academia but potentially wide application in practice. The EBS SPPM provides a framework for analysis of existing strategic planning processes and the potential to act as a guideline for firms engaging in strategic planning for the first time. Crucially, the model allows for a systematic approach to the process with flexibility to modify the strategic plan in line with environmental and/or competitive variations. The augmented process model offers an array of empirically tested tools and techniques from which the strategist may chose to aid the process based on company-specific requirements.

Strategic planning has been investigated within many industry sectors (Robinson and Pearce, 1983; Fredrickson and Mitchell, 1984; Brews and Hunt, 1999; Grant, 2003), however, there remains a paucity of empirical investigation professional service firms (PSF's) including Quantity Surveying (QS) practices. The EBS SPPM provides a useful framework for the analysis of such organisations.

3. THE CONSTRUCTION INDUSTRY IN IRELAND

3.1 Introduction

Existing research in the field of strategic planning has tended to focus on manufacturing industries and there is a notable lack of investigation of the construction industry (Hillebrandt and Cannon, 1990), in particular professional service firms (PSF's) within it. This is particularly apparent when considered in an Irish context.

In Ireland the construction industry plays a sizeable role in the economy, in terms of contribution to Gross Domestic Product (GDP) and employment in particular, and is currently undergoing a period of significant change. The necessity for all construction-related firms, including Quantity Surveying (QS) practices to think and act strategically has never been more apparent, yet there remains a dearth of information as regards the type, scope, approach and characteristics of strategic decision making within these practices.

The purpose of this chapter is to outline the nature of the construction industry, the importance of the industry to the Irish economy and the role of the QS within the industry. Aspects of QS practice management and organisation are considered prior to a scrutiny of strategic planning processes specifically within QS practices being undertaken in the following chapter.

3.2 Characteristics of the Construction Industry and Role of QS

Construction is by nature, a hugely complex industry due to its investment intensity, the fragmented nature of the construction project team, the unique nature of the completed output on various sites around the country, knock-on effect on associated industries as well as increasingly demanding clients (Ashworth and Hogg, 2002; Hillebrandt and Cannon, 1994). The distinctive characteristics of the industry have implications for the strategic management of construction firms:

"Those characteristics go a long way to explain methods of production, and organisation, price determination, payment methods, financial decisions and controls, and an industrial structure unlike those met in other sectors. Hence the construction process generates management problems and opportunities at the level of the firm which may differ in scope, in scale, in time and in the type of appropriate solutions from those met in firms of other industries " (Hillebrandt and Cannon, 1994, pp. xviii)

The industry is categorised into four sectors, namely residential, private non residential, productive infrastructure and social infrastructure, each with its own sub-sectors (see appendix C). Demand for construction output may be from private clients, developers or the public sector, all with differing demands and methods of financing the investment. As a consequence of the investment intensity of construction output, the demand is potentially postponable and cyclical in nature, which contributes to the uncertainty prevailing in the industry (Hillebrandt, 2000; Ashworth and Hogg, 2002).

The output of the construction process may be characterised as a physical product or a one of several services involved in the process. The unique nature of the final product (building, road, bridge etc.) is clearly distinct from manufacturing where goods are similar and produced in a repetitive manner (Løwendahl, 2005).

A large number of construction trades, contractors and professionals are involved in the complex construction process combining a unique combination of resources for each project. The client's participation on the project team is a unique feature of the process, and a key influence in not only the project but strategic management process within construction firms (see section 3.5.1).

The role of the QS lies primarily in construction cost control with responsibility to ensure the project is kept within the agreed cost, time and quality criteria. The contracting organisation on the other hand undertakes building work in accordance with contract documents, and will have overall responsibility for site operations, as well as any subcontractors employed. It is important to note that the contractor may employ its own QS therefore the services of a Quantity Surveying (QS) practice would not be required. It is increasingly the case, however, whereby the financer of the project may request an independent assessment of costs and a QS practice may be employed for such (bank monitoring) purposes.

There are numerous factors which must be considered in the positioning of a QS practice in pursuit of competitive advantage. It is possible that a number of firms operating facing similar market forces will develop broadly similar strategies. The existence of "strategic groups" (Porter; 1980) and the mapping of said groups provides a useful frame of reference for an industry.

At present the limited investigation into QS practices in Ireland results in a lack of thorough investigation into the existence of strategic groups within the profession. Porter notes the importance of the relationship of a firm with its parent company in defining strategic group characteristics which may be relevant to QS practices in Ireland, given the number of practices linked to larger international organisations. Once again, there is currently a paucity of reliable information in this regard.

To lend support to the need to investigate QS practices, it is necessary to contextualise the industry in terms of its contribution to the Irish economy. The following section provides a snapshot of the key economic indicators in this regard.

3.3 Construction Industry and the Irish Economy

3.3.1 GDP, Employment and Inflation

Over several years the construction industry accounted for a considerable portion of GDP in Ireland. Since 2007, with the collapse of the US subprime market, the resulting global crises, the domestic banking crises and fiscal difficulties, the industry has suffered a rapid and severe contraction. Overreliance on construction, both in terms of creating employment and as a source of revenue for the government, has left the economy as a whole in dire circumstances, and the industry a shadow of its former self.

The construction industry is cyclical in nature, however, unlike other cyclical industries the divergence between peak and trough tends to be more intensified than for the economy as a whole. The affect of which is clearly visible from the table below.

	2006	2007	2008	2009	2010*
GNP (constant 2008 price €m)	153,398	160,299	154,672	138,161	136,089
% volume change in GNP	+6.5	+4.5	-3.5	-10.7	-1.5
Gross domestic fixed capital formation (GDFCF) (constant 2008 prices, €m)	45,193	46,456	39,806	27,482	21,106
Volume change in GDFCF (%)	+4.6	+2.8	-14.3	-31.0	-23.2
Total Construction Output					
Value of output (current prices €m)	38,631	38,601	32,593	18,048	11,733
Change in value of construction output	+14	+0	-16	-45	-35
Value of output (constant 2008 prices, €m)	34,838	35,057	32,593	20,646	14,540
Change in volume of construction output (%)	+10	+1	-7	-37	-30
Construction output as a % of GNP (output; current prices)	25.1	23.7	21.1	13.8	9.2

Table 3.1 Key Construction Industry Statistics

*Estimate

(Adapted from DKM Economic Consultants, Annual Construction Industry Review 2009 and Outlook 2010-2012; pp. i)

[Note: The Construction Industry Review and Outlook is commissioned and funded by the Department of the Environment Heritage and Local Government but is an independent report prepared by DKM Economic Consultants]

The table above displays the stark reality of the severity of the construction industry downturn. As is evident, 2006 represents the peak of the construction boom in terms of contribution to the economy, particularly proportion of GNP, with estimates for 2010 indicating a fall to approximately 9%. This may be considered to be a more sustainable

level of construction, however, the severity and speed from peak has had devastating consequences.

One of a number of consequences is the affect on employment. Employment in construction is categorised into two main components, namely direct (those working directly on site) and indirect employment (employment generated as a result of construction e.g. plant hire, construction transport and distribution and professional services). There is no specific mechanism for the estimation of indirect employment, however, it is considered to be approximately 40% of direct employment. The table below provides detail in this regard:

1 abit 5.2	Construction Employment	

Construction Employment

				_0101
268,000	264,000	216,000	137,000	110,000
108,000	105,000	87,000	54,000	44,000
376,000	369,000	303,000	191,000	154,000
	268,000 108,000 376,000	268,000264,000108,000105,000376,000369,000	268,000264,000216,000108,000105,00087,000376,000369,000303,000	268,000264,000216,000137,000108,000105,00087,00054,000376,000369,000303,000191,000

(DKM, 2010)

Table 2.1

The construction industry comprises a number of professionals included in the "indirect employment" statistics above. The most recent study undertaken by FAS (the National Training and Employment Authority in Ireland), entitled "The National Skills Bulletin, 2010" recorded approximately 29,000 construction professionals employed in the final quarter of 2009 (representing 1.9% of national employment). An overview of construction professionals is indicated in the following table:

Table 3.3Construction Professionals

	No.	% of total
Civil/mining engineers	9,500	33
Architects and town planners	6,700	23
Building Managers	5,300	18
Quantity Surveyors	3,100	11
Draughtspersons	1,700	6
Building etc. Surveyors	1,500	5
Archit. & Civil Eng. Technology	1,400	4
Total		100

(adapted from "National Skills Bulleting 2010", Fas, pp. 88)

The table above provides an indication as to the importance of the QS profession within the industry, representing over 10% of construction professionals.

The downturn in the industry has resulted in intense competition for considerably fewer projects being undertaken. Construction tender price deflation of approximately 17% was recorded in 2009 with a further decline in tender prices of between 5-10% estimated for 2010 (DKM, 2010; pp. 14). These figures are in contrast to the price of materials, capital and labour which together reduced by -1.1% in 2009 with an increase of 1% forecast for 2010 (DKM, 2010; pp. 13). The trend in tender prices (prices charged to clients) versus construction costs are of serious concern and are signalling below cost tendering. This situation will inevitably result in company closures and is wholly unsustainable.

A forum for the analysis of European construction industries, Euroconstruct, was established in 1975 and currently has members from 19 European countries. The purpose for establishing the body was to conduct research, analysis and provide forecasts for the construction industry across member states. The latest data from Euroconstruct (Euroconstruct Conference Proceedings, Dublin, June 2010) predict a slow and prolonged path to recovery within the Irish construction industry, a position similar to that in Spain and Portugal.

The key risks to the projections include:

- a further shock to the global economy
- a "W-shaped" recession following a too early withdrawal of stimulus interventions
- the possibility that, as public expenditure cuts start to bite, the hoped-for growth in private investment does not materialise.
- a risk that public sector cuts may need to be greater than planned
- an increase in interest rates before the construction industry upturn becomes fully established
- a reluctance on behalf of the private sector to participate in PPP projects due to economic uncertainty
- difficulties raising the matching finance for infrastructure projects in Central and Eastern Europe

(Adapted from Euroconstruct Conference proceedings, Dublin 2010; pp.139)

In Ireland's case, the full cost of the banking crises has transpired to be well in excess of original estimates which has hampered the government's ability to meet capital spending targets on infrastructure. This has important implications for future demand for construction demand stemming from the public sector.

3.4 The Quantity Surveying Profession in Ireland

3.4.1 Evolution of the Quantity Surveying Profession

As noted previously the construction industry is comprised of a number of professional practices and contracting firms. The Quantity Surveying (QS) profession is a relatively new profession, having developed in the 20th century. The role of the QS has evolved over time, and continues to do so. Many high profile reports have been published regarding the change in the construction industry (Latham, 1994; Egan, 1998) influenced the evolution. Numerous other functions of the QS have developed to include:

- investment appraisal
- advice on cost limits and budgets
- whole life costing
- value management
- risk analysis
- insolvency services
- subcontract administrator
- cost engineering services
- environmental services measurement and costing
- planning and supervision
- evaluation for insurance purpose
- project management
- facilities Management

(Adapted Ashworth and Hogg, 2002, pp.9)

Historically the lowest price tendered for a project was likely to win a project however, value management for the client as well as environmental sustainability has become a major priority, in some instances at a higher cost. A consequence of this is a requirement for a more integrated project team (IPT), involving client participation at a early stage in the process. The QS will seek to manage the supply chain as efficiently and effectively as possible, to minimise waste and provide value to the client based on their needs in the use of the final building. This has been underpinned by changes in legislation, building regulations and directives. The use of IT has changed the way in which a QS performs their function. Computed Aided Design (CAD) and billing software (such as BillSoft®) have also created efficiencies in the construction process.

In Ireland, a key development within the profession has been the advent of the recent Capital Works Management Framework (CWMF) for public sector projects. The CWMF is a structure developed to deliver public sector procurement reform, resulting in a suite of best practice guidelines, standard contracts and generic template documents which together form four pillars of the framework (namely public works contracts, standard conditions of engagement, cost planning and control and guidance notes). Public sector construction projects under the new framework are awarded on a fixedpriced basis thus transferring risk onto the contractor. The full realisation of the effect of such a contract has not been apparent given recent construction price deflation and few projects have been completed using the new contract.

As is evident, the current and potential role of the QS is considerable. The profession has become the second largest division in the membership of its professional body. The Society of Chartered Surveyors (SCS), which is broadly in line with the UK counterpart, the Royal Institution of Chartered Surveyors (RICS). QS's are employed by contractor organisations, public sector and in private practice.

3.4.2 Education, Training and Employment

The RICS in "*The Core Skills and Knowledge Base of the Quantity Surveyor*" (1992) (<u>www.rics.org</u>) note the knowledge and expertise of a practicing QS should include:

- construction technology
- measurement rules and conventions
- construction economics
- financial management
- business administration
- construction law

While the key skill base needs to include management, documentation, analysis, appraisal, quantification, synthesis and communication, Nkado and Meyer (2001) add information technology and "professional practice" to the list of skills required to undertake the role. The QS is increasingly employed as the lead consultant on many large projects due to the diverse range of skills and expertise held by the profession (Hore et al. ,2009).

In Ireland there are a number of third level institutes offering degree programmes in Quantity Surveying/Construction Economics at both level 7 and level 8 under the National Qualifications Authority of Ireland (NQAI) framework. Several programmes are accredited by the SCS. Education and training does not, however, cease on

graduation from a third level institution. The SCS provide a number of routes to further educate and train members.

A QS is deemed to be "chartered" on successful completion of the Assessment of Professional Competence (APC), through the SCS. At least fifty percent of company directors must be chartered for a QS practice to be awarded chartered status. Continuous Professional Development (CPD) is also encouraged in recognition of the need for lifelong learning within the profession. The benefits of engaging in CPD (Ashworth & Hogg, 2002) are strongly noted by Maister (2003) who proclaims that:

"What you know now and are able to do now, what your current success is built on, will unavoidably depreciate in value unless you actively work on learning new things and building new skills. Continual professional development is a lifelong requirement, not an option." (Maister, 2003, pp. 147)

Changes within the industry and the profession are presenting opportunities for qualified QS's to diversify the range of services provided (see section 3.7.1), thus reinforcing the need for further training and continuing professional development.

3.4.3 Role of the Professional Body – The Society of Chartered Surveyors

In Ireland, the professional body representing the QS profession is the Society of Chartered Surveyors (SCS), which maintains a close link to the Royal Institution of Chartered Surveyors (RICS) in the UK. Currently there are in excess of 2000 full members of the SCS plus 1000 undergoing APC. The Society's main function is to regulate the profession from education/qualification through to CPD.

The SCS also provide expert advice on built environment matters; make submissions to government on behalf of the profession, formulate the professional code of conduct, publish bi-annual cost indices and provide information to members via the website and quarterly magazine.

In June 2010 members of the SCS and the Irish Auctioneers and Valuer's Institute (IAVI) voted in favour of merging institutions into a new single professional body representing members of each institution. The new institution will be named the Society of Chartered Surveyors Ireland (SCSI) and will be operational from 1st March 2011.

3.5 Quantity Surveying Practice Organisation and Management

A QS may be employed in the public sector, by contractor organisations or in private practice. The focus of the study at hand is centred on Quantity Surveying (QS) practices.

The management of a professional service firm (PSF), such as a QS practice, displays many differing characteristics to that of a manufacturing organisation and thus warrants detailed consideration. A crucial difference between the two lies in the fact that manufacturing firms have routine and standardised operations for the production of a product within a repetitive process. The work of a PSF is client-specific and a highly customised service (Løwendahl, 2005).

The next number of sections outlines some of the key considerations in the management of a QS practice.

3.5.1 The Professional Service Firm (PSF)

The term "profession" is synonymous with the presence of specialist knowledge in a particular area of competence (Ashworth and Hogg, 2002; Maister, 2003; Grönroos, 2007), a large amount of customisation hence a high degree of interaction with the client (Greenwood et al., 2005). All of this is undertaken within an ethical framework and professional code of conduct (Løwendahl, 2005).

Professional services differ from manufacturing in that they are produced and consumed simultaneously and the customer participates in the "production" process (Grönroos, 2007). This has important implications for the PSF as the client will evaluate the service provision throughout the life of the project rather than solely on its completion (Hoxley

et al., 2007). A professional service provider must therefore instil a confidence in the client not only for the technical expertise and knowledge they possess, but in all aspects of the service provision. Maister (2003) notes that

"...while goods are consumed, services are experienced. The professional service provider is (or should be) as much in the business of managing the client's experience with respect to professional services as in the business of executing technical tasks." (Maister, 2003, pp.71)

Maintaining positive client relationships is hugely important to the firm in securing the reputation upon which repeat business depends. Maister (2003) prompts us to recall that the client often engages a professional firm in times of "significant uncertainty, importance and risk. It is this atmosphere of risk and importance that makes them prepared to pay the traditionally high fees of the professional service sector" pp. 73.

3.5.2 Human Resources

Central to the success of a PSF are the people employed in it (Greenwood et al., 2005). Human resource management (HRM) is therefore an important component in the management of a QS practice. Ultimately, the three requirements a client demands in engaging a professional service firm are human resource issues, namely expertise, experience and efficiency, however, the priority given to each by the client will vary (Maister, 2003). The need for customisation of service requires face-to-face interaction with the client and therefore "*the core of the resources base of the professional service firm thus resides in the professionals employed and their ability to solve whatever problems the clients may want them to solve*" (Løwendahl, 2005; pp. 45). The emphasis on internal resources of the PSF is clearly aligned with the RBV of strategic management (Barney, 1991; Teece et al., 1997) outlined in section 2.2.

3.5.3 Marketing

Marketing a PSF differs to that of a manufacturing organisation given that production and consumption overlap in the former. It is also possible that no tangible product is created at the end of the process to use as a marketing tool or exemplar of work completed (Barrett, 1993). An architectural practice will have a portfolio of previous designs, however, not so in the case of the QS. Furthermore, within the construction industry particularly, no two units of output are identical therefore there is a level of uncertainty in the ex-ante evaluation of the service quality (Løwendahl, 2005).

Maintaining high quality service provision positively affects the firm's reputation, which is a crucial aspect of professional practice marketing strategy. It has been said that:

"To succeed, PSF's must generate a superior reputation. The importance of reputation, moreover, permeates core strategic decisions, bounding the appropriate form of diversification." (Greenwood et al., 2005, pp. 661)

In the study of accounting practices undertaken by Greenwood et al. (2005) a definite link between reputation and firm performance was discovered, thus highlighting the importance of maintaining a high-quality reputation. Maintaining the reputation may afford the company the opportunity of repeat business.

It has been noted that the most probable source of new business stems from existing clients (Maister, 2003), and as previously noted, firm reputation is a critical aspect of the marketing function. Word of mouth is one way in which a firms reputation is communicated, however, how reputation is managed, enhanced and used as a strategic tool remains unclear (Davies et al, 2010). Ironically, it is believed that considerably more resources are invested in pursuit of new clients than in attempting to engage existing clients in new business (Maister, 2003).

Murugavarothayan (2000) proposes a conceptual model of criteria for the evaluation of profession services practices. The model divides the process quality (tangibles, reliability, responsiveness, empathy and assurance) from outcome quality (function,

cost, time, and quality). The relative importance of each component to clients was tested by means of a survey of 94 construction professional service clients. Based on the survey the top five dimensions that clients consider important are:

- 1. reliability (ability to perform the promised service accurately)
- 2. assurance (knowledge and courtesy of employees and their ability to inspire trust and confidence)
- 3. time (certainty of programme, project completed on time, early commencement, early notification of delay)
- 4. empathy (caring, individualised attention to customers)
- 5. responsiveness (willingness to help customers and provide prompt service)

If it is the case that clients measure PSF performance based on the five criteria above, it is prudent to be aware of these issues throughout the course of undertaking the project, and build a reputation based on achieving them.

3.5.4 Quality Assurance

Quality assurance and the marketing function are inextricably linked with the QS practice. Delivery of a quality service, within budget and time constraints, are the three components necessary for the successful delivery of a project. The perceived quality of service provision is subjective. Good service quality occurs when the experienced quality meets the expectations of the customer (Grönroos, 2007). This is particularly important given the involvement of the customer throughout the process.

A client employs a PSF to fill a gap in their knowledge in a particular area, and is willing to pay for such knowledge and experience. The knowledge gap between the client and QS must be managed. The wider the gap "*the more pedagogical the service provider must be*" and the greater degree of trust in the professional required (Løwendahl, 2005). Membership of the relevant profession body will instil confidence in the PSF so too may independent quality certification.
Quality assurance can be certified by the International Organization for Standardization (ISO) as a measure of the quality of service provision by the firm. Importantly, for ISO certification firms must satisfy a number of criteria including the clear articulation of a business plan (which is subject to an annual review) and soliciting client feedback. The process for gaining certification is necessarily laborious requiring the documentation of every aspect of the PSF. This may act as a deterrent to seeking certification.

3.5.5 Finance

Company finances are obviously important to the QS practice. Given the current downturn in the construction industry in Ireland, it is difficult to accurately estimate the financial situation of professional practices. Turnover has undoubtedly declined, and the number of corporate failures is disproportionate in construction, with reports of c. 40% of all corporate failures in the first six months of 2010 arising from construction (DKM, 2010; pp.62).

Access to finance is increasingly difficult with Irish banks imposing stricter lending conditions, and the Financial Regulator compounding the situation with the implementation of tighter control mechanisms. Although the measurement of the financial performance of QS practices would undoubtedly be beneficial, there is a significant gap in terms of the information currently available in this regard.

Furthermore any assessment of the financial performance of QS practices in the current study would only serve to present a very exceptional perspective rather than the norm. Consequently the examination of financial wellbeing, as well as company financial performance as a relationship to strategic planning remains outside the scope of the investigation.

3.6 Conclusions

The preceding analysis has detailed the nature and importance of the construction industry in Ireland. The QS plays a fundamental role within the process with ultimate responsibility for construction project cost control (amongst other functions). QS's may be employed in the public sector, in contracting organisations or by QS practices. The focus of the study at hand relates to QS practices, therefore particular attention is given to aspects of QS practice management.

Construction by its nature is complex for many reasons, and differs greatly from manufacturing on a number of levels, not least of which is lack of uniformity of final output. The strategic management of construction firms will therefore necessarily differ from that of manufacturing. This is clearly evident when considering professional service firms (PSF's) such as QS practices.

The construction industry in Ireland is undergoing a period of enormous change, following unprecedented growth over a prolonged period of time. The need for QS practices to think and act strategically has never been more apparent. Chapter 2 provided a comprehensive analysis of strategic management and planning practices and demonstrated the importance of engaging in the process, be it on a formal or informal basis. This chapter has proven the importance of the construction industry and the role of the QS within it. The following chapter investigates strategic planning processes in QS practices.

4. STRATEGIC PLANNING IN QUANTITY SURVEYING PRACTICES

4.1 Introduction

The previous chapter demonstrated the importance of the construction industry in Ireland and the role of the QS profession within it. At a macro level, comprehensive research is undertaken to ascertain trends and forecasts in the industry, however, the micro-level investigation lags significantly behind.

Strategic planning is an important aspect of any business regardless of industry structure, organisation size or nature of the business, and the importance of the discipline is reflected in a plethora of research spanning several decades (Ansoff, 1957, 1970; Porter, 1979; Steiner, 1979; Pearce et al. 1987). While general principles of strategic planning may apply in any industry (Desarbo, 2005; Gibbons and O'Connor, 2005; Brews and Purohit, 2009) there remains a notable lack of focus on construction (Hillebrandt, 1994; Venegas & Alarcon, 1997) and specifically on the QS profession (Jennings and Betts, 1996).

The limited research undertaken within the construction industry is heavily biased towards large contracting organisations (Betts and Ofori, 1992; Hillebrandt and Cannon, 1994; Dansoh, 2005; Green et al., 2008) and while it is reasonable to assume that such investigations are applicable to the industry as a whole, it is important to ascertain how the principles apply in practice to QS firms.

This chapter focuses on strategic planning within the QS profession. Aspects of the process are aligned with general strategic planning theory such that a comparison may be made with the existing literature base and its application to the profession. As noted previously, a large proportion of the existing body of research in construction is related to contracting firms, much of which is equally applicable to professional service firms (PSF's). Where differences occur, or are likely to occur, they are highlighted and discussed where applicable.

4.2 Strategic Planning in the Quantity Surveying Profession

Determining the extent of strategic planning in the construction industry is limited, as the focus of strategic planning research concentrates on manufacturing rather than construction (Hillebrandt, 1994; Venegas & Alarcon, 1997). Research undertaken in construction is heavily weighted in favour of large contracting organisations (Betts and Ofori, 1992; Hillebrandt and Cannon, 1994; Dansoh, 2005; Green et al., 2008) rather than the professions, including QS practices (Jennings and Betts, 1996). Additionally, significant emphasis is placed on project planning and management within construction rather than organisational planning:

"The scant empirical attention devoted to investigating the performance determinants at the firm level is in stark contrast to the heavy focus given to the investigation of the performance determinants at the project level." (Phua, 2006, pp.309)

Limited as construction strategic planning research is, there are opposing views regarding the prevalence of strategic planning within the industry. On the one hand Betts and Offori (1992) posit that:

"One can observe a significant and dominant structural shift in the emphasis of business planning from the tactical to the strategic. There is also a shift from internal concerns with such narrow performance criteria as ROI to external concerns with such concepts as value and competition in global markets." (Betts and Offori, 1992, pp.513)

On the other hand Hillebrandt and Cannon (1994) argue that there is a lack of strategic planning in construction firms because it is not necessary to plan for large investments in fixed assets.

The lack of consensus as to the degree, extent and formality of strategic planning in the industry necessitates further investigation. The profession must be analysed along the dimensions outlined in chapter 2, including approach to strategic planning, organisational type and planning characteristics. This will facilitate the transition from a

theoretical discussion of existing strategic planning literature to its practical application within QS practices.

4.2.1 Construction Industry Environment and Strategic Planning

The importance of environmental analysis has been established (see section 2.2.3) within the context of numerous industry settings (Fredrickson and Mitchell, 1984; Eisenhardt, 1989; Grant, 2003). Examination of the construction industry environmental analysis is equally important (Lansley, 1983; Warszawski, 1996).

Within the construction industry, and hence the QS profession, the macroeconomic environment is exceedingly important with factors such as inflation, employment, interest rates, government programmes and general demand influencing strategic decision making (refer to Chapter 3). The dependence on government in construction is perhaps more pronounced in construction than other industries as the government is an significant client of the industry. The cyclical nature of the industry requires QS practices to keep abreast of trends and to remain flexible and responsive in their strategic approach under such conditions (Lansley, 1983).

QS practices must also work within the legal and regulatory framework governing the profession. Such legal and regulatory influences may stem from government, EU and international agreements as well as from accreditation from professional bodies including the SCS, RICS and ISO. The impact of these forces cannot be ignored in the development of strategy given the opportunities or constraints such factors may present.

Internal analysis is especially important for PSF's (see section 4.4.), and much of the analysis of the internal organisation for such a practice bears resemblance to the RBV of the firm. Phua (2006) notes the importance of industry and internal environments by declaring that:

"...the endowment of unique resources of firms coupled with the existence of certain rules, regulations and norms that are present within the industry determine the kinds of strategies that firms pursue to achieve optimal economic performance." (Phua, 2006, pp. 310)

Despite the importance of examining the relationship between the construction firm and its environment "*little research has been undertaken to examine the extent of empirical association between the two.*" (Phua, 2007;, pp.753). The complexity of the construction industry and fragmented nature of the strategic management discipline, as noted previously, may explain the deficit. One study of contracting firms goes some way to sealing the gap (Lansley, 1987), demonstrating the implications of a changing environment on strategy, skills, structure and style. Three periods of time were investigated by Lansley (1987) and key findings are summarised in the table below.

1 able 4.1 Effect of environmental change over tim	Table 4.1	ct of environmental change ov	ver time
---	-----------	-------------------------------	----------

Characteristic	1960's	1970's	1980's
Environmental Change	Operational	Strategic	Competitive
Critical Success Factors			
Strategy	Specialization & efficiency	Broadening and survival	Selection and efficiency
Skills	Technical	Organizational	Extra- Organizational
Structure	Tight control	Looser control; boundary regulation	Tighter control; boundary regulation
Styles	Task	People; Corporate	Task Corporate
*Miles & Snow	Defender	Prospector	Analyser

Lansley, (1987); pp. 154

*not contained in table in article, but discussed in the text.

From the table above it is evident that changes in strategy, structure and style evolved in a similar pattern to that of strategic management theory (see section 2.2), and the drive towards efficiency resulting from intense competition requires an analytical approach to strategy formation. Incorporating Miles and Snow (1978) typologies in such a way demonstrates how the environment influences the approach to decision making, which as noted in section 2.3, will influence strategic planning process characteristics.

The role of the strategist is thus of paramount importance (Phua, 2007). The decision maker(s) will be faced with numerous sources of data with potential implications to the firm including inflation, interest rates or unemployment trends etc. The importance placed on each indicator by the decision maker, will ultimately shape strategy formulation. It is likely that it will vary from one firm to the next operating within the same sector. Junnonen (1998) goes one step further by asserting that:

"...environmental uncertainty, turbulence and understanding are not the primary driving force of strategy development and formation. These are simply the part of the cognitive structure of the organisation's managers that becomes the context for making strategic decision." (Junnonen, 1998, pp.111)

The assertion above is important in determining a cause and effect relationship between the environment and strategy. It is argued that an uncertain environment will not cause a practice to develop strategy, rather the environment forms part of the context within which strategic decisions are reached.

There remains little doubt of the importance of environmental analysis for any firm. Excluding environmental analysis from strategic decision making is potentially a grave mistake. Lansley (1987) claims that "...*the failure of a firm to approximate its structure and processes to its environment will lead it to become uncompetitive*" (pp,142).

Given the complex, multi-faceted nature of strategic planning and environmental analysis, models have been developed to assist in the process of environmental analysis (Porter, 1980) and strategic planning (Lynch and Roche, 1997) alike. As noted previously, many of the models have been empirically tested in various industry settings yet there remains inadequate testing within the construction industry, and in particular QS practices.

The following section expands on previous analysis to ascertain the context in which strategic management theory (and models) may be applied to the construction industry and QS profession.

4.2.2 Application of Strategic Planning Theory in Construction

The practical application of traditional strategic management and planning theory in practice within the construction industry is disputed (Langford and Male, 2001).

In a study of large UK contracting organisations, Price et al. (2003) presented Chandler's (1962) ten step model to participants in their survey and drew interesting conclusions including:

- few organisations state that environmental scanning should be done before setting objectives or even before formulating a mission
- 25% state that they did not formulate a mission
- 30% did not establish performance indicators
- 20% stated that they do not identify critical success factors. (pp. 353)

The findings demonstrate that a notable proportion of construction firms do not follow trends in traditional strategic planning practice, such as the development of a mission or performance indicators. Despite the focus centring on contracting firms, it is reasonable to assume the same applies to PSF's within the industry. In the same study it was also concluded that little emphasis is placed on competitor analysis when compared to customer analysis, with the most common ratio being 2:8 (pp. 356). One explanation provided was that:

"Most construction organizations try to build good relationships with customers in order to ensure repeat work, which is considered as a primary source of all work, rather than "over-competing" with rivals to attract their customers." (Price et al., 2003, pp. 357)

Competitor analysis is regarded as an important component of the strategic planning process (Scott, 2007) and has featured in many highly regarded models of strategic management (Porter, 1980). The lack of competitor analysis evident in construction represents a significant divergence from strategic planning research in other industries, and one which requires further empirical testing. The emphasis on the customer stems from the unique nature of the output and clearly takes priority over competitor analysis.

In the case of QS practices it is unsurprising that emphasis is placed on client relationships for the reasons noted above. However, why both client and competitors are not considered requires further investigation.

As noted previously, a large proportion of strategic planning literature concentrates on manufacturing with proportionately less on construction (Hillebrandt and Cannon, 1994; Venegas & Alarcon, 1997). Given the findings above, it is clear that strategic planning in construction differs along a number of dimensions to that of manufacturing. Prior to examining these dimensions (see section 4.4) one may ask whether traditional strategic management and planning theory be applied within the construction industry?

The answer of course is that strategic planning and management models can be applied to research undertaken in the context of the construction industry. Betts and Offori (1992) demonstrate this in their scrutiny of the industry applying Porter's (1980) five forces model to the industry. The analysis provided valuable insight into the industry. However, the extent to which such models are used by firms in practice is doubtful when they concluded that *"there is little evidence of construction enterprises systematically examining these five forces and their relevance to the strategic planning and management of their firms."* (pp.516). The conclusion drawn by Betts and Offori (1992) demonstrates that the use of such models may remain in academia and certainly in the case of construction, are not rigorously applied in practice.

Furthermore, in PSF's such as QS practices the potential application of strategic planning models may be even more limited. This view is made explicit and extended to incorporate Porter's (1985) value chain analysis when Løwendahl (2005) with reference to PSF's comments:

"...the value chain (Porter, 1985) is difficult, if not impossible, to adapt to a firm where no linear production process with input, transformation, and output exits. Similarly a portfolio analysis with emphasis on market shares and market growth seems meaningless for a firm which is used to changing in size and competitor ranking with every large project added or completed." (Løwendahl, 2005, pp. 33) All is not lost, however, since the emergence on the RBV. The RBV of strategy provides a considerable contribution to the analysis of strategic planning within PSF's across any industry sector. As outlined in section 2.2, the RBV proclaims that the competitive advantage of an organisation is based on the internal resources and competencies of a firm which are heterogeneous, and often intangible. Traditional strategic planning models are replaced with an emergent approach to strategy formation, centering on the capabilities of the firm in developing value creating strategies (Barney, 1991; Teece et al., 1997).

The RBV is particularly relevant to PSF's in the management of intangible resources such as "*reputation*" (Løwendahl, 2005). The management of reputation is of paramount importance to QS practices, whereupon the future of the practice is largely dependent upon repeat business and client referrals. How this features in the characteristics of the strategic planning and decision making process are considered in the following section.

4.3 Strategic Planning Characteristics in Quantity Surveying Practices

Previous sections (see section 2.3) have analysed and discussed strategic planning characteristics as they relate to a varied range of industry sectors. The next number of sections analyse the characteristics as they apply in QS practices.

4.3.1 Strategic Planning Formality and Approach

As is evident from section 2.2.2, the degree of strategic planning formality will differ from one organisation to the next and may have implications on organisational performance.

Company size plays an influential role in the type, scope and extent of strategic planning undertaken in construction firms. Large firms are more likely to engage in the process (Bausman, 2002). Price et al. (2003) revealed that 75% of large construction

firms and 50% of medium sized firms had formal strategic planning processes, proclaiming that:

"...as an organisation grows its business processes and environment become more sophisticated; it becomes increasingly important that the direction is clear, and that it strives for the same purpose and in the same way." (Price et al., 2003, pp. 353)

The process is also more likely to be documented in the former than the latter and the aim of the process was to clarify company goals and policy which is consistent with findings of Larsen et al. (1998). Furthermore, construction firms which are subsidiaries of international firms are more likely to engage in formal strategic planning, and despite the conclusion being drawn from contracting organisations, professional practices are likely to follow this pattern (Dansoh, 2005).

A very significant contribution from the comprehensive study undertaken by Hillebrandt and Cannon (1994) is that many construction firms undertake strategic planning without realising it. Company principals are thinking and acting strategically, however, often do not formalise strategic planning into a distinct, structured process. Strategic decisions are made continuously but are not necessarily formally recorded or form part of a systematic decision making procedure.

Strategic planning in construction thus bears resemblance to the emergent approach to strategy formation (Wilkinson, 2003; Green et al., 2008). This is particularly apparent in PSF's when it has been noted that:

"...an evolution without any explicit overall plan or strategy seems to be the rule in professional business service firms, rather than the exception. In a sense, then, proactive strategic management is often more or less neglected in professional service firms, at least in the early years." (Løwendahl, 2005, pp. 76)

In professional practices, the tailoring of services provided and interaction with the client necessitates a great degree of judgement on the part of the professional. Therefore

professional practices must have flexibility built into the strategy to ensure that client needs are met by the professionals responsible for the project (Løwendahl, 2005). The need for flexibility in strategic planning is strongly linked to the emergent approach to strategy formation (Mintzberg, 1994) and is particularly relevant to PSF's. Client involvement in the process of service delivery results in heterogeneity of services provided by the practice, thus corporate strategy must remain sufficiently flexible to allow for a varied client base as well as movement across construction industry sectors.

What remains unknown, particularly in an Irish context, thus requires further investigation, is the extent to which strategists consider aspects of traditional models of strategic planning, even unbeknown to themselves. Are components such as industry and competitive analysis undertaken, even informally? Does the QS practice have specified objectives or a clear vision of where they would like the practice to be in the future? These are key considerations for any organisation and must not be neglected despite the informal, unstructured approach to strategic decision making.

The emergent approach, and likelihood of an informal strategic planning process has implications on other characteristics of the strategic-decision making process. Strategic planning flow and participation in the process are considered in the following section.

4.3.2 Strategic Planning Flow and Participation

Participation in the strategic planning process is an important characteristic for analysis (as outlined in section 2.3.2). The degree of participation may influence the sense of ownership and implementation of a strategic plan (Maister, 2003). Once again, there is scant evidence of the degree of participation in QS practice strategic planning.

The limited available research in the industry suggests that strategic planning initiatives have a top-down flow with limited company-wide participation (Price et al., 2003; Dansoh, 2005). It has been argued, however, that professional practices require a consultative strategic decision making process rather than a top-down approach, in order to increase the ownership and thus effectiveness of the process (Maister, 2003; Wilkinson, 2003).

This view is shared by Løwendahl (2005), who contends that a consensus-based process is required based on the premise that in PSF's strategy must be sufficiently flexible to allow for independent judgement to be made by professionals within the organisation in order to meet individual client needs. Without participation in the process, the strategy may not be disseminated clearly amongst staff dealing directly with clients. Although the investigation was not originally specific to the construction industry, the unique nature of construction professional service provision and increasingly demanding client undoubtedly requires the suggested flexibility.

Participation of clients in the decision making process is another factor for consideration, particularly for practices with ISO certification. Client feedback is part of the quality assurance process whereby feedback may be solicited on completion of a project, this may happen on an exception basis only.

4.3.3 Strategic Planning Tools

The complexity of the strategic planning process has led to the development of tools to facilitate the process. Section 2.3.3 investigated the extent of strategic tool usage and it was concluded that in practice, tools were not often used and thus remained in the realms of academia. While the discussion incorporated many industry sectors, the construction industry has yet to be analysed in this regard.

Evidence suggests that construction companies do not favour the use of strategic planning tools (Price et al., 2003). Two possible explanations were put forward, namely a shortfall in the necessary skills for use of strategic management tools and a lack of recognition of the benefits of tool usage. Furthermore, it was noted that the low level of response to the study from small organisations was because:

"...these organizations may have had little knowledge of key strategic issues and do not employ strategic planning tools and techniques...Their main aim was to survive and continue in operation with an acceptable level of profitability." (Price et al., 2003, pp. 352)

The lack of tool usage within construction firms presents a constraint on the development of strategic decision making and perhaps even a wasted opportunity of QS practices. Tools such as scenario planning provide a useful mechanism for determining alternative courses of action within an uncertain environment (Benes and Diepeveen, 1985), and have potential for use within QS practices, both at strategic as well as project level. The benefits of scenario planning are noted by Langford and Male (2001) whom declare that:

"A scenario attempts to portray the future shape of the construction world and how the company can adapt to fit it. In a sense this is corporate ecology – the firm evolves to suit the environment in which it lives." (Langford and Male, 2001, pp. 167)

The potential benefit of strategic planning is apparent, however, what is also notable is the necessity for the firm to "evolve" to suit the environment thus reinforcing the need for an emergent and flexible approach to strategy formation (Mintzberg, 1994). The emergent approach, as noted previously, is more likely to result in an informal strategic decision making process in construction (Larson et al., 1998).

It is important to remember that the education and training of a QS in quantitative analysis, economics and project management equips them with the skills to carry out such analysis, therefore it is surprising that it is not prevalent in practice (Price et al., 2003).

4.3.4 Strategic Planning Time Horizon

Section 2.3.4 discussed the variations of strategic planning time horizon as an important dimension of the process. Strategic planning within the construction industry tends to focus on a relatively short time horizon of up to five years (Price et al., 2003; Dansoh; 2005) which is broadly consistent with findings from previous research undertaken in other industries (Brews and Hunt, 1999; Stonehouse and Pemberton, 2002).

The short time horizon is not unusual given the uncertainty in the construction industry environment, as noted by Benes and Diepeveen (1985):

"In a situation of uncertainty the firm's planning should be based upon a shorter period, its capacity upon minimum workload and management should look for quite new solutions. This all leads to a need for flexible planning." (Benes and Diepeveen, 1985, pp.29)

Interestingly, however, Price et al. (2003) discovered that although the majority of firms engage in the strategic process annually, nearly a quarter of those with a time frame of less than three years intended to increase their strategic planning horizon. However, ISO certification is awarded on an annual basis therefore it is usual to undertake planning within that timeframe.

4.4 Strategic Choice for Quantity Surveying Practices

The process of strategic planning, whether undertaken on a formal or informal basis using a planned or emergent approach, ultimately requires making choices between alternative strategies. The use of strategic planning models and tools have been developed to assist in the process, however, within construction firms and hence QS practices, the models are limited in their practical application (Price et al., 2003).

Teece et al. (1997) noted the importance of considering past performance, and the "path" that the organisation has travelled in determining future strategy. Similarly the history of the firm was also noted by Mintzberg (1994) as being an important factor in the decision making process. This is particular important to a QS practice that depends on repeat business and the strength of its reputation for future business (Maister, 2003).

It is within the context of a changing environment, intense competition and influence on the history of the organisation that strategic choice is considered.

4.4.1 Corporate Level Strategy

Corporate strategy is discussed in section 2.4.2.3 using Porters (1980) framework of analysis. Limited data is available regarding the corporate strategy of construction firms, indeed much of the research in the area focuses on business level strategy (Hillebrandt and Cannon, 1989; Betts and Ofori, 1992; Price et al., 2003).

At the time of writing, no data is available regarding the corporate strategies being pursued by QS practices in Ireland. Clearly there is a necessity to undertake such an investigation.

Anecdotal evidence suggests that QS practices are ultimately pursuing a "survival" strategy given the economic downturn however, there is no empirical evidence to support this.

Benes and Diepeveen (1985) posit that in times of greater uncertainty with fewer opportunities for growth, firms must be able to plan for stability or even a regression strategy. It is argued that even with the necessary flexibility incorporated into the strategic plan it should cover no more than a year. They also outline a model for corporate planning in construction firms, however, do not provide evidence of it's use in practice.

4.4.2 Business Level Strategy

Business level strategy, as defined by Porter (1980) has received proportionately greater attention than corporate level within a construction context (Hillebrandt and Cannon, 1989; Betts and Ofori, 1992; Jennings and Betts, 1996; Price et al., 2003). The challenge of applying traditional strategic planning models to construction firms has been noted (section 4.2.2), however, the use of Porter's (1980) generic strategies does not pose a similar challenge. On the contrary, the generic strategy framework is extremely useful in the analysis of construction firm behaviour and has practical application to contracting and PSF's alike.

Cost leadership is prevalent in the construction industry particularly with the change in tendering methods. The intensity of price competition is clearly visible in light of tender price indices, which in part is influenced by the competitive tendering process for public sector projects in Ireland. The potential for cost leadership in QS practices lies in creating internal efficiencies and to "*optimise the level of resources*" to reduce the cost base (Jennings and Betts, 1996; pp.176). Cost leadership is more suitable to an industry whereby there is little room for differentiation (Warszawski, 1996) therefore is clearly not the only option available to a construction firm where the products and services provided are heterogeneous. In some instances it is the change in procurement methods that have resulted in construction firms concentrating on a differentiation strategy (Price et al., 2003).

Differentiation occurs when a firm tries to make the product/service more appealing to the customer than the competition thereby potentially commanding a higher price. Many mechanisms for product differentiation in construction contractors are outlined in the work of Hillebrandt et al. (1995) and Warszawski (1996) and for QS practices in the work of Jennings and Betts (1996) which are discussed below.

A focus strategy involves applying either of the aforementioned strategies to a particular segment within an industry in order to develop a specialisation. Examples of which for QS practices may include focusing on a specific sector (e.g. private non-residential construction), geographical area or developing a specialist expertise (e.g. sustainable construction). This strategy is particularly appropriate to a smaller firm (Warszawski, 1996). Caution must be expressed with a focused strategy based on specialist expertise, as it may be imitable over time therefore may not provide a sustainable competitive advantage (Barney, 1991).

Porter (1980) warns of the potential dangers of becoming "*stuck in the middle*" (see section 2.2.2).

Possibly the most significant investigation of QS business strategy is contained in the work of Jennings and Betts (1996) which went a considerable distance in addressing the dearth of research in the profession within the UK. The application of Porter's (1980) generic strategy to QS practices was the aim of the investigation undertaken. The results

of a survey of senior management discovered that 60% of QS practices were pursuing a differentiation strategy, 32% a focus strategy and 8% cost leadership (pp.169). This has important implications, and the authors contend that:

"...the QS profession considers clients to be influenced more by the quality of the service they receive than the price they pay for it. Such a low preference for cost leadership may reflect the profession's dissatisfaction with the fee tendering systems (since the abolition of fee scales), which has reduced the fees charged to the bare minimum, and they therefore feel that there is no long term competitive benefit in trying to reduce fees any further." (Jennings and Betts, 1996, pp. 174)

Further testing of the data revealed that the size of QS practice impacted on the aspect of differentiation being followed, with larger practices (over 20 QS's) competing using their reputation and smaller practices (fewer than 5 QS) on repeat business with a *"faithful group of clients"* (a view that is shared by Hillebrandt et al, 1995; Green et al, 2008). Medium sized practices (6-20 QS) tended to have a diverse strategy choice ultimately differentiating through the range of service provision. However, regardless of how the practices differentiate from competitors, the most important element across the board was deemed to be *"punctuality of service, identifying the client's requirements and maintaining good relationships."* (pp. 175). This supports the work of Maister (2003) and Løwendahl (2005) in other PSF's and is crucial in enhancing the QS practice reputation, upon which repeat business depends (see section 3.5.3).

Greenwood et al. (2005) identify two common strategies amongst PSF's, namely "boutique" (which resembles focus strategy) or diversified. The former occurs when the firm provides a narrow range of services and specialises in these services, and the latter resulting in a multidisciplinary practice (MDP) which offers a number of "*competitive benefits*" for the client as well as employees. At present there is no evidence to determine the extent to which QS practices in Ireland pursue either strategy

The findings outlined above provide considerable insight into a previously underinvestigated area and clearly demonstrate the prevalence of differentiation over other generic strategies. However, Jennings and Betts (1996) concluded that: "The results of this study indicate that Porter's generic strategies may generally be suitable for use by QS practices in strategy formulation; however the unpopularity of cost leadership indicates that Porter's work does not provide a balanced definition of competitive strategy for QS practices." (Jennings and Betts, 1996, pp.176)

Jennings and Betts (1996) combine elements of Porter's (1980) generic strategies with Maister's (2003) approach to strategy and produce a model of competitive strategies for QS practices. The model defines strategy on the basis of the client base (be it specific or varied) and level of service (simple or complex) which then forms the basis of four generic strategies for QS practices, namely:

- 1. Execution: varied client base and simple level of service
- 2. Expertise: varied client base and complex level of service
- 3. Efficiency: specific client base and simple level of service
- 4. Experience: specific client base and complex level of service

(adapted from Jennings and Betts, 1996, pp.178)

The strength of the model lies in the fact that it was developed specifically for QS practices, however, it lacks discussion of the mechanisms by which a practice may identify which quadrant(s) it is currently in or indeed which it should strive towards. Furthermore, there is no correlation considered between each quadrant vis-à-vis profitability. The issues resulting in lack of strategic planning tool usage previously identified (see section 2.3.3 and 4.3.3) are likely to be equally applicable to this model.

Another important study of strategic planning QS practices was undertaken by Boon (1996, 2001, 2008) whom tracked changes in strategic management of QS practices in Auckland, New Zealand over several years. A five-point framework was developed to assess survival strategies of participants which identify the key factors to be addressed by management of QS practices, including:

- 1. market orientation
- 2. knowledge base

- 3. flexibility in cost structure and productive capacity
- 4. efficiency and price competitiveness
- 5. financial resources

The initial study discovered that during the construction downturn the majority of practices downsized, new business came from repeat clients, practices expanded the range of services provided and a general tightening up of budgeting and control process. The experience of the downturn resulted in practices taking steps to protect the firm against the financial impact of another downturn, the most notable of which included competing on price, offering a value added service and provision of additional services (Boon, 1996). These practices competed on price and value added concurrently in order to survive, therefore were classified as being "stuck in the middle".

In subsequent years, the participants were re-interviewed using the original framework. Changes within the five components of the framework were investigated and presented at RICS Cobra conferences in 2001 and 2008. A summary is contained in Appendix D, however, the most notable long term changes may be summarised as follows:

- dominant firms had international linkages
- return to core QS business
- more formalised quality assurance (yet only 1 ISO certified)
- increased emphasis on staff training and mentoring
- half of the practice competing on price, half on the basis of value-added

The key limitation of the study is that it is based on a sample of only four "significant" QS practices therefore the findings may not be generalisable across all QS firms. Despite this limitation the results provide invaluable insight into the changing strategy of QS practices over time, which are largely related to changes in the environment within which they are operating.

4.5 Monitoring and Implementation

The changing nature of the construction industry environment (section 4.2.1) necessitates continuous monitoring to determine the best way to respond to shifting circumstances. To this end Maister (2003) notes that:

"The strategy development task is not an exercise in forecasting. Rather, the goal is to create the responsive organization; to put in place a set of operating procedures which force the practitioners to listen to their marketplace on an ongoing basis, not as a one-time piece of market research..." (Maister, 2003, pp. 230)

In practice, this advice may be easier said than done. Many hindrances to strategic planning are clearly evident (see section 2.4.2.4). In construction, lack of knowledge of the core concepts related to strategic planning is a key internal factor (Hillebrandt and Cannon, 1990) which must be addressed. The fragmented nature of the industry, the unique nature of each project (Betts and Ofori, 1992) and changing market demand further impose challenges to the process. Venegas and Alarcon (1997) in their study of the strategic decision making process for construction firms noted that:

"...the inability of construction executives to collect all relevant information pertaining to their decisions makes this process notoriously complex, novel, uncertain, open and endless." (Venegas and Alarcon, 1997, pp.388)

QS practices must retain organisational flexibility (Lansley, 1983) as well as flexibility in the strategic planning process (Benes and Diepeveen, 1985) to cope with these demands. Despite the challenges, QS practices have considerable expertise (and experience in many cases) in project management, many of which could be applied at the strategic level in order to overcome the challenges and sustain competitive advantage.

4.6 Conclusion

Strategic planning is clearly an important function of any business, and the construction industry is clearly an important industry for any economy. The question therefore becomes: why do construction firms neglect strategic planning given its potential benefits and why does this continue to be the case given the importance of the industry to the global economy?

It may be the case that the purpose and benefits of strategic planning are not known by the strategist, and hence it is not undertaken (Dansoh 2005). Or perhaps it is being undertaken and the strategist is not aware that the decisions they make and process by which they are reached is in fact strategic planning (Hillebrandt and Cannon, 1990). Consequently, the type, scope, extent and purpose of strategic planning in the industry remains unclear. The available research suggests that strategic planning, when undertaken, involves an informal top-down process with an emergent approach.

Reviewing the research it is apparent that the strategic planning process is complex, particular for firms within the construction industry. A paucity of research relating to the strategic management of professional services within the industry, specifically QS practices is evident and compounded by a complete lack of empirical scrutiny of Irish QS practices. At the time of writing not a single empirical study was available. Clearly there is a need to undertake an investigation as to the type, scope and extent of strategic planning within the sector.

5. LITERATURE SYNTHESIS

5.1 Introduction

Extensive research has been undertaken in the field of strategic management with numerous contributions from several authors over a number of decades (Ansoff, 1957; Hamel and Prahalad, 1996; Grant, 2001 and 2003; Mintzberg, 1994; Porter, 1980, 1985 and 1996; Steiner, 1979). It is assumed that an organisation seeks to achieve and maintain advantage over competitors (Porter, 1985) and strategic planning is instrumental in such endeavours. The link between strategic planning and performance continues to be debated and there remains a lack of consensus in this regard (Boyd and Reuning-Elliott, 1998). Clearly strategic planning has potential benefits to the organisation in shaping its future direction, however, no single definition of precisely what strategic planning entails is universally agreed upon (Chafee, 1985)

What cannot be disputed is that strategic planning is a multi-dimensional discipline which may be examined in terms of formulation or implementation, process or content. The research at hand focuses on strategic planning processes specifically.

The strategic planning process will vary from one organisation to the next and concentration on certain aspects of the process will also differ depending on the industry sector (Langley, 1988; Pearce et. al 1987, 1988; Grant, 2003). The degree of formality of the process has been well documented (O'Regan and Ghobadian, 2007), in particular as it relates to organisational performance (Fredrickson, 1984; Pearce, Robins and Robinson, 1987). However, there are a number of alternative approaches to strategic planning which are equally valid (Mintzberg, 1994) and in certain circumstances, such as a turbulent environment, are more applicable in pursuit of competitive advantage (Grant 2003). The approach to the process will be largely dependent upon the strategist and a taxonomy of strategic organisational behaviour is presented by Miles and Snow (1978) to facilitate analysis in this regard.

Due to varying approaches to strategic planning, characteristics of the process will also differ from one organisation to the next. Areas such as flow, participation, comprehensiveness, tools, intensity and planning horizon (Fredrickson and Mitchell, 1984; Glaister and Falshaw, 1999; Papke-Shields et al. 2006) are attributes which may be examined to provide insight into features of the process.

The environment within which a firm operates will clearly influence the extent and scope of strategic planning undertaken, and it has been argued that the need for strategic planning increases with the degree of environmental uncertainty (Eisenhardt, 1989; Grant, 2003). Industry environment, competition and the firm's position within this context will also be critical in deciding organisational strategy (Porter, 1980). Whether the process is formal or informal and whether the strategy is written down or otherwise, elements such as these must be considered.

The challenges in embarking on such a complex, multi-faceted process have been aided by the development of a number of strategic planning tools (Frost, 2003) and process models (Scott, 2007), however, the extent to which such tools and techniques are used in practice remains uncertain (Stonehouse and Pemberton, 2002).

To date the concentration of existing research lies within manufacturing while there remains a scarcity of empirical investigation of strategic planning processes in the construction industry (Green et al., 2008), and in particular QS practices. Construction is a hugely important industry for the global economy and is by nature complex and fragmented (Ashworth and Hogg, 2002; Hillebrandt and Cannon, 1994). There remains limited empirical research undertaken within the industry and what has been undertaken has focused primarily on contracting organisations (Dansoh, 2005; Green et al., 2008; Price et al., 2003) and seldom on professional service firms (PSF's) including QS practices.

When research has been conducted to examine strategic planning in QS practices it has been conducted in the context of practices in Hong Kong (Phua, 2006), the UK (Jennings and Betts, 1996) and New Zealand (Boon, 1996, 2001, 2008), and to date no empirical study investigating the strategic planning practices within Irish QS firms is known to exist. In Ireland the construction industry was the key driver of growth over several years, and now the industry is suffering most significantly in the downturn. The need for strategic planning in QS practices in Ireland has never been more apparent, yet little is known about the type, scope, extent and key considerations influencing the strategic planning processes of such firms.

There are mixed findings relating to the applicability of general strategic management theory to the industry (Betts and Ofori, 1992; Price, 2003) and to professional service practice (Maister, 2003; Løwendahl, 2005). The purpose of this exploratory study is to investigate the type, scope and extent of strategic planning in Irish QS practices by using existing strategic planning theory and the EBS SPPM to aid the analysis.

5.2 Merging Two Strands of Investigation

The preceding literature review has established the importance of strategic planning to an organisation and the significance of the construction industry to the economy as a whole. The literature also demonstrates the multi-dimensional nature of strategic planning and the fragmented characteristics of construction. Consequently investigating strategic planning within the industry is both challenging and complex.

Strategic planning and management research in construction has concentrated on contracting firms, however, they may display characteristics which are not applicable to PSF's. It is thus not appropriate to discuss the industry in its entirety, with a "one size fits all approach" in the knowledge that it is multi-faceted and fragmented. That being said, elements of the strategic planning process are undoubtedly relevant across firms, particularly the affect of the macroeconomic environment.

The service sector is growing in importance to the Irish economy and thus supports the necessity to investigate professional services within this important industry. As has been noted, there remains limited empirical investigation of the strategic planning processes of such firms. This research seeks to address the apparent gap in existing knowledge by investigating strategic planning within QS practices. Although the study focuses on QS practices elements may be equally applicable to other professional practices within the industry such as architecture or engineering firms.

5.3 Defining Measurement Criteria

This exploratory study in a multi-dimensional discipline requires the selection of criteria against which the subject matter is analysed. The discipline of strategic management and planning may stretch across an immense range of related disciplines so it is necessary to define the scope relevant to investigating the process of strategic planning. The EBS SPPM acts as a framework within which key areas of the discipline relevant to the investigation are considered.

Organisational Objectives and Strategist

The basis of analysis for this component lies in organisation types identified by Miles and Snow (1978) and approaches to strategy noted by Mintzberg (1994)

Environmental Analysis

The work of Porter (1980) plays a significant role in analysing the industry and competitive environment within which a firm operates. The uncertain nature of the profession has an impact on the strategic planning process, therefore this component draws on research by Eisenhardt (1989) and Grant (2003) amongst others.

Internal organisational analysis is an important facet of the strategic planning process, and is particularly pertinent for PSF's. Two key experts in the field of strategic management and planning within service firms are Maister (2003) and Løwendahl, (2005), whose invaluable contribution shapes analysis in this area.

Strategic Choice

Although several authors have investigated the strategic choices available to firms the most prominent is undoubtedly Porter (1980) who noted four alternative corporate and business level strategies available to a firm. These strategies are the yardstick against which QS strategies are measured.

Strategic Planning Characteristics

A number of distinct characteristics are apparent in the strategic planning process, which vary from one organisation to the next. Once again the potential scope is enormous in this regard therefore the selected characteristics and the authors that dominate research for the particular characteristic under investigation for this research are noted in the following table:

Characteristic	Author(s)
Comprehensiveness	Fredrickson and Mitchell (1984); Eisenhardt (1989);
	Segars and Grover, (1999); Papke-Shields et al.
	(2006),
Flow	Baker et al. (1993); Segars and Grover, (1999);
	Papke-Shields et al. (2006),
Participation	Dyson and Foster (1982); Segars and Grover, (1999);
	Collier et al. (2004); Papke-Shields et al. (2006),
Planning Horizon	Glaister and Falshaw (1999); Koufopoulous (2000);
	Stonehouse and Pemberton (2002); Papke-Shields et
	al. (2006),
Formal/Written Plan	O'Regan & Ghobadian (2007)
Strategic Planning Tools	Frost (2003); Stonehouse and Pemberton (2002)

Table 5.1Strategic planning characteristic and relevant author(s)

Implementation

Strategic planning is a continuous process and the importance of communication and feedback has been noted in chapter 2. Challenges may be encountered in the implementation of the strategic plan (regardless of whether it is formally written or not) hence the obstacles noted by Mintzberg (1994), Baker et al. (1993) and Dansoh (2005) particularly, inform the analysis in this regard.

The requirements under ISO certification incorporate feedback processes which are particularly relevant to QS practices.

5.4 Conclusion

The gap in existing knowledge in the strategic planning processes of Irish QS practices has been clearly established. The research at hand is an exploratory study the foundations of which are entrenched in strategic planning literature, therefore parameters must be established to define the scope of the investigation. The previous section has outlined dimensions of strategic planning to be addressed in the study.

Strategic planning is a complex discipline and the process will be influenced by numerous, often firm-specific considerations. As a consequence, the primary research method for this exploratory study is qualitative in nature in order to discover not only the type, scope and extent of strategic planning but to gain insight into the influences and factors shaping the process itself.

The qualitative component is subsequently supported by a quantitative phase which involves a widespread survey of QS practices in Ireland.

The mixed methods approach to the investigation not only provides statistical data against which characteristics of the process may be measured, but also insight into how and why decisions are made regarding the strategy and strategic planning processes within QS practices in Ireland. The following chapter provides further detail regarding the chosen methodology.

6. RESEARCH METHODOLOGY

6.1 Introduction

The preceding chapters provided a comprehensive examination of existing literature in the field of strategic planning in the Quantity Surveying (QS) profession. A significant gap in the knowledge has been identified with regard to the strategic planning and management of QS practices, specifically in Ireland. This chapter outlines the methodology employed in the investigation to tackle this disparity and to address the stated research question, aims and objectives.

A flexible design using mixed methods provided the optimum approach given the exploratory nature of the study. The research was thus undertaken in two phases. The predominant method was qualitative in nature and took the form of semi-structured interviews with the managing director/partner of eleven QS practices. The second phase involved a large scale online survey of all QS practices registered with the SCS, for which a response rate of over 40% was achieved. Triangulation was achieved through the use of the two methods and the inclusion of phase one participants in the second phase.

This chapter provides an outline of the methodologies chosen to address the research question, aims and objectives.

6.2 Research Aim

Following a thorough investigation of the literature in the field a gap in the knowledge was identified regarding strategic planning process within Irish QS practices. This formed the basis of the research question, aims and objectives.

The research question is:

"To what extent are systematic strategic planning processes evident within Irish Quantity Surveying practices?"

The aim of the research is:

"To investigate current and recent strategic planning processes evident within Irish Quantity Surveying practices."

The research has the following objectives:

- to investigate current and recent trends in strategic planning in Irish QS professional practices
- 2. to evaluate existing strategic planning practises within the Irish QS profession against the strategic management literature
- to develop a set of guidelines to assist systematic strategic planning in Irish QS practices

As determined in the literature review, the Edinburgh Business School Strategic Planning Process Model (EBS SPPM) provides a practical framework for the evaluation and analysis of the strategic planning processes within QS practices. The research design is loosely based on the process model structure, while incorporating further elements from relevant strategic planning literature.

6.3 Research Paradigm and Theoretical Framework

Two major research paradigms exist in the field of business research, namely positivism and phenomenology. This study follows the usual social science methodological assumptions about these research paradigms, and both are evident within the study.

The choice of research method must be appropriate to the study at hand. In terms of strategic management research, the appropriateness of methodologies has changed in line with the evolution of the disciple itself. A comprehensive overview of strategic management research and methodology is provided by Hoskisson et al (1999) whom

refer to the choice of methodology "*swings of a pendulum*". The dominant methodologies used over four periods is summarised as follows:

Decade	Examples of key authors	Primary emphasis of research	Primary Method
1960's	Ansoff, Chandler	Primarily internal	Qualitative
1970's	Porter	Primarily external	Quantitative
1980's	Williamson	Primarily internal	Mixed
1990's	Barney, Conner	Resource based view	Mixed

 Table 6.1
 Evolution of primary research methods in strategic management

Abridged from Hoskisson et al. (1999)

The table above purports the use of mixed methods when the emphasis of research is centred on the RBV of the firm. Any investigation of the strategic management of PSF's is heavily influenced by the RBV of the firm (Løwendahl, 2005), consequently mixed methods are appropriate for such an investigation. The literature review contained in previous chapters demonstrates the dearth of research into strategic planning practices in construction firms (Phua, 2006) particularly in the context of PSF's such as QS practices in Ireland. This research seeks to address the existing gap in this regard and is thus exploratory in nature. The use of mixed research methods is appropriate in such circumstances to gain insight into the strategic planning processes of QS practices, as well as to gain deeper insight into factors which influence the process and ultimately how decision are made in this regard.

The phenomenological philosophy is apparent in the initial phase of the research which sought to discover the extent of strategic planning and provide an understanding of the strategic planning process within QS practices. This phase involved in-depth semi-structured interviews with the principal of eleven QS practices, and provided significant insight into the type, scope, extent of strategic planning as well as the process by which strategy was formulated.

The positivist approach is evident primarily in the second phase of research, whereby a widespread survey was undertaken of QS practices registered with the SCS. Once again,

the type, scope and extent of strategic planning was identified and the identification of relationships between variables was possible.

Triangulation was used in the study as a strategy to ensure against possible threats to validity and in this study the primary mechanism of triangulation was methodological triangulation (Robson, 2002).

The research was deductive in nature, in that specific aspects of strategic management and planning theory were investigated within the QS profession as a whole. Deduction is noted to be:

"...a form of inference that purports to be conclusive – the conclusion must necessarily follow from the reasons given. These reasons are said to imply the conclusion and to represent a proof. There is a much stronger and different bond between reasons and conclusions than is found with induction." (Cooper and Schindler, 1998; pp.30)

The combined research findings provide empirical data results rather than a conceptual framework of analysis.

6.4 Research Design

6.4.1 Phase 1: Semi-Structured Interview

The nature of the subject matter necessitated the involvement of senior management/principal with input into the strategic planning and management of participating QS practices. To gain a full understanding of the numerous facets of strategic planning processes, interviews were deemed the most appropriate mechanism by which to collect the data. Interviews are commonly known to provide a deep understanding of the phenomena being investigated, assist in the portrayal of multiple views (Gubrium & Holstein, 2002) and are particularly useful where the investigation is exploratory in nature (King, 1994).

A number of techniques exist for conducting interviews (Robson, 2002) and a semistructured interview approach was applied for the purposes of this investigation. A set of questions (or Interview Prompt Sheet) was developed in advance of the interviews addressing the key dimensions and characteristics of the strategic planning process identified in the literature (see appendix E). The semi-structured interviews ensured a clear focus to each meeting while ensuring flexibility in how issues were addressed. Considerable scope remained to probe respondents for fuller explanations when required which is one of the overriding advantages in the use of this method (Bryman and Bell, 2007).

The interview prompt sheet formed the basis of each interview thus comparison between cases was made possible. Questions were not necessarily asked in the same order in each interview, rather depended on the breadth of response provided thus matters were often discussed out of sequence. Each component was ultimately addressed in every interview.

The interview prompt sheet contained five sections which are outlined in the next section.

6.4.1.1 Semi-Structured Interview Structure

Section 1: General Company Information

Interviews commenced with questions relating to company demographics which were easily answered and served to establish a rapport with the respondent. Company size, number of years in business, range of services provided, sectors serviced, number of employees and changes within each of the aforementioned were addressed.

At times respondents discussed strategies pertaining to changes in these issues over time and the process by which such decisions were made. Respondents were encouraged to speak fully on the topics as and when they arose, rather than sticking rigidly to the interview prompt sheet sequence. The second group of questions centred on organisational objectives. What they are, how had they changed, who was involved in the development of objectives and how did the company seek to achieve the objectives? The strategist type and approach to strategy were also addressed.

The existence of a written strategic plan was identified in this set of questions. Where a written plan was confirmed, respondents were probed as to the driver for formulating the plan, process for plan formulation, participation, time horizon, content, use of strategic planning tools and other matters as they arose (e.g. influence of head office where practice formed part of a larger international construction cost consultancy).

Section 3: Environmental Analysis

The extent of analysis of the industry, competitive, macroeconomic and internal environments undertaken by QS practices is unknown. This section sought to identify the comprehensiveness of such analysis within the chosen sample.

Questions relating to critical factors of the environment affecting the practice were put forward. Changes and drivers of change were queried and the processes by which the organisation gathered and communicated relevant information in this regard were solicited.

Section 4: Strategic Choice

The next series of questions centred on the corporate and business level strategies pursued by respondents, both at the time of interview and recent past. Drivers of change and challenges in pursuit of the chosen strategy were identified and addressed.

Section 5: Implementing Strategy

The final section of the semi-structured interview related to the implementation of strategy. Matters surrounding challenges in implementation, communication and feedback were considered.

Respondents were provided with the opportunity to raise any other issues they deemed relevant to the discussion prior to concluding the interview. The confidentiality of responses was assured, and respondents were informed that the written transcript of the interview would be forwarded on completion for their approval prior to any analysis being undertaken. It was also agreed that an overview of the key findings would be forwarded on completion purposes, which informants were invited to provide feedback on. This is an important tactic for testing and confirming findings (Miles and Huberman, 1994).

6.4.2 Phase 2: Questionnaire

Following the initial phase of research a meeting was held with the SCS who expressed an interest in extending the research to incorporate a widespread survey of all members of the QS division of the Society. A research proposal was submitted to the SCS Research Committee and a modest amount of funding was secured subject to presenting findings of the survey in a Continuous Professional Development (CPD) event as well as publishing an article for the SCS magazine (see Appendix F for a copy of the article).

The questionnaire was designed to cover similar issues as those addressed in the semistructured interviews undertaken in Phase 1. Questions were informed by the findings from the initial phase as well as existing literature in the field. Some terminology that could be perceived by practitioners as being ambiguous was reworded. The decisions not to replicate verbatim from existing studies, as well as shortening the question length were significant contributions of pilot study feedback (see section 7.3).

The questionnaire comprised 31 questions in 3 sections, including open ended, multiple choice and Likert-scale questions. Likert scale questions remained consistent

throughout the survey and consisted of five answer choices, where one was always the negative and five the positive end of the scale. The navigational guide was also consistent throughout the survey, a simple style and symmetry was employed plus a dark print for questions and light print for answers was used to make the response task easier (Dillman, 2000).

The following section outlines the structure of the online survey (which is contained in Appendix H).

6.4.2.1 Survey Structure

Section 1: General company information

The first number of questions sought demographic information regarding the QS practice. The questionnaire was designed such that the questions were applicable to all and were straight forward to answer (Dillman, 2000). Questions asked regarded company size, number of years in business, ownership structure, ISO certification, sectors serviced and services provided. Changes in service provision, sectors serviced and numbers employed between peak and trough over the current construction cycle were also recorded.

Similar demographic information has been obtained in other major studies in the field of strategic management (Koufopoulos and Chryssochoidis, 2000; O'Regan and Ghobadian, 2002) and is particularly important for the comparative analysis of strategic planning practices between cases.

Information gained from this section alone represents a sizeable contribution to knowledge given that it, at the time of writing, it was not available through any other source.
Section 2: Strategic Management Practices

This section lies at the heart of strategic planning and management literature. The purpose of this section was to ascertain the corporate and business strategies (as identified by Porter, 1980) currently being pursued as well as reflecting changes in this regard from construction cycle peak to trough.

Questions were posed regarding strategic type (Miles and Snow, 2003), risk attitude (Covin and Slevin, 1989), extent of environmental analysis undertaken (Phua, 2006) and communication of strategy (Brews and Hunt, 1999). A further question regarding the influence of lifestyle choices was included as it became apparent from phase 1 that this may influence strategic decision making in some instances.

Once again, this section provides a sizeable contribution to knowledge for reasons previously stated.

Section 3: Strategic Planning and Decision Making Process

The purpose of this section was to identify they extent, type and scope of strategic planning in Irish QS practices. This section was largely derived from existing studies in the field. Characteristics of strategic planning (Papke-Shields, 2006), experience with strategic planning and formality of strategic planning process (Baker et al., 1993; O'Regan and Ghobadian, 2002) were addressed.

When a company confirmed the existence of a formal/written strategic plan, further questions regarding content (Larsen, 2000), plan horizon (Brews and Hunt, 1999), difficulties in formulation, strategic plan use and obstacles to implementation were posed (Baker et al., 1993). It was possible for respondents with no written plan to exit the survey at that point, however, it was also possible to answer the remaining questions if they desired. This produced very valuable information regarding the strategic decision making process of practices without a formal/written strategic plan.

Although a number of questions were influenced by existing empirical studies within the field of strategic management, additions were made where relevant to the QS profession, and the wording of questions was often amended for clarity.

As is evident, matters pertaining to company finance were not addressed in the survey. The strategic planning/performance relationship is clearly an important one, however, remains outside the scope of the research. It was considered prudent to "*minimize requests to obtain personal information*" (Dillman, 2000) to encourage a greater response.

Furthermore the role and effectiveness of the professional body were not addressed in the survey. This was a strategic decision by the researcher as the timing of the survey coincided with a controversial merger proposition (which has subsequently been approved) between the SCS and the IAVI (Irish Auctioneers and Valuers Institute) which may have negatively influenced responses in this regard. The pilot test of the qualitative phase had highlighted the lack of involvement of the SCS in the strategic management of QS practices, therefore the inclusion of such questions would serve to increase the number of questions without contributing significantly to the research.

Notwithstanding the two conscious exclusions from the survey, the responses provided a substantial contribution to knowledge within the field. At the time of writing, no demographic information was available regarding Irish QS practices and no empirical investigation had taken place as to the type, scope and extent of strategic planning in Irish QS practices. The contribution to knowledge is thus apparent.

6.5 Sample Design

6.5.1 Phase 1: Qualitative Data

A selective sample of eleven QS practices were chosen based on access to the CEO/Managing Partner in practices of varying sizes.

Classifications of small, medium and large firms differ from one source to the next. Price et al. (2003) classify firm size based on number of employees and turnover however, the bands provided are not appropriate to professional practices. Nkado and Meyer (2001) classify QS practices as being large if they employ over 20 people and small if it employs fewer which is arguably too broad to be meaningful. Jennings and Betts (1996) are very specific in their survey of QS practices providing respondents with 8 options relating to number of employees (it was revealed that 55% of respondents employed between 3-10 people).

For the purposes of the first phase of research a middle ground was found between the aforementioned studies and practices were classified based on three bands of employment as follows:

Company Size	Number of QS Practices
0-10	2
11-20	4
>20	5
Total	11

Table 6.2	Phase one	practices and	number of	employees

Participants were initially contacted by telephone to outline the purpose of the study and to secure participation. Interviews were scheduled to take place at a date and time of their choosing and a letter of confirmation was subsequently forwarded to each participant.

6.5.2 Phase 2: Quantitative Data

A list of QS practices registered with the QS Division of the SCS, together with the company contact details was provided by the Society. In order to ensure the accuracy of the list and to avoid coverage error (Denscombe, 2002), each respondent on the list was investigated. The nature of the research required the participation of a key informant from the senior management team, where possible the CEO or managing partner. A

research assistant was temporarily employed to contact each company by telephone in order to:

- 1. establish if the practice was operational
- 2. outline briefly the purpose of the study
- 3. gain confirmation, in principle, of the company's participation
- determine the direct contact details of the key informant i.e. CEO/senior partner (i.e. strategist)
- 5. provide notice that an e-mail with survey link would be forwarded
- 6. ensure confidentiality of responses

A number of discrepancies were uncovered in the data provided by the SCS. For example, a number of contracting organisations were included in the list, therefore they were removed as the study concentrated on QS practices. A small number of practices were no longer operating due to the principal(s) having retired. There were also a number of companies included in the initial list which despite offering cost consultancy as part of their services, were not QS practices and were thus removed.

It became evident that there were a number of errors in contact details provided by the SCS, which is a cause for concern given the contact details were those advertised in the *"find a surveyor"* section of the SCS website. The researcher undertook a comprehensive investigation of each anomaly to ascertain the correct contact details, through searches on the internet and local business directories. In a large number of cases the correct details were discovered and included for data collection purposes.

In some instances telephone numbers were no longer in service, and/or website decommissioned. It is assumed that given the number of reported construction company closures, that these practices are no longer operational and were hence removed from the sample for data collection purposes. This is an important assumption underlying the sample therefore these practices were investigated once again by the researcher a number of months after the initial attempt with the same conclusion drawn in each case.

The final aspect in refining the sample was to identify those practices that did not wish to participate in the study. For various reasons, primarily company policies, a small number of practices were excluded from the final sample upon their request.

The following tables outline the survey population, sample size and response rate:

Tabl	e	6.3	Sı	ırv	vev	P	or	ul	latio	n
	-		\sim •		~		v r			

	Number of QS Practices
List received from SCS	285
Double entries (i.e. regional offices)	43
Listed but not QS Practice	39
Survey Population	203

Table 6.4Sample Size and Response Rate

	No. of QS Practices
Closed	4
Assumed Closed	20
Undeliverable	6
Did not wish to participate	23
Final Sample Size	150
Total Number of Responses	62
Response Rate	41%

Phase one participants are included in the total above. This served to triangulate findings from the initial phase, and also to record changes within these practices given the time lag between phase one and two.

The necessity to answer each question was not imposed on respondents (key contribution of pilot study. See chapter 7). In some instances the response rate is marginally lower than above. Where this is the case, it is noted during the course of analysis (chapter 9).

6.6 Conclusion

Mixed methods were employed for the purposes of this exploratory investigation to address the stated research objectives. The initial phase comprised semi-structured interviews with the director/partner of eleven QS practices and the second phase a widespread online survey of QS practices registered with the SCS, which achieved a response rate of 41%. Participants from the first phase took part in the second.

Although different research instruments were used in the study the research design was such that comparisons could be made between cases in phase one as well as between phases. Both methods were broadly based on the EBS SPPM as well as the extensive literature base pertaining to strategic planning process characteristics.

The exploratory nature of the study necessitated the rigorous pilot testing of both methods, the detail of which is contained in the following chapter.

7. PILOT STUDY

7.1 Introduction

The preceding chapter outlined the chosen methodology for undertaking the research. To test the appropriateness of the methodology a pilot study was undertaken prior to the main study. The importance of pilot testing cannot be overstated. Experience from undertaking a pilot "*helps to throw up some of the inevitable problems of converting your design into reality*." (Robson, 2002; pp383)

This chapter provides the results of pilot testing of both phases of research. Feedback and amendments as a result of pilot testing are highlighted and discussed as well as mechanisms by which the issues were addressed for the main study.

7.2 Pilot Testing Phase 1: Semi-Structured Interviews

The literature drew attention to the paucity of research into the strategic planning processes of Irish QS practices. The pilot testing of phase one was therefore crucial in shaping the design of the main study into somewhat unknown territory.

The objectives of undertaking the phase one pilot study include:

- 1. to gain insight into the strategic management of QS practices in Ireland.
- to ascertain the extent of strategic planning undertaken in Irish quantity surveying practices
- 3. to identify critical factors affecting QS practice strategy planning processes
- 4. to determine the extent of application of strategic management theory to practice
- 5. to secure participation for future stages of the research.

Over a three day period five semi structured interviews were undertaken with senior management in Dublin based QS practices. The practices were of varying sizes (in terms of numbers employed) and some had affiliations with larger, global quantity surveying practices. The practices were chosen based on known information as regards size and ownership structure such that a representative range of companies could be included in the study.

Practice	Number of Employees	Ownership Structure	
А	11	Independently owned	
В	40	Independently owned	
С	100	Subsidiary of UK company	
D	25	Subsidiary of UK company	
Е	11	Independently owned	

Table 7.1Pilot study participant practice size and ownership structure

Initial contact was made by telephone outlining the purpose and scope of the research followed by a request to participate. Dates and times were agreed to suit respondents and took place in the respondent's office.

In all but one case interviews were tape recorded with the consent of respondents. Transcriptions were made of each recording to ensure reliability and validity of analysis. As recommended by Miles and Huberman (1994) contact summary sheets were also drawn up following each meeting with key aspects and points of note made, again, to facilitate analysis. One respondent appeared reluctant to participate and in this case it was decided not to tape the interview. The respondent ultimately provided a thorough insight into the strategic planning practice within the company which may be as a consequence of not being recorded (the respondent participated in the main study, which was tape recorded, and was noticeably more guarded in his responses).

Interviews were observed by a research fellow from Edinburgh Business School. The role of the observer was to provide guidance regarding interview technique as well as to gain an insight into the challenges in undertaking the research. The observer did not participate in the interview process, however, brief notes were taken to provide feedback to the researcher.

Each interview commenced with an outline of the study, scope of the research as well as an indication as to how the respondent's participation would contribute to the research. Permission was sought once again for the use of a Dictaphone, and the role of the observer was clarified and confidentiality was assured.

The pre-prepared interview questions contained five sections broadly using EBS SPPM as a guide. The general areas addressed included:

- 1. General Company Information Company demographics
- 2. Organisational Objectives and Strategist
- 3. Environmental Analysis
- 4. Strategic Choice
- 5. Implementing Strategy
- 6. Strategic Planning Process

Participants were not provided with questions either in advance or during the course of the interview. The interviews were semi-structured in orientation to facilitate and encourage discussion and to enable respondents to provide as much detail as they wished. The semi-structured interview ensured each issue was addressed, however, there was considerable flexibility in the sequencing of questions and explanations were provided when necessary (Robson, 2002). Probes were used to gain deeper insight into issues the respondent was willing to discuss pertaining to the strategic management and planning practices of the firm. Interviews were on average ninety minutes in duration.

The EBS SPPM diagram was presented to the first two respondents, however, it served only to confuse the respondents and it was not revealed in subsequent interviews.

Preliminary questions sought information regarding company demographics: company size, sectors serviced and services provided, number of years in business and ownership structure in an attempt to relax the respondent and slowly build up a rapport, as suggested by Robson (2002).

Section 2 sought to discover the extent of strategic planning and thinking within the company. Issues surrounding company objectives, how they are developed and by

whom, methods by which objectives are achieved and difficulties associated with achieving them formed the basis of this section. Characteristics of the strategies and the approach to strategy were discussed as well as the existence of strategic plans and their formality. Where a written strategic plan was in place, further detail regarding the process was sought.

The environment (competitive, macroeconomic, internal and industry) was the focus of the next number of questions. Respondents were asked to describe the company's experience over the last number of years to determine the critical environmental factors influencing the practice. The process for monitoring changes in the environment, as well as the extent of environmental analysis undertaken, was determined following which mechanisms for monitoring and communicating such information were discussed. Crucially the mechanisms for coping with environmental demands were addressed by discussing the strategic choices made by respondents.

Matters pertaining to the current organisational corporate and business strategy were discussed as well as the method(s) by which they were being sought. Through discussion of these fundamental issues difficulties in the pursuit of organisational goals were identified as well as the existence of contingencies.

The final section related to issues of implementation of strategy such as monitoring and reporting structures, information technology as well as human resources issues.

Respondents were asked for advice as regards other matters that could usefully be examined in the main study, however, no suggestions were made. The researcher also enquired as to what respondents envisaged as being the key challenges which may be encountered at future stages of the research. The overriding challenge identified was perceived to be in securing commitment and full engagement of quantity surveying practices given the severity of market uncertainty currently facing the industry (and consequently focus being elsewhere). Notwithstanding the challenges, each respondent noted the importance and usefulness of the research and all expressed a willingness to participate in the main study. As noted in the objectives above, the pilot study further sought to secure participation in future stages of the research. In qualitative research it is not uncommon for pilot study informants to be used in the main study. As noted by Robson (2002):

"Most flexible designs can incorporate piloting within the study itself. The effort needed in gaining access and building up acceptance and trust is often such that one would be reluctant to regard a case study or ethnographic study simply as a pilot". (pp. 383)

All pilot study participants agreed to be involved in the main study.

7.2.1 Findings from Phase 1 Pilot Study

The pilot study provided invaluable insight into the strategic planning practices of QS firms in Ireland. There is clear evidence to suggest respondents were thinking and acting strategically in line with existing models of strategic planning. Whether this was undertaken on a formal or informal basis or indeed whether the output was formally recorded or not differs between respondents. Often planning formed part of ISO certification and was thus seen as a quality assurance exercise rather than a strategic planning function.

A notable difference in approach to strategic planning centring on ownership structure and size of the organisation was also apparent. Strategic planning was a formalised, comprehensive process with widespread participation in practices that are wholly owned subsidiaries. This group tend to be larger firms offering a wider range of services on a domestic and international basis. These characteristics were in contrast to smaller practices which tended to have an informal process driven by the principal which was often not formally recorded or communicated.

In general the use of strategic planning tools, and indeed strategic planning terminology was limited in practice. This proved an important contribution of the pilot study and consequently strategic planning terminology found in academia was necessarily adjusted to better reflect practice for undertaking the main study (both qualitative and quantitative phases).

Firms tended to develop strategy alone the lines of existing models contained in strategic planning literature whether they knew it or not. One notable area of divergence from the literature lay in competitor analysis. In no instance was competitor analysis undertaken and information pertaining to competitors relied on the industry "grapevine". This is a crucial aspect given that it is widely accepted the competition is becoming more intense in the ever shrinking market, and perhaps the lines separating the segments may become increasingly blurred in the future.

7.2.2 Phase 1 Pilot Study Conclusions and Adjustments for Main Study

The pilot testing of in-depth interviews proved invaluable for the main study. A number of conclusions were drawn:

- QS Practices engage in the strategic planning process to varying extents.
- Ownership structure is a key influence on strategic planning formality and comprehensiveness
- Principals think and act strategically, whether they realise it or not
- Limited competitor analysis conducted regardless of practice size.
- Reputation of the firm is crucial for the repeat business on which many firms rely
- Limited use of strategic planning tools, regardless of strategic planning formality
- Use of management/strategy "buzz-words" need to be kept to a minimum in main study
- Use of EBS SPPM diagram confusing to industry practitioners, not to be shown in main study
- Firms recognise the importance of the study and expressed willingness to participate in future stages of the research.

Further lessons were learned in terms of interview technique. The pilot study revealed that keenness to ensure each area was specifically addressed at times resulted in the researcher moving quickly to the next item on the interview prompt sheet. As Robson (2002) noted, detailed information will be obtained by letting respondents talk, even if it

is away from the topic. One should listen more than you speak and this should encourage more depth in discussion. This important lesson was beneficial for the main study.

A limited number of amendments were made to the interview prompt sheet following the pilot test, the most notable of which was the inclusion of a question on how practices set about reducing cost and yet maintaining the quality of services provided. Any other changes to the "Interview Prompt Sheet" revolved around the sequencing (rather than wording) of questions and as stated previously, given that questions were not necessarily answered in sequence the impact of this on the main study was minimal.

Finally, analysis of pilot test results was undertaken manually by the researcher. Given the breadth of information gained and the larger number of respondents involved in the main study, it was determined that an electronic qualitative analysis software package be used for the main study.

7.3 Pilot Testing Phase 2: Online Survey

A questionnaire was developed following extensive research and analysis of existing literature. Previous empirical studies on the type, extent, scope and characteristics of strategic planning provided a useful framework within which to formulate questions. Furthermore core matters arising from phase one were incorporated into the questionnaire. For example, phase one had proved that respondents are not necessarily familiar with strategic planning terminology, thus care was taken to ensure questions were phrased with this in mind.

The breadth of the data required necessitated the inclusion of a large number of questions. In order to maximise the response rate questions were formulated to be as concise as possible with a minimum number of open ended questions. This is consistent with Dillman (2000) who noted that:

"Questionnaires that appear shorter and easy to fill out lessen the perceived costs of responding. Such appearances can be reinforced by indicating in the cover letter that responding should only take a few minutes." (Dillman, 2000, pp. 18)

Dillman (2000) also notes the importance of designing a user-friendly questionnaire:

"...some research has shown that making a questionnaire respondentfriendly is most likely to improve response among people who are least likely to respond to surveys, and thus help reduce non-response error." (Dillman, 2000, pp.81)

A variety of professionals were involved in the pilot testing of the questionnaire including 4 practicing QS's and three research professionals. One QS professional was also involved in phase 1 of the research so clearly understood the purpose of the research at hand. The contribution of this respondent was thus particularly valuable.

A single page questionnaire was administered with the pilot study to gain feedback from respondents (see Appendix G).

Based on feedback received, changes were made to the original questionnaire primarily in sections two and three on the basis that:

- 1. question length was excessive in some cases
- 2. strategic management/planning terminology was not clearly understood
- 3. the requirement to provide a response to every question was off-putting, and may result in a lower response rate.

The final point above relates to a function on the survey software that requires respondents to answer a question(s) prior to being permitted to proceed to the next. The pilot study survey contained a number of such restrictions; however, respondents expressed frustration that it was not possible to skip a question if they did not wish to answer it. The restriction was thus removed for the main study to maximise the response rate. The result of which is that the response rate may not be uniform across all questions. Where a divergence occurs it is noted during analysis (chapter 9).

Examples of changes made to the pilot study are outlined (but are not limited to) the following:

Section 1

• Question 7: wording changed from:

"The following is a list of services provided by Quantity Surveying Practices. Please indicate the key services currently provided by our organisation" to:

"Please indicate the key services CURRENTLY provided by your organisation"

• Question 8: restructured from:

"If the range of services provided by your organisation has changed within the last three years, please indicate the range of services provided by your organisation during the construction boom (c.2007). If the range of services has remained unchanged, please select the first option (i.e. no change in services provided)". (This was a multiple choice listing various QS services) to:

"Has the range of services provided by your organisation changed in the last 5 years?" The question was a Y/N multiple choice, with if Yes, please specify

• Question 9: Question shortened from:

"The following is a list of key sectors within the construction industry. Please rank the sectors in accordance with the proportion of work your organisation undertakes within each. Where 1=greatest proportion of work and 4=smallest proportion (Note: only one response per column is permitted)." to:

"Please identify the proportion of your overall workload CURRENTLY being undertaken within each of the sectors listed below."

Section 2

- Description of section contents removed as it served only to provide more reading for respondents
- Questions 12, 13, 15 and 16 on pilot study reworded and shortened.
- Questions 14 and 17 in pilot study removed (thus numerical order changed in main study).
- Questions 18-21 in pilot study reworded but options in multiple choices did not change. Amendments made to the question are, for example, from:

"Generic corporate strategy relates to the scale and scope of a company's operations Please select the strategy that best describes your CURRENT corporate strategy."

To:

"Corporate strategy. Please select the strategy that best describes your CURRENT corporate strategy/business objective."

Section 3

- Question sequence was changed to improve the flow
- Question 25 on pilot study (23 in main study). The question is largely based on the work of Papke-Shields et al. (2006), however, many options provided in the original study were omitted and remaining options shortened.
- Question 27 and 28 (benefits and disadvantages of strategic planning) were omitted in main study. The questions, although interesting, were not directly related to the research aims and objectives and only served to lengthen the survey.

Validation was achieved by returning the amended questionnaire to three pilot study informants for review and comment. All informants confirmed that the questionnaire was user-friendly, clearer to understand and took less time to complete. The questionnaire was thus finalised (see appendix H).

7.4 Conclusion

The purpose of a pilot study is to "test drive" the chosen methodology such that changes and improvements may be made prior to administering the main study. Mixed methods were used therefore both qualitative and quantitative methods were pilot tested. The experience gained through pilot testing served to improve both research methods.

For the qualitative phase, the pilot study was instrumental in fine tuning not only the questions asked but also interview technique. Importantly the use of strategic planning terminology was reduced and the EBS SPPM was not shown to participants. Furthermore the pilot study proved the necessity to utilise Computer Assisted Qualitative Data Analysis Software (CAQDAS) for the main study.

The quantitative pilot test served to fine tune the questionnaire to ensure clarity of questions, optimum sequencing of questions and thus increase the response rate.

Following the pilot testing of both research instruments, the main study was administered. The qualitative phase took place in June/July 2009 and the quantitative phase in June/July 2010. The following chapter provides a detailed account of the data collection and analysis techniques used for the main study.

8. DATA COLLECTION AND ANALYSIS

8.1 Data Collection

Data was collecting in two phases reflecting the chosen mixed methods approach. Ethical considerations were borne in mind in the handling of data at each stage and every possible effort was made to ensure the integrity of the data collection and analysis processes.

8.1.1 Phase 1 Data Collection: Semi-Structured Interview

Qualitative approaches require a rigorous data collection and analysis procedure to compensate for the apparent reliability and validity difficulties associated with a flexible design (Robson, 2002). Furthermore such rigour can ensure a transparency to prevent against research bias and facilitate the replication of the study in other industry or professional service contexts. Data was collected for this phase of research by means of semi-structured interview whereby questions were prepared in advance of the interviews and followed comprehensive pilot testing (see chapter 7). Participants were not provided with questions either in advance of or during the interview.

The process of data collection commenced with the researcher contacting participants by telephone to outline the scope and purpose of the study, as well as what was required of them should they agree to participate. Interview dates and times were scheduled to take place at a date and time most suited to the respondent and in all cases took place in their own office. Permission was sought at the outset to record the interview for transcription purposes, and in every case it was granted. Confirmation of interview date and time was forwarded in writing to each participant.

Interviews were on average 90 minutes in duration and tape recordings were transcribed by a professional transcription company with experience in transcription for doctoral research. A confidentiality agreement was signed in advance of forwarding the recordings for transcription. For validity purposes, transcripts were returned to

108

informants to confirm the accounts as accurate (Miles and Huberman, 1994) prior to analysis being undertaken. No significant amendments were required.

8.1.2 Phase 2 Data Collection – Online Survey

The second phase of research involved a widespread survey of QS practices registered with the QS Division of the SCS. The questionnaire was administered with use of an online survey facility called "survey monkey" (www.surveymonkey.com). Key informants had been identified and where possible personal e-mail addresses were sought in place of "<u>info@xyz.ie</u>" addresses, to increase the response rate (Dillman, 2000).

A three wave approach to data collection was undertaken. Initially an e-mail was forwarded to each practice which outlined the purpose of the study and assured participants of the confidentiality of responses (see appendix I). The covering e-mail contained a direct link to the survey and also provided the opportunity for respondents to click on another link if they did not wish to participate. This ensured that no further correspondence or follow-up e-mails would be sent to those that did not wish to receive them. E-mails were personalised where possible and informed respondents of the relationship between the research and the SCS. Reference to the SCS is in keeping with recommendations made by Dillman (2000) who contends that the response rate may be improved if the request comes from "*an authoritative source*". Covering e-mails differed between those whom had participated in phase one and those who were participating for the first time (see appendix I). On completion of the survey a "thank you" page was displayed providing contact details of the researcher should respondents have further queries. This final page reiterated the confidential nature of the information provided.

The first prompt resulted in 37 responses. Two weeks after the initial contact, a second e-mail was forwarded only to those that had not responded to earlier correspondence. A covering e-mail encouraging participation accompanied the link to the survey, and once again confidentiality was assured (see Appendix I). In this wave of correspondence a closing timeframe was identified and a further 19 responses were received. The third

and final wave of correspondence was forwarded two weeks after the second which prompted 7 additional responses.

The table below provides an overview including the final response rate:

Contact	Number of Responses
Administer questionnaire.	37
Second wave - additional responses	19
Third wave – additional responses	7
Total number of responses	62
Sample size	150
Overall Response Rate	41%

Table 8.1Response rate to questionnaire

The response rate is good and despite it being lower than some studies (Dansoh, 2005; Robinson and Pearce, 1983) it is considerably higher than many often cited works (Pierce, Robbins and Robinson, 1987; Covin and Slevin, 1989; DeSarbo et al., 2005; Phua, 2006).

As with many surveys, a number of people did not respond (Bryman and Bell, 2007). A difficulty arose in determining the potential for non-response bias, as the SCS do not maintain a company profile of its members. It is difficult to make a definitive judgement regarding non-response bias, however, it is reasonable to assume that the majority of non-responses were from small QS practices. Some practices of a similar size, on initial contact chose not to participate in the study as they believed they did not engage in strategic planning. The researcher contacted a random sample of 10 non-respondents and determined that in every instance the firms had no more than eight employees. Approximately 50% of those that did respond had fewer than ten employees hence the contribution from this group is considerable and is clearly represented in the findings. In light of this contribution, it is concluded that non-response bias is negligible.

Respondents were not obliged to answer every question on the survey (as a result of pilot study feedback). Eight respondents failed to answer the final section of the survey. Given the exploratory nature of the study it was deemed pertinent to follow up with these people to ascertain possible causes of excluding the final section.

The researcher telephoned each respondent in this regard. During the conversation respondents were thanked for participating in the research and were reminded of the important contribution they had already made. Respondents were informed that follow up was being undertaken as the questionnaire was not completed in its entirety and feedback was requested for the purposes of improving future studies. Specifically, the following questions were asked:

- 1. Were you aware that you did not complete section 3?
- 2. If yes, was there a reason why you complete sections 1 and 2 and not section 3?
- 3. Would you be willing to complete section 3 now over the telephone?

In those cases where contact was made, two respondents were not aware there was a third section and both completed it over the phone. In two cases there had been an issue with IT systems. The survey link was e-mailed once again to these respondents however, they failed to complete the final section of the questionnaire.

Given the contribution to knowledge of the first two sections alone, the eight responses were included in the analysis of those sections. The marginally smaller sample (55 responses) was included for the final section, rather than disregarding incomplete responses as "unusable". The 55 responses represent a response rate of 37%.

As previously noted, those that were involved in phase one were included in phase two. It was therefore possible to track the answers provided by this group which served to triangulate results in a number of instances. Where answers varied between phases, respondents were contacted again to confirm and discuss the developments. The relevant changes are discussed in section 9.4.2.

8.2 Data Analysis

8.2.1 Phase 1: Qualitative Data Analysis

The transcripts of interviews, once confirmed by respondents to be accurate, formed the basis of analysis (see Appendix J for excerpt of transcript of exemplar case, Company A). Field notes were taken during each interview, and as recommended by Miles and Huberman (1994) contact summary sheets and summary sheets were compiled for each case based on the field notes taken. Pertinent points and observations of the researcher were also documented and considered during analysis (see Appendix K for sample from exemplar case, Company A). A summary of findings from phase one was forwarded to participants to validate results.

A considerable amount of data was accumulated which favoured the use of qualitative data analysis software. Nvivo 8 was the software chosen as a licensing agreement was already in place for its use through the researcher's workplace. Training was required and provided by a specialist training company. A confidentiality agreement was signed in advance of training taking place.

The use of computer assisted qualitative data analysis software (CAQDAS) is becoming increasingly prevalent in qualitative research reducing many time consuming activities like multiple copying of transcripts or cutting out chunks of text for sorting (Bryman and Bell, 2007). The code-and-retrieve approach is common amongst many CAQDAS and provides many advantages. The researcher must still examine the data in great detail in order to code it, however, the retrieval process is considerably more efficient and one may argue that the whole process of analysis is more transparent with the use of such electronic tools (Bryman and Bell, 2007).

Limitations in the use of CAQDAS are outlined by Silverman (2010) when it is noted that word processors could provide similar tools for retrieval of coded data, too narrow an approach to analysis may be adopted and also for small data extracts it may be more time efficient to process manually. The research at hand produced several hundred pages of interview transcripts therefore the manual processing of data would prove too cumbersome, particularly in light of the numerous facets of strategic planning being considered. The potential limitations of CAQDAS were thus not applicable to the study.

A project in Nvivo 8 was established for the research at hand, wherein data files were uploaded including interview transcripts, interview audio-files, pertinent points from interviews, contact summary sheets, company documents (when provided) as well as key literature pertaining to the study. Interview transcripts were once again tested for accuracy by the researcher against audio files uploaded into the software, prior to codification.

A "Case file" was established for each respondent and within the software "Attributes" assigned to each case (see appendix M for exemplar in this regard). Attributes for the project included strategic planning characteristics, generic strategies (current and past), sectors serviced and services provided as well as preliminary company demographics. Once attributes were assigned it was possible to view individual case attributes or alternatively to run queries between cases on a particular attribute.

As noted previously, Nvivo 8 software is a code-and-retrieve system, consequently considerable care must be taken in the design of the coding framework. The main headings (*"tree nodes"*) under which interviews were coded hence analysed were derived from components of the EBS SPPM as well as existing strategic planning process literature. Sub-headings (*"child nodes"*) were derived in a similar fashion and also incorporated elements which arose during the course of the interviews. The coding strategy for the research is detailed in Appendix L. The attention given in designing the coding framework served not only as a basis for the analysis of individual cases, but also for cross case comparison and ultimately the categorisation of QS practices into groups (refer to section 9.2.8 and 9.4.3).

Following the coding of interview transcripts within the software, a random check of coded data was carried out by a third party to verify the appropriateness of same. In no instance were amendments suggested or made.

The coding strategy developed for phase one was mirrored in the design of the questionnaire for phase two. The consistency in approach in this regard is a significant

strength of the research design ultimately resulting in the triangulation of research findings.

8.2.2 Phase 2: Quantitative Data Analysis

Quantitative data was analysed using the online survey tool for the questionnaire (<u>www.surveymonkey.com</u>). Responses to the questionnaire are stored online and basic analysis and display tools were available directly from the survey website.

Statistical information such as the number of respondents to a particular answer choice and response percent was available online. Individual responses to the questionnaire as well as a summary of all responses were also accessible online. For more detailed analysis filtering and cross-tabbing of responses was possible following which displays could be generated. All results were exported to MS Excel for further analysis and comparison.

8.3 Validity and Reliability

Accuracy in the collection and analysis of research data is essential in a research project. The mixed approach employed for the research permits triangulation thus increasing the reliability of findings. The pilot testing of both phases further contributes to the reliability of the chosen methods.

A number of provisions were made to increase the validity and possibility of replication of the qualitative phase of research, including:

- interview prompt sheets were developed prior to undertaking interview
- interviews were taped and transcribed
- transcription accuracy was confirmed by respondent
- a summary of findings was forwarded to respondents
- a third party randomly checked coded data

The second phase of the research was developed in such a manner that satisfies many of Denscombe's (2010) criteria for research reliability in that it is possible that the questionnaire administered may be used:

- in different settings, or
- by different researchers, or
- with the same people at different times, or
- with separate groups of similar people at the same time

(adapted from Denscombe, 2010, pp, 144)

The questionnaire was informed by existing research in the area thus satisfying criterion validity (Denscombe, 2010).

8.4 Research Ethics

Ethical considerations are an important aspect of undertaking research. Ethics are concerned with what is good or bad and influence the:

"...norms and standards of behaviour that guide moral choices about our behaviour and our relationships with others." (Cooper and Schindler, 1998, pp. 120)

The potentially sensitive nature of the data being collected for the study necessitated considerable care to be taken not only at data collection but also analysis stages of the research. At the commencement of each phase the purpose of the research was clearly outlined to respondents and anonymity of respondents was repeatedly assured.

Prior to commencing the first phase of research, permission was sought in advance of to tape record each meeting. Interviews were transcribed using a professional transcription company and as previously noted a confidentiality agreement was signed prior to forwarding the recordings. Company and director names were replaced with a letter in each of the transcripts and information which may identify respondents was not specifically referred to during the course of analysis. In one instance a respondent

specifically asked for a piece of information (which could identify the practice) to be excluded, and his request has been honoured. It should be noted that it did not in any way detract from the quality of data nor scope of discussion. A hard copy of the interview transcript was mailed to each informant, as suggested by Miles and Huberman (1994) to secure confirmation of transcript accuracy. A feedback form was enclosed for completion to confirm the precision and/or to identify any amendment(s) to be made. A deadline date was provided for the completion of the form and respondents were informed that if a feedback form was not returned it was assumed no amendment was to be made.

Nvivo 8 was used in the analysis of the qualitative data for which assistance was required. Once again, a confidentiality agreement was signed with the training company prior to commencement of the training.

Respondents for the second phase of the study were drawn from the SCS QS Division database. Anonymity was assured to all participants and neither individual nor company names were identified either in the analysis of results nor to the SCS. The researcher had sole access to the online survey database used.

8.5 Limitations of Methodology

Strengths and limitations of various research tools have been identified in many previous studies (Miles and Huberman, 1994; Robson, 2002). In order to address the limitations, a mixed method design was employed for the purposes of this investigation. This is important with an exploratory study whereby the researcher endeavours to investigate a phenomenon in previously unexplored territory. Consistency in facets of the strategic planning process being considered between phases of research as well as methodological triangulation ensured that many of the limitations in conducting business research have been overcome.

However, the research was conducted on a single profession within a single industry therefore an inherent limitation as regards generalisability may exist. The paucity of research undertaken in the sector at the time of writing necessitated a thorough and detailed investigation of the profession and legitimises the narrow sample frame.

One could argue that a constraint exists with the sample selection. The phase two sample was chosen from QS practices who are members of the QS Division of the SCS. There are several QS practices that do not hold chartered status, are not members of the SCS, and are therefore excluded from the research. Future studies could usefully draw comparison between the strategic management of each classification of QS practice; however, the difficulty in obtaining data pertaining to practices without chartered status presents a significant obstacle.

8.6 Conclusion

This chapter provided a detailed overview of the chosen methodology for the study. Mixed methods were employed to gain an understanding of the exploratory investigation, thus elements of the phenomenological as well as positivist paradigms are evident. A rigorous research design enabling triangulation as well as a comprehensive data collection process ensured the integrity of the data, thus facilitating the replication of the study by other researchers.

The use of IT in the collection and analysis of data not only increased the efficiency in the process but further ensured a transparency and potential for replication. In order to test the chosen research methodology prior to its administration for the main study a pilot test of both components was carried out. Fine tuning of both methods prior to the main study improved the potential to replicate the chosen methodology.

9. RESEARCH FINDINGS AND ANALYSIS

The previous chapters outlined the methods by which field work was undertaken in order to ascertain the type, scope, extent and process of strategic planning within Irish QS practices. Mixed methods were employed to address the research question. This chapter presents the findings from the two phases of fieldwork.

The first section presents the findings from semi structured interviews, which were conducted with eleven senior managers in QS practices. The second section presents findings from phase two of the research, namely, a widespread survey of QS practices in Ireland. Participants from phase one were included in the survey sample for triangulation and generalisability purposes. The final section of the chapter provides a summary of the key findings of the combined approaches to the research.

9.1 Phase 1 Findings – Semi-structured Interviews

The EBS SPPM provided a framework for undertaking and analysing semi-structured interviews, thus the following sections are presented broadly in line with components of the model.

9.1.1 General Company Background

Every practice in phase one was established at least two decades prior to the interview taking place, thus each had considerable experience within the industry. The practices had survived a previous recession and recognised that as a cyclical industry, the current recession would eventually give way to growth, and as such, practices needed to "weather the storm". Despite the pragmatic approach displayed by respondents, the severity of the current recession had a notable effect on the number of people employed. On average, staff numbers had reduced by 40% from the peak of construction employment (2007) to the time of interview (July 2009).

Company	Number of	Number of	Percentage
	Employees	Employees	Reduction
	(peak 2007)	(current July 2009)	
1	14	10	29%
2	17	8	53%
3	35	17	51%
4	85	45	47%
5	9	6	33%
6	27	17	37%
7	15	9	40%
8	60	30	50%
9	300	200	33%
10	140	75	46%
11	6	4 1/2	25%
Average			40%

Table 9.1Number of people employed in Phase 1 QS practices

[Note: Company names have been replaced with numbers solely for the table above and do not necessarily correlate with subsequent alphabetical names given. This is to protect data which may potentially identify respondents in later analysis.]

The table above demonstrates the extent of the reduction in staff numbers amongst all participant companies. There is no distinct correlation between practice size and percentage reduction in numbers employed.

Measuring changes in employment levels within the profession provides important information regarding the structure of the profession, which prior to the research being undertaken, was unknown. Further detail regarding the change in practice size is demonstrated in the following figures.



Figure 9.1 Number of employees at peak (2007)





The increase in number of small and medium size enterprises (SME's) in the profession is visible, with the majority of practices involved in phase one having fewer than fifty employees. A critical aspect affecting company size lies in the number of practices that made a strategic decision (prior to the recession), to limit the size of the company. In most cases this was driven by the desire to remain a "practicing" quantity surveyor. It was believed that increasing the size of the practice would negatively impact on the quality of service provided: "We made a conscious decision, even at the height of the thing (*construction boom*) that 16 people was our upper limit. Actually, this office can accommodate 20 people, but we never envisaged going near that because...You start to delegate out more and more, you lose quality control - we're control freaks, we're not great delegators!" (Company A)

In some instances lifestyle choices influenced the decision to restrict practice size:

"The driving force behind the decisions we have made and our decisions not to grow... we are only here once and I have no intention of going and dying of a stress related heart attack because we decide to grow the organisation." (Company K)

A direct correlation can be made regarding the strategic decision to restrict the size of the practice to risk attitude, with three quarters of those restricting the size of the practice categorised as being risk averse.

Considerable change is evident within the industry. With the tide turning in the private sector, a change in emphasis towards the public sector has occurred; not as a consequence of increased capital expenditure by the government, but rather as a result of the severity of the reduction in private investment. The figures below reflect the extent of this change.









The range of services offered varies between practices, and no unambiguous link exists between the size of the organisation and the range of services provided. Larger practices tend to offer a greater range of services, however, many smaller practices are beginning to diversify the range of services beyond "core quantity surveying". Intense competition, demanding clients, as well an emphasis on sustainable construction are key aspects in the emergence of new service offerings.

Each of the five larger practices has its head office in Dublin with regional offices in other large cities around the country, and the remaining practices are based solely in Dublin.

An important differentiating factor amongst respondents lies in the ownership structure. Three practices have formal links with a larger UK construction cost consultancy, two of which are subsidiaries of a UK organisation (the remaining one is part of a legally formed strategic alliance). In the case of the former, the practices worked together on previous projects and subsequently formalised the arrangement. In the case of the strategic alliance, although no equity has been exchanged between the partners, the alliance is advantageous for information sharing as well as (from the Irish firms' perspective) providing the capability of undertaking large scale projects in the domestic market.

A link is evident between those practices with international associations and the strategic planning process, which ultimately results in the existence of groups within the profession (see section 8.2.9). This is consistent with the findings of Dansoh (2005) within the construction industry in Ghana. The significance of the relationship is outlined by Porter (1980) who identified the importance of the inclusion of a firm's relationship with its parent in the definition of strategic group dimensions, noting that:

"Often relationships to the parent also translate into differences in the other dimensions of strategy...because the relationship has a lot to do with the resources and other strengths available to the firm and the philosophy with which it is operated." pp. 130

9.1.2 Strategic Planning in Irish QS Practices

The following section provides an analysis of strategic planning processes within the QS practices in phase one. The analysis is broadly structured around components of the EBS SPPM.

Strategist

Regardless of the size of the organisation, senior management ultimately drives the strategic decision making process. In some instances employee feedback is collected; however, despite claims of flat organisational structures and open communication, the lines of demarcation are apparent regardless of the size of the practice.

"They actually all like to have an involvement in management but that's clearly just not practical in any organisation I don't think." (Company J)

Strategist characteristics vary broadly based on the size of the company with the strategists of small/medium sized practices displaying primarily "defender" or "reactor" characteristics and the larger practices being categorised in the main as having "prospector" characteristics (Miles and Snow, 2003). A summary of types as they relate to participants in phase one is represented in the figure below.

Figure 9.5 Strategic type



Risk

The risk attitude varied amongst respondents. In the case of one of the smaller practices the attitude was described as being "conservative":

"We're an old established company, we were conservative during the good times, we've continued to offer the service that we always did during the bad times...We were always being highly conservative so our cost base never expanded hugely and our revenues just marched on." (Company A)

In one of the larger practices, the risk attitude may be described as risk neutral in that significant data gathering was conducted prior to making strategic decisions, but that did not mean that the risk would not be taken:

"...it's about information, it's about knowledge. So you can often discover it's better to keep your powder dry and a lot of people jump first and then wonder where they are. We tend to do it the other way, we debate something before we move and we tend to be cautious in that sense." (Company H)

Whereas one respondent rated the practice seven on a scale of one to ten (where ten was the most risk-taking) and another noted that:

"I wouldn't have viewed myself as being a major risk taker but I am here where I am and I didn't get here without taking risks ... they (the organisation) would take risks, yes; they do all the time. You know working here is a bit of a risk, you work in the private sector it is a bit of a risk." (Company C)

A small number of practices provide risk management advice to clients as part of the service. One company (company B) noted that "*We actually do a risk analysis of the costs of either settling them or going to arbitration and you can usually project it fairly well*." When probed about risk analysis at corporate level, the response was "*we don't structurally risk analyse*". In another instance (Company I) a computerised qualitative risk assessment package is used at project level, however when asked whether the

125

database was used at strategic level "We don't do a formal risk assessment in our internal strategic reviews; we would just run various scenarios when we are looking at options."

Operational risk management is addressed in partial fulfilment of the ISO certification requirements hence the ISO certified practices have undertaken a degree of risk analysis. Notwithstanding the ISO requirement, in no case was a whole-scale strategic risk assessment conducted for the practice as strategic risk management is tactical rather than strategic.

9.1.2.2 Macroeconomic, Competitive and Industry Analysis

The environment within which a firm operates has a significant impact on strategic decision making (Junnonen, 1998; Grant, 2003; Brews and Purohit, 2007). An unequal emphasis is placed on each of the four environmental components (macroeconomic, industry, internal, competitive), within the QS profession based on the chosen sample. Considerable analysis of the macroeconomic, industrial and internal environments are undertaken (not necessarily systematically), with considerably less focus being placed on the competitive environment.

Macroeconomy

The severity of the macroeconomic environment and the consequences for the industry are considerable. Changes within the macroeconomy are monitored by all respondents.

"The level of competition in the market is huge. The supply side is as competitive as it could be. The demand side is as low as it could be and all the factors that would usually be positives to demand, are going in the wrong direction, with the exception of interest rates, but there's a barrier to that, so you can't avail of that because there's no access to finance...uncertainty about the public purse, difficulties in both the private sector both in terms of commercial status, but also access to finance people aren't anticipating and planning for their projects next year." (Company A)
The general consensus amongst respondents is that signs of recovery are unlikely to be evident for at least 2-3 years. In the words of one respondent:

"At this moment in time our expectation is pretty poor. I think the market is going to be stagnant, stuck for the next 6 months at least until Christmas, until NAMA get themselves squared off. I like to think that once the bank situation is regularised and is back in train, that the market might start to move again but it will only be playing catch up to a certain level... it won't be anywhere like it was over the last 5-7 years." (Company F)

The significance of the role of government in the construction industry cannot be overstated. Respondents noted five important aspects of government intervention as having a significant impact on the construction industry and specifically on the QS profession namely:

- the effect of reduction in capital expenditure
- the new form of government contract (GCCC)
- the revised Conditions of Engagement for construction professionals
- the establishment of National Asset Management Agency (NAMA*)
- the blame for severity of impact of economic downturn

*NAMA was established in December 2009 to buy property-linked loans from five key banks in Ireland so that the loans would be removed from their balance sheets

A further change made by government is the implementation of a new form of government contract for construction projects.

"The new platform for contracts that we have for public sector are suggesting not necessarily lowest cost but a definite cost and the difference is that they want to know one figure and that the thing will come through at that one figure." (Company H)

There remains considerable scepticism regarding the contract which was "imposed" on the industry, unnecessarily, according to some.

Further changes have taken place regarding the procurement and conditions of engagement of consultants, which followed (but were not necessarily caused by) the new form of government contract. Each of these issues separately would have a significant affect on the industry, however, their concurrent implementation is posing a significant challenge for the profession. Interestingly the reaction to the reduction in fees has varied from general acceptance:

"You can negotiate your fees, but then the public bodies for instance, will impose an 8% cut in your fee, unilaterally. There's certain bodies resisting that, but again we're price takers not price makers. I do a lot of work for Dublin City Council ...it does not behave me to be giving them two fingers and saying well f*** you and your 8%. I would rather focus on having 92% and glad of it" (Company A)

to encouraging some practices away from public sector work:

"The government's new forms of contract, the government's new conditions of engagement, competitive fees...they are all adding weight to pointing us in other directions, or encouraging us to point in other directions to find new ways of developing our business, increasing our specialisation and hopefully surviving." (Company C)

However, many respondents were of the belief that these changes were wholly unnecessary and there is obvious concern as regards the effect on the quality of service provision, as noted by one respondent:

"I have to say that this relationship that I feel worked very well where you were judged on the service you provide not necessarily on the financial fee that you are going to offer. And that seems to be the road that we are going down now. The breakdown of this professional relationship with a client where you are professional, it's now down to a money value you win it on and God knows what kind of service you are going to get, what kind of relationship I don't know." (Company E)

At a macro economic level, the government is accused by one respondent of not keeping an eye on the "*big dials*", not having a coherent strategic plan for investment and of imposing unilateral changes on the industry. The lack of counter-cyclical investment in Ireland was cited by this respondent as being a critical factor:

"They (*the government*) threw an awful lot of money into the industry in the boom times and they are withdrawing it in the downturn. Government investment should be counter-cyclical and we have been saying this for 20-30 years that Government grants should be counter -cyclical. That they save the money in the boom times and they invest in the down times and that would retain the core skills." (Company K)

It is clear that respondents apportion a considerable amount of blame to the government for the severity of the crises within the industry. However, the government's continued inaction was also criticised:

"But in the meantime they could very easily have set up a group to deal with short term issues which could have got things going while these are being set up, there are a number of projects which are either self-financing or are viable and will give a return that are out there at the minute... And if there was a special group set up, a whole heap of projects could be started straight away while we are waiting for all these things to happen... I don't see why that type of group (similar to IFSC*) couldn't be put together to just identify projects, seek information from people who have it, I could give them 2 or 3 jobs, I'm sure other people know of other jobs that are out there that could get going, to give things a kick-start." (Company I)

*IFSC: International Financial Services Centre. Established by the Irish government in 1987 to boost activity and employment in the financial services sector.

An analysis of the macro economy is common amongst respondents. The principals monitor trends in the key macroeconomic indicators, however, there is little evidence of how trends are disseminated throughout the organisation. Seldom do procedures exist to communicate macroeconomic analysis amongst staff members, who are presumed to keep themselves informed on such matters.

Competitive Environment

The extent of competition within the industry is contributing to the "*firestorm*" facing the profession. The intensity of competition has resulted in the reduction in price levels and "*fees in the quantity surveying profession are changing dramatically and they are being driven downward on a very step curve*" (Company F). On the public sector side of the industry, an 8% reduction in fee levels as well as changes to the terms of engagement of construction consultant professionals are cited as being the key issues.

The reaction of firms to the above differs amongst QS practices. From competing for numerous projects in the hope of securing one:

"We did 130 submissions last year which I suppose we got back responses to 2 or 3 of them. You know we don't know whatever happened to them, but that was very much the spatter gun approach. We were well into 60 or 70 when the penny dropped and we said Jesus, we can't keep going like this, you know." (Company C)

to not competing at all for particular projects:

"If you go public; public sector work, the level of competition is such that you don't want the work at that price... it's almost like the perfect competition, except I'm 100% certain that guys are having to take work way below cost. They're buying turnover. Now, you cannot sustain that." (Company A)

Others believe that it is simply part of the changing nature of the industry, and given that many practices are price takers it is simply a fact of the market:

"That's the environment you are in now...you can be moan it, you can say "I don't know, I don't like doing this", but at the end of the day do you want the

work or do you not want the work? If you want the work you've got to compete for it, if you don't want it well then good luck to you." (Company B)

However, the change in fee structures provides a challenge for some practices that heretofore differentiated their service and charged a premium price for it:

"We would tend to be a company that would be one that provide additional services within a fee, rather than a stripped down fee - "if anybody wants one thing extra bill them for it". Traditionally we have not been a company that did that and that is a challenge for us to change, it requires a directional change." (Company H)

The implications of the reduction in fee levels are enormous and have been instrumental in two practices seeking work abroad, practices diversifying the range of services provided (see section 9.2.9 below) and all in having suffered financially at as a result of reduced prices within the profession. This becomes apparent when one considers the reduction of staff numbers and initiatives undertaken to create greater organisational efficiencies (see "Internal Organisation").

As previously outlined, the QS profession is characterised by a small number of large practices and a large number of small practices. Those within each group are very aware of the differences between the groups with references to the "*big practices in town*" made by smaller practices. Similarly larger practices do not compete for projects below a certain size yet have the capability of undertaking large scale projects for which smaller firms simply cannot compete. Many smaller practices have remained so following a strategic decision not to expand the organisation beyond a certain point (see section 9.1.1), which in some instances has been beneficial for survival through the downturn:

"We're a small to medium size enterprise. There's a niche for us there, we're comfortable in what we do, we're big enough to take on decent size projects and we're small enough to roll into a ball when we have to." (Company A) The size of the practice is also reflected in the range of services offered and generally the larger the practice, the greater the range of services being offered. The correlation between practice size and services offered is somewhat tenuous however, given the evolution of the profession as a whole into different areas such as sustainable development and legal aspects. In one instance the size of the practice has resulted in the use of strategic partners in order to compete on the range of services provided. The sporadic requirement for certain services means the QS practice does not develop expertise in all areas but rather uses strategic partnerships for the provision of specialist services as and when they are required.

Competition takes place both in the domestic market and abroad. Over the last several years many Irish QS practices have developed an international dimension as evident by having:

- international offices
- a wholly owned subsidiary of UK Construction Consultant
- strategic alliances with UK Property and Construction Consultant
- alliances specifically for entering a foreign market

Those practices with formal links to UK based global construction consultancies believe that the link strengthens the practice in terms of scope of service provision as well as gaining from experience in the UK market:

"We have seen great benefits in a lot of what's happening in England with PPP's* and PFI's**...and we just touch on their experience and knowledge for submissions, they have it straight off whereas we would be trying to find it." (Company D)

*PPP- Public Private Partnership **PFI – Private Finance Initiative

The UK company may also provide the knowledge and experience necessary for the practice to service Irish clients in international markets:

"Because we have clients coming in here, they want to look at building things in Qatar or somewhere and we have a local office so we know the local costs and we know local market conditions so we... you can do a lot of the traditional QS'ing stuff in terms of measuring and quantifying the building here, but you can get the cost information from our local people on the ground there. (Company F)

Furthermore, opportunities for staff development in foreign markets may also be utilised as a consequence of the link with the larger practice:

"So I think you should (if its effective –the international relationship) be able to get people to go to different locations for periods of time and then if you get them back, which we have done before in instances, they do definitely come back with a much more rounded experience, which I think is always a good thing". (Company J)

Those practices without links to an international partner and who have interests in international markets, believe that they are better placed to do so in the absence of this relationship (as they have flexibility to investigate wherever foreign market they chose without having to consider a parent company strategy).

"It's actually one of our key strengths as well, in that a lot of other firms in Ireland tied up with international firms over the last 10 or 15 years and while that gave them some advantage over their size in Ireland if you like, its disadvantage now is that they are locked into Ireland a little bit because of their associated firm...So we have no such restrictions, the world is our oyster and we intend to go out and pick as many pearls as we can!" (Company I)

Finally, one practice is using the downturn in the Irish construction industry as the driving force behind exploring international markets for the first time. Due to the size and lack of experience abroad it has chosen to form a partnership with another Irish QS practice to enter and compete in an international market.

Given the importance of the competitive environment in shaping strategy, it is somewhat surprising to note that the overwhelming majority of QS practices do not undertake any formal competitor analysis. A similar lack of competitor analysis was noted by Price et al. (2003) in the context of medium and large UK construction organisations. Respondents in the study put forward a number of explanations for the lack of such analysis, including one stating that:

"We wouldn't have a policy (*for analysing competition*). We kind of stick to our own devices really. No, we wouldn't have a strategy for understanding who our competition would be." (Company C)

"No we don't worry about our competition too much... Well we obviously check pricing levels but we get feedback now from government tenders so you know, there's not much research to be done on that. Other than that no we don't worry too much about the opposition. We try to let them worry about us more than we worry about them!" (Company I)

However, informal (cursory) competitor analysis tends to rely on the industry grapevine emanating from a number of informal sources, including staff members, SCS networking and the bank monitoring process.

Reasons behind the absence of competitor analysis included:

"No, too busy worrying about ourselves." (Company A)

"It's very difficult to find out about competitors. That's the issue there. We try to keep our ears to the ground as best we can, but getting factual information on our competitors' activity is always very difficult." (Company F)

One practice has undertaken formal and comprehensive competitor analysis, and uses an external marketing group to survey practices.

"We would have had surveys conducted about what was happening at our level in the business... We concentrate on what areas people are interested in as diversification and new development, we look at..., and what mix they have between private and public, and even within that what areas and locations, all that kind of thing, you know the geographical spread, and what perhaps international links they would have and why and where they see the industry heading and where they see the profession heading." (Company H)

Industry Analysis

In general, the extent of analysis of the construction industry in QS practices is more comprehensive than competitor analysis, however, not necessarily amongst all respondents. A comprehensive industry analysis is undertaken by two of the larger practices which have specialist teams in place to publish an annual review of the construction industry. These publications are referred to by the other practices and form the main source of industry-specific information alongside SCS publications and the annual industry review published on behalf of the government by DKM Economic Consultants.

Respondents were unanimous in their view that the industry is in the midst of the greatest crisis it has faced in recent history. There is a confluence of issues affecting the industry in terms of the level of competition and its affect on fees, lack of government investment, changes in procurement process, lack of availability of finance on the private side leading to projects being postponed, all of which contribute to the uncertainty within the industry.

On the public sector side of the industry, a number of changes have been implemented in the last two years, some of which were made without consultation, and which have resulted in resentment towards government in particular. Competitive tendering as well as the enforced reduction in construction profession fees in the public sector have had a snowball affect when considered in conjunction with the reduction in construction costs. Anecdotal evidence suggests that individuals are competing at below cost, which has obvious consequences for the profession as a whole in the long run: "There is a lot of pain in the industry now with companies that are in financial difficulties...the big worry for us all is that companies will be bidding for projects that would be below costs and it's going to lead to further problems down the road now with liquidations and uncompleted projects." (Company E)

Given the intensity of competition, serious concern was expressed regarding the level of service that can be provided at a greatly reduced fee (see "Internal Analysis"). The sharp deterioration of the industry as a whole coupled with the reduction in fees has raised concerns regarding the quality of the service that can be provided at the new level.

"If we want to compete with market levels we just can't provide the same quality documentation and that's the way it's going to go." (Company D)

"Well the one challenge for the profession is the competitive bidding is pushing fees below the service that should be given." (Company I)

"I think because of the levels of fees now that it's been dropped down to the quality of service is going to suffer big time." (Company J)

The quality of service is a key determinant in the demand for repeat business, particularly within the QS profession, hence the current trends and forecasts in this regard could prove devastating in the long run.

As is clear from the above, a number of challenges are facing the QS profession. However, a number of respondents noted positive trends in aspects of the industry such as tender price inflation, and they view them as potential opportunities for creating efficiencies:

"If you look at the levels of the industry, you'll see output has fallen back to 2001. 2001 wasn't a bad year. So, what you need to do is put your 2001 hat on and say, "how is it we did business then" and you know, retrace your steps and go back to first principles, and do what you were always doing... A lot of it will be good. It's going to be painful but there's a purge, there's a huge amount of deadwood and inefficiencies." (Company A)

and for improving the level of service to clients:

"We're getting to a level that we were at in 2002 which is probably the right level... I think it's fair to say that the service we provide is better now than it ever has been. It's more focused towards clients. Clients have become... well certainly clients have become more demanding and rightly so in what they want from us." (Company B)

The role of the QS continues to evolve. At the height of the boom QS's became increasingly involved in the residential sector, (particularly in mixed use developments) and many practices grew with the huge surge in demand for this sector. The range of services offered by many practices developed at the same time into project management, bank monitoring as well as services reflecting changes in construction techniques and embracing areas of sustainable development and M&E services.

The profession heretofore relied heavily on personal relationships between stakeholders to ensure repeat business (on which many practices at present rely), and it is generally argued that the new procurement methods will have a negative impact in this regard. It is widely anticipated that the QS will increasingly play a legal role as regards dispute resolution as well as the valuation and management of distressed assets and the training and education of QS's is believed to equip them with the skill necessary to undertake such roles.

Changes in the macroeconomic and industry environment will undoubtedly steer the QS in new directions as it has done in the past.

"We saw that the cake from which we traditionally fed shrink, and unless we made more bread some way or other it wasn't going to happen for us" (Company C) A structural change in how the QS performs his or her function as a consequence of industry forces was outlined by many respondents, none of whom were particularly enthusiastic about the prospect, as it signals a change in the way the QS has traditionally done business:

"And I suppose the American way of doing things, of having the number of hours allocated to a job, that's what we priced and that's what the fee is based on, and when it comes towards the end of a job the American client, sorry the provider of the service would say to you "now Mr Client or Mrs Client, you only have 10 hours of my services available left in this project. As to how you want me to spend that is up to you. It's not enough to give you what you are looking for; if you want me to do more I have to charge you more". (Company E)

Notwithstanding the above, there is an air of optimism regarding the future role of the QS. The issue is discussed by many respondents and analysed at length by one respondent, who identified many areas within which the QS is involved with the construction process. It is believed that the QS will continue to be retained by the client, the developer, the financer and in the words of one respondent:

"So in that way that's how I see the industry and our industry and our function in that industry changing...Now eventually we will be doing the same things as we are doing today but we will be doing them for different masters or different client bodies." (Company H)

Practicing QS's must be members of the Society of Chartered Surveyors and a practice will receive "Chartered' status once at least 50% of it's Directors are chartered. Only one practice considered within this research is without chartered status, and it is deemed by the respondent in that instance, to be a significant weakness of the organisation. This practice is the "deviant case'.

Active participation in the Society however varies in the extreme, and probes regarding the SCS uncovered very mixed views. The positive elements tend to indicate those practices which have an active involvement in the SCS, whilst negative comments tend to emanate from those that are less active in the Society.

The perceived weaknesses of the SCS were noted by a number of respondents, and tend to focus on the degree to which the Society will put "its head above the parapet" in support of members:

"It's lacking direction, its lacking focus, its lacking industry representation. They do nothing for us, it's a talking shop. A complete and utter waste of time." (Company A)

"I am sceptical about the SCS and what they deliver to their membership. I think that their back was broken several years ago by the government...it was just either: "put up and shut up'. So, really as a union, if you would like to call them, we are definitely very poor. We wouldn't even hold the coat of the medical consultants; those people know how to throw a punch, but we haven't been able to do that." (Company C)

While strengths tend to centre on information provision:

"I suppose benefits (*of membership of the SCS*), you're introduced to regular updates...They keep you well informed with e-mails and the website is very good now. They keep you informed just on surveying matters, the government reduction in fees, rules of measurement, any core issues they are very good on." (Company D)

and technical knowledge:

"... the completely indecipherable info. that comes out of Revenue for that sort of thing. It needs the interpretation of a group of people who sit down with Revenue, work it out for our particular part of building... And that sort of work has to be done and comes out and it is (promptly I hope) and to the benefit of the members. (Company K)

Interestingly the comments varied across practices of all sizes, and in the main were

negative. The significance of the finding lies in the necessity for QS practices to be members of the SCS, yet despite their membership many find little practical value in the professional body and are members of same purely because they cannot trade as Chartered QS practices without such membership.

9.1.2.3 Internal Analysis

Analysis of the internal organisation was comprehensive in most cases. As professional service practices, every participant was acutely aware of the importance of the resource base of the firm are particularly of human resources. The structure for undertaking internal analysis in many instances stemmed from either ISO requirements or head office requirements in the case of practices that were part of a larger construction consultancy organisation.

Human Resources

Analysis of human resources (HR) is an essential task of any organisation, and QS practices are no different. Arguably the strength of any professional practice lies with its people therefore a significant analysis of HR matters is undertaken at every level from recruitment, training and development, to retention and work practices.

The experience of QS practices as regards human resource management, has varied from having difficulty recruiting adequately qualified staff during the peak of the construction to retaining staff during it's downturn. The number of people currently employed has changed considerably in the recent past. Staff numbers have reduced in every participating practice primarily as a consequence of reduced workload, however, in a number of instances there has been a natural turnover of staff (people retiring or foreign workers returning home) who have not been replaced. The average reduction in staff numbers is presented in section 9.1.1.

Remaining staff in each practice have taken a pay reduction as practices seek to further reduce costs. In a number of cases changing work practices have been implemented, such as a shorter working week or job-sharing. Some respondents expressed concern at the level of job loss and the potential "brain drain' from Ireland and the implications for the industry when the economy recovers.

Staff training and development is ongoing and comprehensive in all practices. Training of staff is essential for chartered status in the SCS, and CPD is a crucial component of ISO certification. All respondents confirmed that staff members are financially supported, where possible, to undertake training programmes.

A small number of practices have a monthly in-house training programme, whereby staff members are asked to present a paper on a construction-related topic to peers, or external specialists are invited to speak about a particular issue. In-house training programmes are not limited (where they exist) to larger practices.

One practice has won the "50 Best Places to Work" award for two consecutive years. This award is centred on the staff and their response to a detailed survey over a number of months.

In practices with links to a UK company, the relationship is utilised to train and develop staff in international markets:

"We have relocated some staff to the Middle East, or we have relocated them to Australia or wherever. Some of them to pick up different skill sets, others to just experience a different market...and then if you get them back (which we have done before in instances), they do definitely come back with a much more rounded experience, which I think is always a good thing." (Company J)

ISO Certification

The majority of respondents are ISO certified (8 out of 11) and are audited on a regular basis for re-certification. Professional practices seek ISO certification both as a marketing tool and to "tick the box' when making applications for public sector projects. QS practices without certification, have chosen not to seek it due to the time and resource commitment required for completing the process.

As part of the ISO process, the practice must outline the company mission, objectives and many other issues similar to those associated with strategic planning, however, what is of interest is that for certification the practice must have a business plan. On questioning respondents as to whether they had a strategic plan some confirmed that they did not, of which two are ISO certified. Based on the assumption that they would not be awarded ISO certification without a business plan, one may conclude that they have formulated a business plan for ISO purposes but do not consider it to be a strategic plan. Notwithstanding this, the requirements for ISO certification provide a structure for systematically analysing core aspects of the business which may otherwise not exist.

ISO certification represents a standard of quality. The quality of service provision in a private quantity surveying practice is essential for repeat business. Concern was raised regarding recent trends in fee levels and the implications on quality of service provided (see section 9.1.2.2.). Word of mouth is also a fundamental part of the marketing effort, thus the possible reduction in service quality may seriously impact on the level of repeat business. The quality of service provision as a differentiating factor was also noted by Jennings and Betts (1996).

Marketing

Reputation and word of mouth are the key marketing tools used by QS practice (Boon, 1996; Green et al, 2008). Maintaining a profile within the profession is considered important and may involve attending SCS functions, playing golf with clients (or indeed offering a client tickets to the opera) and keeping a profile in the national media.

Two practices publish an annual review of the construction industry which is compiled in part to address a gap in industry analysis but also as a marketing tool. Both practices invite industry and government representatives to the reception held at the publication of its review.

In two participating practices established for over 50 years, the long history of operating within the industry contributes positively to the reputation of the practice. The number of years in business and the reputation and experience of the longer established practices has been noted positively in a number of previous studies within the construction industry (Green et al. 2008; Jennings and Betts, 1996)

Marketing for the profession as a whole is undertaken by the SCS, however, this is one

of the weaknesses currently identified in the professional body, according to some respondents. This issue will become increasingly important with the downturn in the industry.

Information Technology (IT)

There is evidence of an increasing dependence on IT in the construction industry and in the QS profession. It is generally accepted that IT creates efficiency in undertaking a project, and it is recognised that IT is an essential form of communication.

The use and extent of dependency is not uniform between practices. In fact, many remain sceptical regarding the use of IT for undertaking profession-specific tasks and hence they continue to require paper copies of drawings and correspondence. Whilst at the other end of the spectrum, many practices have an almost total dependence on the use of IT in completing projects as well as in practice management.

Use of IT is influenced by the principals' attitude towards it. This is particularly apparent in cases where IT is used extensively, whereby initiatives are driven by the Director. One practice has developed an online collaboration website used by all construction team members which has resulted in improved communication, efficient meetings, elimination of need for minutes and a data bank of information. In two other instances, detailed cost databases are maintained with trends in construction cost and construction prices tracked and indices produced and published accordingly.

Overall, the use and dependence on IT increases with company size. This is unsurprising given the nature of the services provided, size and complexity of projects undertaken and the number of staff involved in larger practices. Smaller practices on the other hand rely on informal communication systems and personal involvement at Director level, which is simply not possible for larger practices. Although not specifically discussed, it is fair to assume that larger practices have the financial capacity to invest in IT, and enjoy economies of scale from so-doing which smaller practices may not benefit from to the same extent.

Finance

The reduction in construction activity resulting in fierce price competition has resulted

in practice finances coming under scrutiny. Financial efficiencies are sought by reducing overheads, which is having particularly far reaching consequences in terms of employment levels, as outlined previously.

Financial turnover is hugely important for firms competing for public sector projects as it is one of the criteria against which a practice is measured during the application process. In some instances, practices are competing aggressively in order to meet turnover thresholds for this reason:

"Turnover is very important, well it's important financially of course but it's even more important when you are applying for Public Sector contracts because there are turnover thresholds that you have got to try and take cognisance of." (Company B)

A decision was made not to probe respondents on financial performance, as the focus of the study relates to strategic planning processes and thus was beyond the scope of the research. Furthermore, given that financial analysis lay beyond the scope of the research, it was deemed unnecessary to seek such sensitive data for fear that it might erect a barrier in answering subsequent (relevant) questions.

9.1.2.4 Strategic Choice

Corporate and business level strategy, like most aspects of the industry, is in a state of transition while the market adjusts to changing circumstances and the challenges therein. Porter's (1980) generic strategies form the basis of analysis of strategic choice. Corporate and business level strategies pursued by respondents vary despite the underlying objective of survival being apparent.

Corporate Strategy

The overriding corporate strategy being pursued in QS practices was survival and in many instances medium term plans were being "*thrown in the bin*" in place of emergency strategising. An important similarity is apparent to the conclusions drawn by Phua (2007) in her empirical study of construction firms, where she revealed "a

conspicuous lack of adoption of high-end strategic functions by firms even when perceived environmental uncertainty is high" (pp. 760).

Despite the number of employees being reduced, retrenchment was not the core corporate strategy being pursued in many cases.

The following table identifies changes in the corporate strategy being pursued:

Table 9.2Change in Corporate Strategy

Company	Past Strategy (Pre 2009)	Current Strategy (July 2009)	
А	Stability	Combination : Domestic market	
		stability; expansion abroad	
В	Stability	Retrenchment	
С	Expansion	Retrenchment	
D	Expansion	Stability	
Е	Stability	Retrenchment	
F	Expansion	Retrenchment	
G	Stability	Retrenchment	
Н	Expansion	Stability	
Ι	Expansion	Combination: Domestic market	
		stability; expansion abroad	
J	Expansion	Stability	
К	Stability	Stability	

A number of notable issues are evident from the table above. First, of the five practices that in the past pursued a strategy of stability, four had made a conscious decision to restrict the size of the practice (section 9.1.1). This group of practices remained small and had no more than twenty employees during the peak of the construction cycle.

A second pattern to be explored involves the relationship between corporate strategy and ownership structure. Three practices have a relationship with a larger UK construction consultancy organisation. In each instance the UK practice is expanding internationally, however, the expansion is not currently reflected in the Irish practice strategy. One could conclude therefore that a combination strategy is being pursued, however, the research at hand is solely related to the Irish practice and consequently the strategy being pursued in the Irish market. All three practices have previously pursued growth strategies domestically; two of the three are currently pursuing a stability strategy while the remaining practice is retrenching in the Irish market.

In the two practices pursuing a combination strategy, the combination is based upon the presumed stability in the domestic market and expansion into international markets. Neither practice referred to in this instance is part of a larger UK construction consultancy group.

Finally, the practices that are now pursuing stability have already downsized operations and have created efficiencies within their firms. To some extent retrenchment has taken place already and the practices are now seeking stability rather than further retrenchment.

Business Level Strategy

Respondents are unanimous in their belief that quality of service provision is crucial for a professional practice and therefore it is no surprise that it is used as a key differentiating factor between competitors. Quality of service provision is inextricably linked to the reputation upon which all practices depend for repeat business (Jennings and Betts, 1996; Maister, 2003).

In smaller practices, the quality of services provided involves direct participation of Directors in client meetings thus offering a personal service. The extent of direct participation of senior management in larger practices was less clear, and despite quality of service being equally important for these practices, it rested upon the reputation of the practice, rather than on personal service.

QS Practices offer broadly similar services to clients, however, larger practices tend to

offer a wider range of services which is a further indication of differentiation. However, it must be noted that the turbulent construction industry environment and the evolving role of the QS within that environment is contributing to the expansion of service provision regardless of practice size.

The dominance of differentiation strategy is clearly evident from the following figure.



Figure 9.6 Business Level Strategy

The dominance of a differentiation strategy lends support to the findings of Jennings and Betts (1996). The importance of providing a personal service and clearly identifying clients' needs are particularly important in the smaller firms.

Another key issue in this regard is that the business level strategy being pursued has not changed significantly over time. Considerable emphasis continues to be placed on differentiation which is likely to continue into the future. One exception lies with the practice that has focused on servicing the needs of a single client which is unlikely to be sustainable in the long run.

Two practices are classified as being "stuck in the middle' and in one instance market segmentation (between domestic and international) is possible and consequently, despite caution being expressed in pursuance of this strategy, the practice in question is well placed to achieve a successful outcome. The other practice is simultaneously embarking on a cost leadership as well as differentiation strategy within the domestic market. The driver behind the combination is to increase turnover such that the firm may be better placed to compete for public sector projects (which have turnover thresholds). In this instance it is not classified as ",best-cost' as the tenders being submitted are not necessarily covering costs and the differentiation is not necessarily based on value-added for the client.

Price competition is a feature of the construction market and all QS practices must be competitive in the prices charged for services provided. This is not to be confused with a cost leadership strategy, whereby low cost relative to competitors' costs becomes the over-riding focus of strategy. In fact, some practices do not compete for certain projects due to the low fee level of a project which may result in compromising on the quality of service provided.

"Lifestyle choices' influence the strategic decisions of the principal in a number of firms. This was particularly apparent in the smaller practices where the principal recognised the potential to have grown during the boom, but for various reasons, including lifestyle choices, made the strategic decision to restrict the size of the practices (see above).

"We were prepared to remain a reasonably small outfit that's easy to manage...We were always happy to be at our size, we never wanted to become the biggest practice in town because of the fact that what we felt is that we could manage this size practice without breaking into a sweat" (Company E)

9.1.2.5 Implementation and Feedback

Once strategy is developed it must be implemented and according to Mintzberg (1991), remain sufficiently flexible to allow for changes in the environment.

The discussion with respondents in this area focused on internal communication as well as client feedback. It should be noted that the discussion of this component could equally be relevant to the ,jinternal analysis' section above.

Communication

Respondents outlined the necessity of effective communication to staff which is achieved through regular staff meetings and increasingly through personal communication. Mentoring systems are evident and open plan offices are preferred in many cases to ensure effective communication. The need for personal communication with staff was highlighted (in place of electronic communication) in a number of instances given the current environment and the necessity to communicate sensitive information.

"We certainly would be going back to much more reliance on individual communication." (Company J)

"I would make a point of sitting them down once a month, everybody, and just communicating. This is where we are, this is where we want to be, this is what we need to do, this is what we need to be careful of...and these are the threats and opportunities...They all hate being called in for that ...for the last six months every one (meeting) has been grimmer than the previous." (Company A)

Electronic communication is still important and the use of Blackberry's® (for practices operating abroad in particular), intranet and collaborative systems and remains invaluable particularly to those practices with regional offices:

"I meet with them (regional office team) on a regular basis, they come here occasionally; the entire group is interlinked on IT so they have daily communication with everybody through the Intranet. They can also have access to the company office database information, very importantly. They will be originating submissions, proposals, things of that nature, some of that is centralised through here. (Company H)

In smaller practices, director involvement in client meetings and signing-off on correspondence ensures that they have considerable involvement at an operational level.

"We are fairly open in terms of we all know what's in the office and around the office, the office is run on an open plan fashion and there directors are hands on so the directors have a lot of knowledge about all projects that are ongoing." (Company F)

Another important component in the implementation process is feedback. Client feedback is sought in the majority of cases, which is unsurprising given the requirement from ISO certification. Often client feedback is by exception or as a consequence of non-conformance. The main source of feedback to the profession comes in the form of feedback from government tenders which can be accessed through the Freedom of Information Act (FOI).

Staff feedback in the main is informal, however, in the larger practices there are more formal mechanisms for staff feedback through a staff forum or "quality review group". In one case feedback was solicited from staff such that the practice has been twice awarded as being within the top 50 "Great Places to Work".

Flexibility

The need for flexibility in the industry given its competitive environment is crucial for survival and many respondents raised this issue, unprompted, in several different contexts including the range of services provided to clients:

"We also like to believe that we listen to clients and we're flexible in the way we operate in terms of bending our systems to suit them". (Company B)

"We are very flexible in services that we can offer. We are...just listening to clients and what they want and we'll do it...if there's 20 hours work to be had there, we'll go and we'll do the 20 hours work and "thank you very much." (Company C)

as well as flexibility of staff:

"Well I suppose the fact that everybody is flexible now in salaries... The future of this practice is based on young people here so everybody has to kind of agree that keeping employment levels is one of the objectives but to try and keep it by everybody being flexible and reducing back."(Company E)

and finally in the context of strategic planning:

"You can't make it too rigid, and in the environment we are in at the moment. So it does need to be flexible enough that it can be kept under fairly constant review. I think at the moment, because of business the way it is, you can't have too long-term planning, it has to be very hands on immediate action as opposed to waiting around for the next quarter, you just can't do that any more. So I think the plans do need to be flexible." (Company J)

Recognition of the need for flexibility in all aspects of private QS practices is clearly apparent which concurs with the work of many leading authors in the area (Mintzberg, 1994; Hamel and Prahalad, 1994; Grant, 2003 and Løwendahl, 2005). This is a significant finding in the context of construction professional practices.

9.1.3 Summary

The preceding sections have provided a considerable contribution to knowledge in the structure and strategy of QS practices in Ireland. Respondents for this phase were selected to represent practices of varying sizes such that comparisons are possible across a number of dimensions.

There has been a reduction in the number of people employed within the profession and there is universal agreement that the QS profession is undergoing a period of change. The confluence of issues facing the profession, include a rapid reduction in demand for construction output, the domestic banking crisis, global financial crises, rising national unemployment and the deplorable state of exchequer balances. Respondents were very aware of the key trends in each facet of the macroeconomic and industry environment. Many also realise that now more than ever they need to think and act strategically to survive. The analysis thus gave an insight into the workings of the profession, which to date were unknown.

An aspect for consideration in the profession is the size and ownership structure of participating practices which influences the strategy pursued at both corporate and business levels. The link to a larger international construction consultancy practice or lack thereof impacts the strategic planning process and thus forms the initial basis for the determination of distinct groups within the profession (see section 9.2.9 below).

The extent of research and analysis undertaken to inform QS practice strategic choice is varied. Macroeconomic and industry analysis is comprehensive in most instances, however, limited competitor analysis has been undertaken, which is an area of divergence from the strategic management literature. Internal analysis is comprehensive and the recognition of the importance of human resources in particular was evident in each case, which is consistent with the RBV of strategy (Barney, 1991 and Hamel and Prahalad, 1994) and of PSF management (Maister, 2003 and Løwendahl, 2005).

QS practices are in "survival mode'. Stability is the dominant corporate strategy with a number of practices combining stability in the Irish market with expansion abroad. Differentiation is the overriding business level strategy, which has not changed considerably over time. As professional practices, the importance of delivering a quality service to clients is essential in developing a positive reputation. The range of services provided may act as a differentiating factor amongst QS practices with larger practices tending to offer a broader range of services. The extent of change within the profession, however, is contributing to smaller practices expanding the suite of services provided.

Ownership structure and size are critical factors not only in shaping strategy, but also in determining strategic planning process characteristics, which are analysed in detail in the following section.

9.2 Strategic Planning Characteristics in Irish QS Practices

The previous section detailed general company information as well as the strategies pursued by QS practices in Ireland. The next area for consideration and analysis concentrated on the strategic planning process within each practice, be the process formal or informal.

A distinction must be made between the formality of the strategic planning process (the existence of a systematic process) and the formality of the written strategic plan. Every practice with ISO certification is required to have a short term (1 year) and medium term (3 year) plan for the company, however, the nature and comprehensiveness of the process of developing the plan is far from uniform amongst respondents. The research at hand is primarily focused on the characteristics, including formality, of the strategic planning process.

The relationship between strategic planning and performance is beyond the scope of the research and is thus excluded from the analysis.

9.2.1 Approach to Strategic Planning

The approach to strategic planning shapes the process by which it is undertaken. Three main approaches to strategic planning have been identified in the literature (see section 2.2.3) which forms the basis of analysis of QS practices. The approach to strategy of phase one respondents is illustrated in the figure below.





As is evident from the above, five practices have a planned approach with a similar number having an emergent approach. Only one practice is deemed to have an entirely resource-based approach, which is an interesting discovery in the context of professional service practices.

The strategist "type' is another important consideration. Miles and Snow (2003) provide four strategic types against which the principals of Irish QS practices are categorised (see section 9.1.2.1). By cross tabulating strategic type with approach to strategy the following conclusions may be drawn:

Strategic Type	Approach to strategy	Company
Prospector	3 planned approach	F, I, J
	1 emergent approach	С
Analyser	2 planned approach	D, H
Defender	1 emergent approach	Α
	1 resource-based approach	Е
Reactor	3 emergent approach	B, G, K

Table 9.3Strategic type and approach to strategy (Phase 1)

The three practices with a planned approach and prospector typology are large practices, two of which are linked to a UK practice, the third is the largest QS practice in Ireland. These practices clearly have the resources and capability to investigate new markets and services, thus earning their classification of "prospector'. Importantly, prospector types are likely to have a planned approach to strategy. The fourth prospector type has an emergent approach. This practice is a medium sized practice with the capability (through use of strategic alliances) of offering specialist services when the client requires it. The approach is categorised as emergent because clients drive the range of services and markets within which projects are undertaken rather than it being as a result of an internal strategic decision.

The two analyser practices are medium to large practices, one of which has a strategic alliance with a UK practice. They both carefully monitor the industry environment prior to making any strategic decisions. They both have structured and systematic mechanisms for doing so, and each has a written strategic plan.

Defender practices vary in their approach to strategy development with a member of one such form admitting that they are "*sheep*" and react to the environment rather than driving change, whilst the another claims that all decisions are based upon internal resources. Neither firm has a written strategic plan.

The final three practices have an emergent approach to strategy which corresponds to reactor type. Changes in approach to strategy development are noted between the two phases of the research in section 9.4.2.1

By cross tabulating risk attitude against approach to strategic planning, some interesting conclusions may be drawn. Perhaps the most notable of these is that those that have an emergent approach to strategy formation are deemed risk averse, and similarly the majority of those that have a planned approach are classified as risk takers.

Confirmation of the relationship is presented in the following figure.





It may be concluded that risk takers do so in a planned way, or similarly that the planned approach allows the practice to take more risks due to a more structured and comprehensive process of strategy formulation being undertaken. Similarly practices that have an emergent strategy generally are risk averse which may be as a result of adopting a "*wait and see*" approach to industry and professional developments.

9.2.2 Strategic Planning and Process Formality

The existence of a formal/written strategic plan is an important characteristic of the strategic planning process itself. Less than fifty percent of QS practices involved in phase one had a written strategic plan. This percentage is considerably less than in other manufacturing and service industries (Koufopoulos and Chryssochoidis, 2000; Stonehouse and Pemberton 2002) as well as in construction contracting organisations (Price et al., 2003).

Although it appears that ISO certification imposes a structure on the strategic planning process, written strategic plans are clearly not the norm in the sample. Eight participants are ISO certified yet only four confirm the existence of a written strategic plan.

In one instance a client required a strategic plan to accompany an application and

consequently the principal downloaded and completed a strategic planning template from the internet and "*cleared it*" with other directors.

Evidence suggests that ISO certification is little more than a "box ticking' exercise, the paperwork for which in one instance was completed by the principal the night before the ISO visit. It therefore may be concluded that the requirements for ISO certification do not form an integral part of a strategic planning process.

The process by which strategic planning is undertaken in QS practices tends to be informal with a top-down approach evident in most cases. The process is more systematic and structured and involves more people in practices that are part of a larger UK construction consultancy practice. In this group, the formal process involves discussions at various levels of the organisation, analysing the organisation's strengths and weaknesses, market analysis, generation of objectives, developing alternative strategies and determining preferred strategy for achieving the objectives.

The strategic planning process when formally applied takes the general form of existing models (Dess, 1993; Higgins, 1981; Scott, 2003). One such example is outlined below:

"The process of the plan started off with a think tank day when each director was asked to present data and market indicators on particular sectors. The meeting, the think tank meeting or day, developed a couple of key objectives which were then distilled down into the business plan. Exercises involving identification of constraints within the business were also carried out. It also involved some of the more middle level staff, not all directors. So we went through the motions with the yellow stickies and putting them all on the wall and grouping them together under different headings and trying to understand what they all meant...I think that day we actually involved the entire staff. We put everybody into a hotel room and we had a convener with us to help us structure the debate. (Company F)

In the cases where the process is informal, the process is undertaken by the Director(s) in charge and involves little staff participation, which is consistent with the findings of

Jennings and Betts (1996) in particular with smaller QS practices. Despite a number of respondents agreeing that no formal strategic planning is undertaken, it is clear that principals are thinking and acting strategically, even if there is no formal process or indeed written plan. This corresponds to the findings of Hillebrandt and Cannon (1994).

Many reasons were provided for the lack of strategic planning which are presented as constraints on the strategic planning process in section 9.2.6.

9.2.3 Strategic Planning Comprehensiveness

According to Papke-Shields (1997) strategic planning comprehensiveness may be characterised by the:

- exhaustiveness of information gathering process
- thoroughness of organisational evaluation of its present strategic position
- exhaustiveness of environmental evaluation and assessment
- emphasis placed on the generation of alternative strategic options
- extent to which strategic alternatives are evaluated.

There is no distinct pattern as to the comprehensiveness of the strategic planning process based on the above. The comprehensiveness of analysis varies depending on whether it is considered in terms of industry analysis (which tends to be reasonably comprehensive), competitor analysis (which is not comprehensive in most cases) or internal analysis (which is very comprehensive in all cases). Two practices undertake a comprehensive industry analysis which is published on an annual basis and is clearly a comprehensive investigation, however, neither undertake a competitor analysis at all.

The link between strategic planning comprehensiveness and planning formality is also tenuous. In one instance where a formal strategic plan is in place, it was formulated using a template which was not changed significantly, therefore based on the above classification it cannot be considered as "comprehensive". In another case where there is no formal/written strategic plan, significant industry analysis is conducted, discussed at regular meetings with participation from all levels of staff and considered in conjunction with client feedback as well as knowledge gained from active membership of the SCS and CEEC (Conseil Européan des Economistes de la Construction – The European Council of Construction Economists).

It may be concluded that those with a formal strategic plan have undertaken comprehensive analysis (bar one), however, many practices without a written plan have undertaken a similarly rigorous process.

9.2.4 Strategic Planning Flow

Strategic planning flow may be characterised as "top down" or "bottom up". It is unsurprising that in general it was found that the flow of strategic planning comes from the directors of the practice, which confirms findings from Jennings and Betts (1996).

In a number of cases employee and client feedback were incorporated into the process, particularly in practices with a formal strategic plan. Practices with ISO certification must comply with guidelines for client feedback, which in some instances is on an exception-bases only.

In one case the flow or driving force behind the formulation of the strategic plan came from a client requirement but this was the exception rather than the norm.

Where the Irish practice formed part of a larger international group, the process tended to be initiated by head office, although plans were tailored to the Irish market.

9.2.5 Strategic Planning Tools

The use of strategic planning tools and jargon is limited, which concurs with the conclusions drawn by Stonehouse and Pemberton (2002). Where strategic planning tools are used, they tend to be concentrated amongst those practices that engage in a formal strategic planning process, with practices identifying the use of SWOT as the main tool being used.

One practice uses "Balance Scorecard" as a strategic management tool, in particular to identify "red light" areas within the organisation. Scenario planning is also used by many practices, however, it tends to be for tactical rather than strategic purposes. Similarly, a number of practices undertake comprehensive risk analysis on behalf of the client, however, seldom (if ever) is it done for their own organisation.

9.2.6 Constraints to the Strategic Planning Process

Difficulties in the strategic planning process are well documented, particularly from a manufacturing firms' perspective. The following were the key obstacles associated with the strategic planning process for QS practices:

1. Formalising what is known intuitively:

"The hardest thing is translating those general intuitive understandings into a formalised set of goals or objectives." (Company A)

2. Time Constraints:

"The trouble with this whole situation at the moment is that you're so busy just doing quantity surveying that you don't get a chance to do any navel gazing." (Company B)

"The primary difficulty at the time was over-commitment of individuals. It took longer than expected to put together because people were too busy servicing their project work and couldn't devote the time required towards doing the interim back of house work. So that was probably the most difficult part." (Company F)

3. Lack of agreements on the way forward for the practice:

"Getting agreement on what the priority issues are to follow up on. A lot of people I suppose personalities bring different general not agendas, but different insights and they are convinced maybe on their own so we were trying to give them some sense of the best way forward...So it was getting agreement on what were the priorities, that was the main difficulty." (Company D) 4. Lack of commitment from senior staff:

"The MD wasn't comfortable with the way it was going and he believed that it was too slow, the management consultants would argue if they were given the chance, that he wasn't feeding them with the necessary information to allow them to move it faster. So, and there was an element of truth in both sets of opinions in my opinion!" (Company G)

9.2.7 Summary

The preceding analysis highlighted the characteristics of the strategic planning process within QS practices in Ireland. Despite the structure imposed by ISO requirements, it is apparent that formal/written strategic plans are not the norm within the sample.

The size and ownership structure is the overriding factor influencing the extent of strategic planning, with practices forming part of a larger international practice more likely to have a structured process, widespread participation and written plan. Where this is the case, the process and the plan conform to existing planning conventions.

The majority of practices do not have a written plan, and the strategic decision making process is driven by the principal in the firm. Limited participation from staff is evident and the process on the whole is unstructured. However, environmental analysis is reasonably comprehensive albeit undertaken on an individual rather than on an organisational level.

There are a number of commonalities across all practices, the most notable of which is the absence of competitor analysis in the strategic decision making process. Furthermore there is a limited use of strategic management tools amongst all firms, and where such tools are used they are at a project level, hence they are used for tactical rather than for strategic purposes.

It is clear that the foregoing analysis provides considerable insight into the strategic planning process and practices of Irish QS firms. Further scrutiny confirms the existence of distinct groups within the profession which are considered in the following sections.

9.2.8 Quantity Surveying Practice Groups

The preceding sections have provided a comprehensive analysis of the type, scope and extent of strategic planning in Irish QS practices. A number of patterns have become apparent which allow for the clustering of firms into distinct groups across which themes are further explored and comparisons made between groups. The categories within which the groups are analysed were derived from the EBS SPPM framework, and include specific items which emerged during the interviews. Furthermore, the categories were kept consistent for the second phase of research for comparative purposes. The following table outlines the categories against which cross case comparisons were made. Three distinct groups become apparent within the context of Irish QS practices.
Table 9.4Phase 1 QS Practice Groups

Characteristic	Group A	Group B	Group C	Devient
Company	F, J	C,D,H, I	A, B, E, K	G
Link with UK practice	Yes	No	No	No
Strategist	2 x Prospector	2 x Prospector 2 x Analyser	2 x Defender 2 x Reactor	Reactor
Risk attitude	1 x Taker 1 x Neutral	2x Taker 1 x Neutral 1 x Averse	1x Taker 3 x Averse	Averse
ISO	2 x Yes	3 Yes 1 No	3 Yes 1 No	No
Macro analysis	Comprehensive	Comprehensive	Cursory	Cursory
Industry analysis	Comprehensive	Comprehensive	Cursory	Cursory
Internal analysis	Comprehensive	Comprehensive	Comprehensive	Cursory
Competitive Position	Large Projects	Large Projects	Small-Medium Projects	Single client
Corporate Strategy (Past)	Expansion	Expansion	Stability	Stability
Corporate Strategy (Present)	1 x Stability 1 x Retrench	1 x Retrench 2 x Stability 1 x Combination	1 x Combination 2 x Retrench 1 x Stability	Retrench
Business strategy	Differentiate	3 x Differentiate 1 x Stuck in middle	3 x Differentiate 1 x Stuck in middle	Focus
Strategic Variation	Diversify International connections	2 x Diversify 2 x International	1 x international 3 x partnership	None
Written Strategic Plan	Yes	2 x Yes 2 x No	No	No
Process Formality	Formal	Informal	Informal	Informal
Participation in process	Company-wide	Company-wide	Director driven	None
Integration	Fully	Fully	Finance/ operations	Not integrated
Approach	Planned	3 x Planned 1 x Emergent	3x Emergent 1 x Resource	Emergent

Group A practices are part of a global organisation. The group is characterised by a formal strategic planning process with participation from across the company, comprehensive industry and economic environmental analysis and the strategist displaying "prospector" characteristics. In each instance, Group A practices pursued an expansion strategy in the past and are currently seeking stability. Growth has been achieved through mergers and acquisitions as well as through the diversification of services provided. The practices tend to be organised across divisions specialising in a particular area. As a consequence of the link to a global organisation, the practices have the ability to undertake construction projects of any size and can draw upon their partners expertise in a wide range of specialisms. Furthermore, the global nature of the larger practice enables it to potentially offer services to Irish clients pursuing opportunities abroad, however it also restricts it's ability to investigate foreign markets independently of the parent company.

Group B practices are medium to large indegenous firms that agressively expanded their business in the construction boom through diversifying the range of services offered. As a consequence, these practices grew significantly during that period, and have subsequently reduced in size with the downturn in the economy. Growth has been primarily focused on the domestic market, however, one company has offices in a number of cities overseas. These practices have grown to the extent that they possess the capability of undertaking projects of various sizes across the construction sectors. The strategist's approach, and indeed risk attitude, is more varied within this group and consquently the current strategy being pursued also varies. The strategic planning process is considerably more informal compared to Group A practices. Comprehensive industry analysis is undertaken and in one instance a construction industry analysis as well as cost indices are published annually.

One practice within group B is notable due to the existence of a formal strategic alliance with a large international construction consultancy. The practice displays many similarities to those in strategic group A in that the strategic planning process is formal, involves participation at all levels of staff, facilitated by an external consultant and is reviewed regularly. Strategic planning processes were in situ in this firm prior to the alliance being formed. The alliance is considered something that can be "tapped into' when needs be and the use of the alliance services is perceived to be a useful "option"

for the Irish practice. The practice is independently owned with no equity shared between the Irish and the UK practice, thus it is categorised in group B.

Group C includes practices that have consciously remained of a certain size (less than 20 employees) with the strategist tending towards "defender" or "reactor". In most instances, a strategic decision was made to restrict the size of the practice even when the potential existed to expand, and it is no coincidence, therefore, that risk aversion is the dominant attitude of the principal in the firm. Lifestyle choices were cited as being an important influence in the decision making process. Organisational structure tends to be "flat" given the size of the practices. Although comprehensive internal analysis is conducted only cursory analysis is conducted of the external environment. Interestingly, although past corporate strategy was predominantly concerned with stability, the current corporate strategy varies amongst the members of this group, demonstrating alternative mechanisms being utilised for survival. The strategic planning process is driven by the strategist and is informal in all cases with limited participation from staff in the process. Significant emphasis is placed on the importance of staff which would usually be reflected in a resource based approach to strategy, particularly in a PSF. Members of this group acknowledge that they do not drive but rather react to market trends, consequently the approach to strategy development is primarily emergent.

One participant differed significantly from the others in a number of ways, centring on the fact that it:

- does not hold chartered status with the SCS
- focused on a single client for at least the last decade.

The case is therefore classified as a "deviant" case. The strategy of the practice rested entirely with the managing partner, who was close to retirement and who allegedly shows a limited desire to make significant changes. A single client has been the overriding source of business for a considerable amount of time and the fortune of the practice therefore lies with this client. Unfortunately the client is involved in a sector that has seen a dramatic decline in demand, and the QS practice now has little experience in undertaking work outside of this sector. This in turn leaves the practice in a very weak position when submitting applications for work for which there is already intense competition. No analysis of the macro economy, competition or industry is conducted, and this case differs from others in that internal analysis is also lacking according to the respondent and the future of this practice is the most uncertain under the present economic circumstances (see section 9.4.2.1 for developments between both phases of research).

9.2.8.1 Comparative Analysis of Groups

The diagram below sets out some of the key linkages between groups.



Figure 9.9 Links between groups

The overriding concern for all practices is survival through the recession. Given the intense competition in light of reducing demand, all practices are seeking to

differentiate their service from competitors by offering superior quality. It is noted by many that this is increasingly difficult to achieve given the sharp fall in fee levels, the new form of government contract and competitive tendering in particular.

Groups A and B share many common characteristics in that they tended to grow quite rapidly during the construction boom in terms of number of employees as well as in the range of services provided. Both are capable of undertaking large projects and are often in direct competition with each other. The key difference between the groups being the existence of a link with a global construction cost consultancy. Strategic planning tools such as scenario planning and risk management are used, however, they tend to be for tactical rather than strategic purposes. One Group A practice is exceptional in that it uses "Balanced Scorecard" which clearly has operational as well as strategic application.

Practices within Groups B and C are indigenous and much of the growth that has occurred has followed this pattern. The approach to strategic planning in both cases tends to be informal, with few practices having a formal strategic plan in place outside of ISO requirements. Particularly in the cases of Group C practices (although it could be equally argued for Group B), much of the success is attributable to the strategist(s) and the personal relationships that have been developed over a number of years with other professionals within the industry as well as clients which is essential in securing the repeat business upon which they rely.

Regardless of classification, all highlight the importance of having quality staff. It is believed that the strength of the practices lies with its people which is not unusual for a PSF. Each practice invests time and financial resources in the training and development of staff which is a requirement for ISO certification and also for the purposes of maintaining chartered status with the SCS. Interestingly, despite the overall reduction in numbers employed, no significant knowledge gap was identified which in many cases may be attributed to the "hands on" management style and consequent involvement at project level.

Another common trait shared by all groups is the absence of any competitor analysis in the strategic planning process with a lack of accurate information as well as being too busy looking after their own practice to worry about the competition being cited as the reasons for such absence. Macroeconomic analysis is prevalent, however whether this has been driven by the current economic environment, or whether the extent of macroeconomic analysis is always at the same level is unclear.

Effective communication and feedback within practices, (both internally from staff as well as externally from clients) is seen as crucial to the workings of the professional practice. To some extent the process external feedback is driven by ISO requirements and internal communication is believed to be even more important in times of uncertainty than before. Meetings are held frequently with staff which is preferred to electronic forms of communication.

The key differentiating factor between groups lies with the existence of a link with a global construction cost consultancy. It has proven crucial as the type, scope and extent of strategic planning is significantly influenced by the link. Those with the link believe it to be invaluable in terms of potentially providing services to Irish clients wishing to work abroad. Those without the link believe they are better placed to explore international markets themselves without the restrictions such a link may impose.

In terms of competitor behaviour, Groups A and B tend to compete directly with each other as both have the capability to undertake large scale projects across all sectors of the industry. Groups A and C on the other hand view themselves as servicing different markets and consequently have not in the past considered themselves as being in direct competition. Groups B and C often compete for similar projects, in particular the larger of the Group C practices. This issue requires ongoing attention as practices reduce in size and projects (of any size) become increasingly scarce.

9.2.8.2 Causal Maps

The use of displays is an important aspect of qualitative data analysis (Miles and Huberman, 1994). The depth of data obtained in this phase of analysis facilitates the development of causal maps along key dimensions of the strategic groups identified.

The causal maps presented have a number of similarities in that the strategist in each case has been trained in Ireland and all practices are members of the SCS. Each strategist has significant experience in the QS profession, albeit in practices of varying sizes. Nonetheless industry experience, training and qualifications, SCS membership and lifestyle choices influence the strategist in each case. Competitor analysis is limited whereas the emphasis on staff is very apparent across all groups. Furthermore, the quality of resources (plus ISO certification in some cases) results in a quality service being provided based on differentiation which ultimately leads to repeat business.

Practices in group A are part of a larger international construction consultancy firm. The parent company forms an integral part of the decision making process and hence it is at the first point on the causal map. The other two strategic groups are indigenous practices therefore no such link exists.

Causal Map Group A

As noted above the first strategic group forms part of a large international practice therefore decisions regarding strategic planning in the Irish practice are influenced by the parent company. Practices within this group display signs of being risk takers but whether this is due to the security enjoyed by being part of a large international practice, or whether this reflects the risk attitude of the strategist remains unclear. However, the risk attitude is related to a prospector type and ultimately leads to an expansionary strategy being pursued.

The strategic planning process is formal (the influence of the parent company is apparent in this regard), and comprehensive industry, macroeconomic and internal analysis is undertaken prior to the development of a formal/written strategic plan. The following figure provides an overview of the decision making process:





Firms included in this group have a planned approach to strategy development which involves many layers of practice staff in the process and again, this is directly driven by parent company requirements. The planned approach is consistent with the structured strategic planning process ultimately leading to a written strategic plan, which is reviewed regularly. Comprehensive industry, macroeconomic and internal analysis is undertaken, with limited emphasis on competitor analysis. All practices are ISO certified which, together with the reputation of the firm of providing a quality service, is the basis of the differentiation strategy.

Staff training and development is prioritised which in this group may include relocating to one of the international offices. The potential to provide employees with a "global" experience is not always realised but is clearly an option open to the Irish practice.

The range of services provided is broad as practices enjoy economies of scope as well as scale in many instances. The influence of head office is once again apparent, as the Irish practice has the potential to (but does not necessarily) offer the full range of services provided by the parent company.

Causal Map Group B

Firms within this group have many similar characteristics to those of strategic group A, with the chief difference being in the lack of an international link. All practices in group B are indigenous firms. The strategist tends to be a prospector willing to take risks, and the overriding corporate strategy has been to expand the practice (although in recent times this has been replaced by stability).

Despite a planned approach to strategy formation being apparent, it does not necessary result in a formal process nor in many cases a written strategic plan, however, comprehensive industry, macroeconomic and internal analysis is undertaken. Human resources are once again critical and business strategy is based on differentiation in terms of quality as well as a broad range of service provision. Participation of staff in the strategic planning process is somewhat more limited than in group A firms in many cases. The following figure illustrates the range of issues pertaining to the strategic planning process of group B.





As is evident from the preceding figure, group B practices display similar characteristics to that of group A, with the exception of parent company influence. However, this is a critical factor in determining the formality of the strategic planning process as well as the existence of a written strategic plan.

Causal Map Group C

Practices within this group have fewer than twenty employees and are indigenous firms. Many have made the strategic choice to limit the size of the company, even during the construction boom, thus displaying a tendancy towards risk adversity. These practices do not drive the industry or the profession and the strategists tend to be defenders in most cases. The industry influences the strategic direction of the practice and an emergent approach to strategy formation is apparent.

Seldom do these firms have a written/formal strategic plan, nor do they have a systematic process for strategy development. Strategy is decided by the partners/directors of the firm and is communicated ex-post to staff members on an informal basis. Furthermore, although comprehensive internal analysis is apparent, consideration of the industry, competitors and macroeconomic environments is cursory. Seldom are mechanisms or structures in place for the dissemination of such information.

The following figure illustrates the main components of the strategic decision making process for practices in group C.





As may be seen from the previous figure, these practices sought stability, even at the peak of the construction boom. Differentiation is the key strategy for achieving this, however, rather than being based on a broad range of services provided, it is based on offering a quality and personal service to the client, with participation of senior management at project level. Although repeat business is an important feature of any PSF, it is absolutely crucial for this strategic group as it relies on repeat business to survive.

9.2.9 Summary of Phase One – Qualitative Data

The research at hand was undertaken in two phases, qualitative and quantitative, with the purpose of reaching the stated aims and objectives. The aim of the research is:

"To investigate current and recent strategic planning processes evident within Irish Quantity Surveying professional practices."

The research has the following objectives:

- to investigate current and recent trends in strategic planning in Irish QS professional practices
- 2. to evaluate existing strategic planning practises within the Irish QS profession against the strategic management literature
- to develop a set of guidelines to assist systematic strategic planning in Irish QS practices

Phase one of the research addressed objectives one and two above. In order to achieve the objectives, in-depth semi-structured interviews with the principals of eleven Irish QS practices were undertaken. This phase yielded a large amount of data which, heretofore, has not been available on an empirical basis, thus providing a considerable contribution to the body of knowledge in the field.

Strategic planning practices and processes vary amongst QS practices which are influenced by the ownership structure of the firm. Practices linked to a large international construction consultancy practice tend to have a formal process and a written plan which is developed following comprehensive analysis of the environment and with participation from numerous levels of the organisation. Smaller indigenous practices do not have a systematic approach to strategy formation nor do they have a written strategic plan. The decision making process rests with the principal(s) of the practice and an emergent approach to strategy is evident. It may appear that ISO certification imposes a structure on the strategic planning process, however, its use in practice is nominal; formal/written strategic plans are not the norm.

Regardless of the formality of the process it is clear, however, that directors/partners think and act strategically often unbeknown to themselves, which is consistent with the findings of Hillebrandt and Cannon (1994).

Three distinct groups are evident within the profession based largely on ownership structure and practice size, which influences the decision making process. Patterns emerge regarding the type, scope, extent and characteristics of the strategic planning process amongst the groups. These findings represent a significant contribution to knowledge within the field of strategic planning in construction, specifically in QS practices and most notably in an Irish context. At the time of writing, this study was the sole source of such information.

As noted previously, the study was undertaken in two phases. In sections 9.1 and 9.2 the findings from phase one, the qualitative phase are outlined and considered. The following sections present the findings and analysis of a widespread survey conducted on all Quantity Surveying (QS) practices registered in the QS Division of the SCS.

9.3 Phase 2 Findings - Survey

Phase two of the research involved a widespread survey of QS practices registered with the QS Division of the SCS. A list of QS practices was obtained from the SCS and contact was made by telephone prior to undertaking the survey to outline the purpose of the research and to obtain an e-mail address for the principal of the practice. Given the nature of the survey it was deemed appropriate to target senior management likely to be involved in the strategic planning process. An online survey tool was used to collect responses (<u>www.surveymonkey.com</u>), and a response rate of 42% was achieved.

The survey was administered approximately one year after phase one interviews took place and respondents from phase one participated in phase two in order to reflect changes in the intervening period as well as for triangulation purposes. When the survey result did not agree with that which emerged from the interview, contact was made with the respondent to seek clarification. Notable developments in strategy and strategic planning of those that participated in both phases are discussed in section 9.4.

This section provides findings and analysis from the online survey and is presented in the sequence in which the questions were asked in the questionnaire.

9.3.1 General Company Information

A targeted response was sought from senior management within QS practices therefore the overwhelming majority of respondents satisfied this criterion, as may be seen from the diagram below:

Figure 9.13 Position held by respondent



The overwhelming majority of respondents (63% or 39 respondents) were the managing director of the organisation, with 13 respondents being director and 3 at associate

director level. Of the seven respondents who chose the "other" category, two were sole traders, three were senior consultants, one was a partner in a partnership and one was a junior QS.

The majority of practices had been in operation for more than five years with only 21% being in business for shorter periods of time. Future studies could usefully track changes in this regard over time with the number of spin-off practices (resulting from downsizing existing firms) likely to increase.

The majority of practices are independently owned Irish practices (87%), with the remaining 13% forming part of a larger international group and 61% currently having ISO certification. The numbers employed within the profession have changed between the peak and trough of the construction cycle. This is evident from the tables below.

Question: Please indicate the number of people CURRENTLY employed in your organisation					
Answer Options	Response Percent	Response Count			
0-5	66.1%	41			
6-10	12.9%	8			
11-20	9.7%	6			
20-50	6.5%	4			
50-99	3.2%	2			
>100	1.6%	1			
Total number of respondents		62			

Table 9.5Number of people currently employed (2009)

Please indicate the maximum number of people your organisation employed during the PEAK of the construction boom (c. 2007)						
Answer Options Response Response						
	Percent	Count				
0-5	41.0%	25				
6-10	21.3%	13				
11-20	13.1%	8				
20-50	14.8%	9				
50-99	4					
>100 3.3% 2						
Total number of responses to question		61				

Table 9.6Number of people employed at peak (2007)

[One respondent did not answer this question, therefore it is reasonable to assume that the number of people employed has not changed between 2007 and 2009.]

As is evident from the table above, many QS practices have downsized over recent years. The structure of the profession is such that it comprises a proportionately large number of small firms with a proportionately smaller number of large firms. This is consistent with previous findings (FAS, National Skills Bulletin, 2009).

As discussed in chapter 3, a range of services may be provided by a QS practice.

Table 9.7Services currently provided

Key services CURRENTLY provided by the organisation.					
Answer Options	Response Percent	No. of respondents			
Traditional Quantity Surveying Services	96.8%	60			
Construction Management	30.6%	19			
Project Management	75.8%	47			
Health and Safety Management	17.7%	11			
Facilities Management	11.3%	7			
Bank Monitoring	54.8%	34			
Sustainability	22.6%	14			
Other	17.7%	11			
Total number of responses to question62					

The "other" services mentioned by respondents include:

- planning
- expert witness provision
- building surveying
- dispute resolution procedures
- insurance valuations
- BER (Building Energy Rating)

The range of services provided has not changed significantly amongst participating practices (71% of respondents) over the current construction cycle.

A shift in emphasis in the construction sectors serviced is evident. The proportion of work currently being undertaken within the four primary sectors is provided in the figure below:





83% of respondents noted a shift of emphasis in the sectors serviced since the peak of the construction boom. The proportion of work undertaken within each of the primary sectors during the peak of the construction boom is provided in the figure below.

Figure 9.15 Proportion of work undertaken within key sectors at peak



The figures above illustrate the change in emphasis in the residential and private nonresidential sectors particularly. A notable proportion of respondents confirmes that they did/do not undertake projects within the productive infrastructure sector which may be explained by the size of such projects and by the fact that the majority of respondents represented SME's.

9.3.2 Strategic Management Practices

This section of the questionnaire sought information regarding the strategic management practices of respondents. This is an area within which a considerable body of existing knowledge exists in many industry contexts, with the exception of the construction industry, and specifically the QS profession.

9.3.2.1 Strategic Types

The first question was broadly based around the Miles and Snow (2003) typologies. The answer choices thus represented the prospector, analyser, defender and reactor types. In the context of a turbulent environment where many of the forces of change are external to the organisation, it is unsurprising that the reactor typology is commonly found. Furthermore in a rapidly changing environment a defender strategy may be the most appropriate. A summary of the typologies chosen by respondents is illustrated in the following figure:





As is evident from the above there is a broad distribution across strategic types with the exception being prospector types which are noticeably less common within the profession. A possible explanation of the distribution may be that prospector types are the least likely to be found due to the depressed demand, lack of finance available and continued market uncertainty. However, based on the findings of phase one, it may be the case that only a small number of practices (large practices particularly) drive change within the profession, whilst other tend to follow or defend the current market position.

9.3.2.2 Approach to Strategy Formation

The approach to strategy is almost evenly distributed along the three main dimensions. This finding is particularly important when compared to, for example, Green et al. (2008) when it was concluded that amongst contracting firms an emergent approach was predominant.



Figure 9.17 Approach to strategy

An interesting comparison between the strategic typology and approach to strategy formation is provided below.

Table 9.8Strategy type and approach to strategy (Phase 2)

	Approach to strategy formation.				
Strategic Typology	Planned	Emergent	Resource- based.		
Prospector	4	2	3		
Analyser	12	3	6		
Defender	5	4	5		
Reactor	3	10	8		

*Two respondents chose more than one option for "typology" – first chose 2 & 4; second chose all four options.

The preceding table demonstrates a clear tendency towards a planned approach by analyser types and an emergent approach by reactors. Those with a resource based approach have a marginal tendency towards a reactor type, however, the distinction is less apparent. These finding are consistent with phase 1.

Based on the above it is not surprising, therefore, that analyser types with a planned approach are more likely to have a formal/written strategic plan. Evidence of this trend is provided in the table below.

	Approa				
Written strategic plan	Planned Emergent		Resource- based.	Response Percent	
Yes	10	2	5	31%	
No	11	12	15	69%	

Table 9.9 Formality of strategic planning across strategic approach

9.3.2.3 Risk Attitude

Risk attitude was measured by determining the extent of agreement with a number of statements. Disagreement with the statements signals a risk taker attitude.

Responses to the question displayed a tendency towards low-risk projects being explored incrementally, however, not necessarily with a "wait-and-see" approach. A summary of response is provided below where 1=strongly disagree, and 5 = strongly agree.

Table 9.10	Risk attitud	e
-------------------	---------------------	---

Answer Options	1	2	3	4	5	Response count
The organisation has a strong tendency for low-risk projects	2	15	14	26	5	61
Top managers in the organisation believe it is best to explore opportunities gradually via incremental behaviour.	0	11	14	34	2	61
The organisation typically adopts a cautious "wait-and-see" posture.	5	27	10	18	1	61

9.3.2.4 Environmental Analysis

Much of the strategy and strategic planning literature emphasises the importance of undertaking industry, competitor and environmental analysis, in order to successfully compete within an industry. To that end participants were asked to rank the degree of importance of a number of issues derived from the literature on a Likert scale. The assumption behind the question is that the greater the perceived importance the more comprehensive the analysis and attention the issue is given.

Most notable from the responses was the overwhelming agreement on the importance of repeat business (86% selected "very important"). The second largest proportion (46%) confirmed that construction industry analysis is important with a large proportion (42%) of respondents noting the importance of relationships with other industry professionals. Each of these findings is consistent with phase one. Interestingly 33% of respondents deemed competitor analysis to be unimportant. These findings are consistent with the findings from phase one.

The table below provides details of the actual responses to each option presented

Table 9.11Importance of Analysis

Question: Please indicate the degree of importance of each issue below for your							
Where 1 = unimportant and 5 = very important							
Answer Options	1	2	3	4	5		
Analysis of competitors	4	20	20	13	4		
Construction industry analysis	3	10	13	28	7		
Macroeconomic analysis	4	10	26	18	3		
Internal organisation analysis	3	8	17	22	11		
Repeat business	0	0	0	7	54		
Relations with other industry professionals	0	3	8	24	25		
Long range planning	2	9	13	21	15		
Staff training and development	1	3	21	23	13		
Research and development	3	14	15	18	11		
Quality control	0	3	8	23	23		

Clearly repeat business is the single largest influence on QS practices. This is an issue that was apparent from phase one which is now validated based on the results above.

9.3.2.5 Strategic Choice

The corporate strategy currently being pursued by respondents is broadly spread across the four outlined by Porter (1980). Approximately a third of respondents noted that a combination strategy is currently being pursued, with the most frequent combination being stability and expansion (10 respondents) and the second being stability and rationalisation (8 respondents). Perhaps more significantly, only 31% of respondents noted a change in corporate strategy over the recent past.

Business level strategy is heavily concentrated on best cost strategy (59%) with differentiation (39%) being the second most favoured strategy. This is broadly similar to the findings of Boon (1996) in New Zealand QS practices. A number of respondents (17) selected more than one option indicating a "stuck in the middle" (Porter, 1980) strategy. Nearly 30% of those who selected a number of business level strategies noted that three in particular were being pursued; namely:

- Striving to achieve lower overall cost/price than rivals
- Seeking to differentiate our service from rivals to appeal to a broad spectrum of clients
- Providing the best value compared to rivals offering services with similar attributes

Participants were subsequently asked to confirm if the business level strategy had changed in recent years. In 75% of cases the business level strategy had not changed. In the remaining cases a number of explanations were provided to support the need for a change including:

- "Price appears to be the main criteria rather than the quality of service offered."
 (Respondent no. 47. Also participant in phase 1)
- "Fees are being driven down to unsustainable levels due to maintain business. This will certainly result in a reduction in the quality of services offered". (Respondent number 58).

A sizeable number of respondents (53%) affirm no strategic variation was currently being considered, however, 33% confirmed that foreign markets were under

consideration. When specifics were provided, the UK and Middle East were the most common foreign markets being considered.

9.3.2.6 Communication of Strategy

Strategy is most commonly communicated verbally, be it informally or through formal company meetings. Similar findings were observed by Price et al. (2003) within UK construction companies. Seldom is strategy communicated through the distribution of documents, however, in one case it was noted that strategy was inherent in the quality control documents which are available to staff.

A summary of findings regarding communication of strategy is presented in the table below.

Question: Please indicate how your organisations strategy is communicated to firm members.						
Answer Options	Response Percent	Response Count				
Informal discussion/word of mouth, in a broad, general fashion	57.4%	35				
Formal meetings, where firm strategy is communicated verbally	41.0%	25				
Through the distribution of documents explicitly containing firm strategy	6.6%	4				
Other (please specify)	8.2%	5				

Table 9.12Communication of Strategy

The majority of those who chose "other" confirmed that they were sole traders and consequently the communication of strategy to firm members was not applicable.

Findings from this question are consistent with those from phase one.

9.3.2.7 Lifestyle Choices

Lifestyle choices were noted in phase one as playing an important role in the strategic decision making process. The table below confirms the validity of this finding.

Table 9.13Lifestyle choices

Question 22: To what extent have personal lifestyle choices influenced your company's strategic decisions?					
Answer Options Response No. of Respondent					
	Percent				
No influence at all	21.3%	13			
Little influence	24.6%	15			
Some influence	37.7%	23			
Significant influence	16.4%	10			
Please feel free to expand if you wish	7				
Re	esponse Count	61			

Interestingly one respondent involved who participated in both phases discussed the lifestyle influences in strategic decision making during interview but indicated that it had "no influence at all" in the second phase. This reflects the "*survival mode*" noted by the respondent in the questionnaire.

9.3.3 Strategic Planning

The final section of the questionnaire sought to ascertain the strategic planning characteristics of respondents. This section provided a significant contribution to knowledge in the field, albeit with a marginally smaller response rate than previous sections.

9.3.3.1 Strategic Planning Formality

The overwhelming majority of QS practices (69%) do not have a formal/written strategic plan. Of the 17 practices that do have a formal strategic plan, 41% are linked to a larger construction consultancy practice. This is consistent with findings from phase one.

Six of the seventeen practices with a written strategic plan were participants in phase one. In two of these cases no formal strategic plan was in place during the initial phase therefore follow up phone calls were made to these respondents to ascertain the rationale behind the change.

Changes within the industry, as well as the internal organisation in one case were the drivers behind formalising the strategic planning process. To this end a business coach was employed to facilitate the process, as well as to formulate a more targeted marketing campaign, including the development of the company website. The respondent confirmed that discussion of issues pertaining to strategic planning with the researcher during phase one was a contributing factor in highlighting the need to consider the issues in a more systematic/structured fashion.

In the second case, changing ISO requirements had been the main driver in formulating a formal/written strategic plan to encompass more than quality-related matters.

Strategic planning and company size are often correlated, and this sample is no different. The larger practices are more likely to have a written strategic plan. The table below outlines practice size of those that have a written strategic plan. Company size is taken at peak level for this comparison as nearly half of those with a strategic plan confirmed that there is a long history of strategic planning within the company.

Company size at peak	Response percent	Number of
(2007)		respondents
0-5	11.8%	2
6-10	11.8%	2
11-20	17.6%	3
20-50	29.4%	5
50-99	23.5%	4
>100	5.9%	1
Total	100%	17

Table 9.14Company size and formality of strategic plan

As is evident from the above, 76% of QS practices with a written strategic plan employ more than 10 people. This percentage decreases to 47% at current levels of employment.

A number of other conclusions may be drawn as regards QS practices with a written strategic plan, including:

- 59% confirm a planned approach to strategy formation; with 29% having a resourcebased approach
- 76% consider competitor analysis as important (which is proportionately larger than the sample as a whole)
- 65% have plans to operate in a foreign market
- 65% communicate strategy through formal meetings
- 65% have a strategic planning horizon of between 1-3 years
- Market uncertainty was the predominant obstacle to strategic planning formation

Each of the findings listed above is consistent with the conclusions drawn from phase one.

9.3.3.2 Strategic Planning Comprehensiveness

The extent of strategic planning comprehensiveness is a complex matter for analysis. The question, designed to determine the extent of comprehensives, was greatly influenced by the work of Papke-Shields (2006).

Respondents were asked to indicate the extent of agreement/disagreement with a number of statements. The assumption is that the greater the extent of agreement the more comprehensive the process. The following table provides results in this regard, where 1=strongly disagree and 5=strongly agree

Table 9.15Strategic planning comprehensiveness

Answer Options	1	2	3	4	5
Strategic/business planning process is very structured	3	7	22	20	2
Written guidelines exist to structure strategic/business planning	5	19	19	11	0
Outputs of strategic/business planning are formally documented	5	17	16	13	2
We are exhaustive in gathering relevant information	2	22	21	8	0
We determine optimal courses of action from alternatives	1	8	25	19	0
We delay decisions until all alternatives have been evaluated	1	12	19	18	2
The planning flow is "top down"	0	6	14	30	2
Our process of strategic/business planning involves numerous participants at all levels of the organisation	3	13	18	14	3
Participation of (external) specialists in strategic planning is high	6	23	19	4	0
Line managers and staff are involved in the strategic/business planning process	5	12	13	20	2
We constantly evaluate and review strategic plans	1	5	22	22	1
We frequently adjust strategic plans to adapt them to changing conditions	2	4	16	25	4
Strategic planning is a continuous process	1	2	9	31	9

A number of important conclusions, consistent with phase one, may be drawn from the above, namely:

- 1. Written guidelines seldom exist to aid the strategic planning process
- 2. Information gathering is not exhaustive
- 3. Planning flow is top-down with limited participation at other levels
- 4. Participation from external consultants is limited
- 5. Strategic planning is a continuous process which is frequently adapted

9.3.3.3 Experience of Strategic Planning

Strategic planning practices within QS firms in Ireland are clearly emerging phenomena. The driver behind the need for strategic planning is undoubtedly the changing conditions facing the industry (and the economy as a whole). The industry has gone from one where firms faced difficulties associated with rapid growth (e.g. staff shortages) to one characterised by over-capacity and market uncertainty.

The experience in strategic planning practices is considered important with only 24% noting that it is not deemed necessary. The table below demonstrates the experience of respondents with strategic planning.

Answer Options	Response	Response
	Percent	Count
The organisation has no experience of strategic	24.1%	13
planning at present, and it is not deemed necessary.		
Procedures and policies for conducting planning are	40.7%	22
just beginning to emerge.		
Planning activities have been formally developed and	16.7%	9
tested.		
Long history of strategic planning activities.	18.5%	10
Total number of responses		54

Table 9.16Experience with strategic planning

The dominant trend is in emerging strategic planning practices. Of these practices, 73% note that there has been no change in corporate strategy, while 86% note no change in business strategy. The same number of respondents confirm that the company presently does not have a written strategic plan. The conclusion is that the need for strategic planning is becoming apparent not as a consequence of a change in the strategic direction of the firm, but rather from industry, competitor and/or macroeconomic changes.

Practices that do not perceive strategic planning to be important tend to be micro-firms employing less than five people (85% of those that chose the option) without ISO certification (92%) have not changed the range of services provided over the medium term (85%) are reactor types (85%) pursuing a stability strategy (75%) and do not currently have a formal strategic plan (92%).

9.3.3.4 Strategic Plan Content

The content of written strategic plans will vary from one organisation to another. The table below provides a summary of the items most frequently included in the strategic plans of Irish QS practices.

Which of the following are contained within your strategic plan?				
Answer Options	Response	ponse Response		
	Percent	Count		
Mission/vision statement	54.5%	12	5	
Organisation objectives	77.3%	17	1	
Organisation structure	72.7%	16	2	
Sales targets	59.1%	13	4	
Profit targets	50.0%	11	6	
Industry analysis (construction)	22.7%	5	11	
Industry analysis (QS profession)	45.5%	10	7	
Competitor analysis	31.8%	7	10	
Corporate appraisal (e.g. SWOT analysis)	31.8%	7	10	
Organisational core competencies	22.7%	5	11	
Marketing strategy	63.6%	14	3	
Production/operation process	18.2%	4	12	
Quality Assurance process	36.4%	8	9	
Staff appraisal	40.9%	9	8	
Staff training and development	54.5%	12	5	

Table 9.17Strategic plan content

As is evident from the above, organisational objectives and structure are most commonly chosen by respondents, followed by marketing strategy and sales targets. Competitor analysis is less frequently contained in the strategic plan which is in line with the extent of importance being placed upon the issue as concluded in section 9.1.2.2 above.

9.3.3.5 Strategic Plan Time Horizon

The time horizon of formal strategic plans will also vary from one organisation to the next. The table below confirms that the favoured time horizon is between one and three years which verifies the results of previous studies (Stonehouse and Pemberton, 2002) within the context of Irish QS practice.



Figure 9.18 Strategic plan time horizon

9.3.3.6 Obstacles to Strategic Plan Formation

The complexity of the strategic planning process presents many challenges to the formulation of a strategic plan. Respondents were asked the frequency which a number of possible challenges were experienced.

The table below presents the findings in this regard.

Table 9.18 Obstacles to Formulating Strategic Plan

Please indicate the frequency with which each has been experienced by your organisation. Where 1 = Never and 5 = Very frequently					
Answer Options	1	2	3	4	5
Time constraints	0	2	10	7	3
Market uncertainty	0	1	7	11	3
Lack of expertise	2	5	8	5	2
Budget constraints	3	7	6	4	2
Staff resistance to change	5	7	4	3	2
Mgt. resistance to change	7	4	8	2	1

Market uncertainty is the single largest obstacle facing strategic plan formulation, whereas budget constraints are least likely to be an issue. Resistance to change by staff or management is rarely experienced. Interestingly, despite only 17 practices confirming the existence of a written strategic plan, 22 respondents answered this question indicating that the respondents are thinking and acting strategically, however, strategic thinking is not necessarily transformed into a written strategic plan.

9.3.3.7 Strategic Planning Use

The purpose or use to which a strategic plan is put is clearly to guide company operations, and interestingly it is unlikely to be used to obtain funding. Whether this is related to the current environment and the associated difficulty in obtaining funding regardless of the existence of a strategic plan remains unconfirmed.

Table 9.19Use of strategic plan

Please select the use(s) of your organisation's strategic plan. (Note: select as many as are appropriate, as multiple answers will be accepted)				
Answer Options	Response	Response		
	Percent	Count		
To guide company operations	72.7%	16		
To measure company performance	68.2%	15		
To establish incentives for managers	31.8%	7		
To improve allocation of resources	36.4%	8		
To obtain funding	9.1%	2		
To assist in attracting major customers	36.4%	8		
To gain ISO certification	13.6%	3		

Once again it is interesting to note that of the 22 respondents who answered this question, only 17 of them confirmed the existence of a written strategic plan, further supporting the notion that principals are thinking and acting strategically but not necessarily formalising or documenting the process.

9.3.3.8 Difficulties in Strategic Plan Implementation

A number of difficulties may be encountered in the implementation of a strategic plan.

In the case of difficulties experienced within the sample, no single overriding difficulty is evident. Answers tended towards the median, however, in the cases of comprehension of organisational goals, staff capabilities and communication of strategy, there was a stronger weighting towards the negative.
Table 9.20 Difficulties in strategic plan implementation

How often has your organisation experienced the following difficulties in the					
implementation of your strategic p	lan?				
Where 1 = Never and 5 = Very frequently					
Answer Options	1	2	3	4	5
Day to day operations distracted	6	6	14	10	5
attention from implementation					
Implementation took longer than	4	2	17	10	4
anticipated					
The stated objectives were	6	13	13	4	1
unachievable					
Employee's capabilities were	6	19	10	1	0
insufficient					
Communication was inadequate	5	19	11	3	0
Overall goals of strategy were not	6	20	10	1	0
well enough understood by staff					
External factors impacted on	1	6	18	9	2
implementation					
Co-ordination of implementation	4	9	19	5	0
not effective enough					
Unanticipated problems arose	2	7	20	4	3

9.3.4 Miscellaneous Comments Received

Respondents were provided with the opportunity of offering any further comment they wished at the end of the survey but only a small proportion of respondents wished to make such comments. A list of comments received is contained in the following table.

Table 9.21 Additional comments

Comment	Comment	
Number		
1.	This is a family run business - only employing family members	
2.	Some of the questions above are not applicable to the company as a sole	
	trader. I did not want to answer them and give false information	
3.	While there is the requisite expertise in terms of professional QS services,	
	there is only now a process of formal business management being	
	implemented including strategic planning. While there is a broad	
	appreciation of the relevance of strategic planning and its importance,	
	there is a lack of formal knowledge and a concise vision of its application	
	to small/medium professional practices.	
4.	At present, getting paid is far more important and the present poor	
	industry climate makes strategic planning a purely academic exercise	
5.	As a Sole Trader some of the questions are not relevant	
6.	As a small to medium QS practice we cannot obtain public contracts as	
	the current requirements to qualify for Government projects are geared	
	towards the top five QS firms in the country. The majority of small QS	
	firms will have to rely on private work as they will not have the range of	
	previous experience, turnover quality assurance of the bigger firms.	
	Ultimately there will be little competition in 3-5 years as all the smaller	
	players who have not qualified for work over the previous period cannot	
	obtain public contracts. The future of the small QS firm is bleak and has	
	been completely un-aided by the RICS or the SCS. In order to survive we	
	have to become part of a bigger organisation or fall by the wayside. Our	
	strategy is to become an arm of a major UK QS practice.	

9.4 Comparative Analysis of Phase One and Phase Two Findings

The findings from each phase have been presented and analysed in earlier sections of this chapter. The following sections provide a summary of key findings across the two phases of research, and where applicable, developments between phases are also presented.

9.4.1 Triangulation, Validation and Generalisation

A number of findings from phase one of the research were confirmed and reinforced (triangulated) in phase two. Thus triangulation was achieved between the two phases and methods used in the research. The most notable of which include:

9.4.1.2 General:

- Influence of international partner is a key differentiating factor between practices
- Considerable reduction in the number of people employed within QS practices in Ireland
- A change in range of services provided is evident (with firms offering more over time)
- A change in the proportion of work undertaken within sectors serviced is evident (with a shift from private to public)

9.4.1.3 Strategic Management

- Strategists display predominantly analyser and defender types; however evidence of prospector and reactor to a lesser extent
- Repeat business is very important for practices (regardless of size, sectors serviced or ownership structure)
- A majority of practices do not hold ISO certification
- Changes in corporate and business level strategies are evident
- Communication of strategy is informal
- A distinct lack of competitor analysis is undertaken

- Only cursory macroeconomic analysis is undertaken
- Internal analysis is comprehensive in most cases
- Industry analysis is undertaken to varying degrees

9.4.1.4 Strategic Planning

- An informal approach to strategic planning is apparent in the majority of practices
- Strategic plan formality (i.e. existence of a written strategic plan) is correlated to ownership structure
- The strategic planning process tends not to be comprehensive
- Regardless of strategic plan formality, a top down approach to strategy development is evident
- Principals are thinking and acting strategically which is not always transformed into a written strategic plan or formalised process

9.4.2 Key Developments between Phase One and Phase Two

9.4.2.1 General Company Information

The most significant change lies with the deviant case (Company G) from phase one. This company displayed many unique characteristics which at the time were considered critical weaknesses. One key differentiating factor lay in the fact that the company did not hold chartered status with the SCS (as 50% of company directors must be chartered for the practice to hold the title). The flexible approach to undertaking the research determined that a second phase would take place with QS practices holding chartered status; therefore this firm was excluded from the quantitative stage. However, an important development occurred during the intervening period. The respondent for company G and a co-director resigned from Company G (approximately seven months after the interview) to establish a separate QS practice. The newly formed practice is fully chartered and consequently the respondent participated in phase two of the study.

9.4.2.2 Strategic Type

A change in the approach to strategy amongst three practices initially classified as prospectors was evident following phase two.

Table 9.22Changes in strategic type between phases

Company	Example of Prospector Quotes:	Phase 2:
	Phase 1 July 2009	June
		2010
F	"Increased market share primarilyWe sought to achieve the increase in market share by growing the business geographically around the country, and our intention to do that at the time as well as organic growth was by acquisitionWe have been looking at other opportunities in Saudi Arabia but again they are supporting Irish businesses or Irish developers going to that market as distinct from flying out to those markets and trying to cultivate fresh business."	Reactor
Ι	"First of all we have set up a number of centres outside of Ireland; secondly we have followed clients who we've done work for in Ireland abroad; and thirdly we are chasing new clients abroad. And it's very simple; all you have to do is look around the world and see where there's work and that's where you have to go! There's no magic to itIt's fine talking about survival but you have to look beyond that now and start thinking about where we are going to be in 5 years time and I've a strong feeling that we'll have more strings to our bow in 5 years time than we have now and therefore we will be more resilient to further difficulties in the years ahead the world is our oyster and we intend to go out and pick as many pearls as we can!"	Analyser
J	"We would have certainly looked to having wider expertise, specialist services, that's what we have gone for. Some other people I'm sure in the current market are quite happy to completely retrench and become niche players in a smaller, in a very small section of the industry. That's not what we're trying to do we are still trying to be in a position where if there are what we believe are really quality opportunities, we're trying to make sure we're always in a position where we can really compete for those."	Analyser

The table above reflects the shift from prospector types, which is undoubtedly a consequence of the deterioration of the economic environment within which QS

practices are operating and with fewer potential opportunities arising therein (particularly in the domestic market).

It should be noted that Company C was also categorised as a prospector in phase one which was confirmed as being the case in phase two.

9.4.2.3 Corporate Strategy

Another noteworthy development evident between the two phases is reflected in the change in corporate strategy, which is outlined in the table below.

Table 9.23Corporate strategy

Company	Pre- 2009	July 2009	June 2010
А	Stability	Combination	Combination: Maintain/Expand
В	Stability	Retrenchment	Combination: Maintain/Rationalise
С	Expansion	Retrenchment	Retrenchment
D	Expansion	Stability	Combination: Maintain/Expand
Е	Stability	Retrenchment	Retrenchment
F	Expansion	Retrenchment	Stability
G	Stability	Retrenchment	N/A
Н	Expansion	Stability	Combination: Maintain/Rationalise
Ι	Expansion	Combination	Expansion
J	Expansion	Stability	Combination: Maintain/Rationalise
К	Stability	Stability	Combination: Maintain/Rationalise

As may be seen from the table, the corporate strategy currently being pursued by phase one respondents places increasing emphasis on pursuing a combination of strategies. Often part of the combination is centred on exploring overseas markets. Where stability is being pursued it is due to the completion of the retrenchment process: "We were on a growth and acquisition strategy, to gain market and geographical coverage, we then switched to a down sizing, to reflect market and fee activity, we have completed the down size, and are now running a stable smaller business, maintaining income and market share." (Company F)

9.4.2.4 Business Level Strategy

Another change is apparent within the business level strategy being pursued. The table below provides an overview of the change in business strategy being pursued.

Table 9.24Business level strategy

Company	July 2009	June 2010	Stuck in the Middle
А	Differentiation	Differentiation	
В	Stuck in the middle	Stuck in the middle	Cost leader, differentiate & best cost/value provider
С	Differentiation	Stuck in the middle	Differentiate & narrow market segment
D	Differentiation	Differentiation	
Е	Differentiation	Best cost/value provider	
F	Differentiation	Stuck in the middle	Differentiate & best cost/value provider
G	Focus	N/A	
Н	Differentiation	Stuck in the middle	Cost leader & best cost/value
Ι	Stuck in the middle	Best cost/value provider	
J	Differentiation	Best cost/value provider	
К	Differentiation	Stuck in the middle	Cost leader & best cost/value

There is a clear shift in emphasis from differentiation to competing on the basis of offering the best value to clients; it is reasonable to conclude that within the profession, differentiation alone cannot guarantee sustained competitive advantage within a declining market.

9.4.2.5 Strategic Planning Formality

The time lag between phases resulted in a number of changes in the formality of strategic planning within a small number of practices. Where changes were apparent, respondents were contacted by telephone to determine the rationale behind the change.

In one instance due to the change within the industry and the internal organisation, it was deemed necessary to formalise the strategic planning process. To this end a business coach was employed to facilitate the process as well as to formulate a more targeted marketing campaign, including the development of the company website. The respondent noted that the discussion of strategic issues for phase one of the research was also a contributory factor.

A second director explained the trend towards a more comprehensive and formalised process of strategic planning was derived from changes in ISO requirements. The company now had a more structured and comprehensive strategic plan as partial fulfilment of the ISO process.

One practice which during phase one had considered the written strategic plan that was in place, now confirmed that it has been "scrapped" as it was not deemed appropriate within the current industry environment. This practice had formulated the plan based on a template from the internet and consequently the process itself had never been formal.

9.4.3 QS Practice Groups

Phase one of the research resulted in the clustering of practices into three distinct groups. The second phase of the research sought to establish whether findings from phase one were general to the population as a whole. When comparisons were made using criteria applied to the development of strategic groups in the initial phase, triangulation was achieved for a number of the criteria. The following sections provide evidence to support the general existence of distinct groups within the profession.

Group A

The first group of practices were part of a larger international consultancy, displayed a planned, systematic, participative approach to strategic planning. The following table presents an overview of the responses obtained from the survey against the criteria by which strategic group A were analysed.

Table 9.25Comparative Table Group A

	Phase 1 Respondents	Phase 2 Respondents	
	(Responses July 2009)	(Incl. phase 1 June 2010)	
Number of Practices	2 (F,J)	8	
Link with UK practice	Yes	Yes	
Established	2 >50 years	88% > 11 years	
		38% >50 years	
Strategist	Prospector	38% Prospector	
		38% Analyser	
Risk attitude	1 x Taker, 1 x Neutral	75% risk averse	
ISO	Yes	88% yes	
Current Employment	F = 6-10, J = 50-99	25% <10	
(June 2010)		75%>10	
Peak Employment	F = 20-50, J = >100	88% > 20	
Macro Environment	Comprehensive	Comprehensive	
Industry analysis	Comprehensive	Comprehensive	
Internal analysis	Comprehensive	Comprehensive	
Competitor Analysis	Limited	Limited	
		(50% moderately important)	
Corporate Strategy (Past)	Expansion	63% note change	
Corporate Strategy (1 ast)	Expansion	(From expansion & move	
		to public sector)	
Corporate Strategy	1 x Stability	63% stability	
(Present)	1 x Retrench		
Business strategy	Differentiate	88% best value	
Strategic Variation	Diversify	50% operate in foreign	
	International connections	market	
Written Strategic Plan	Yes	75% Yes	
Formality of process	Formal	Formal	
Experience with SP	Long history	88% long history	
Participation	Whole company	Whole company	
Approach	Planned	63% Planned	
Communication	Formal meetings	88% Formal meetings	

Practices within group A are part of an international construction consultancy organisation and employed greater than twenty people at peak. These practices tend to have a planned approach to strategy development, a written strategic plan, company

wide participation in the process and formal communication of the strategy. Comprehensive macroeconomic, industry and internal analysis is evident with limited competitor analysis being undertaken. Corporate strategy historically has centred on expansion and currently stability is the overriding corporate strategy amongst the group. The overwhelming majority of practices in this group are ISO certified.

The findings from phase one are triangulated along ownership structure, strategic planning formality (existence of a written strategic plan), strategic planning characteristics (approach, strategist type, participation, extent of analysis, communication) as well as corporate and business level strategy dimensions.

Group B

The second group identified in phase one were larger established Irish-owned practices. The key differentiating factor between this group and strategic group C is that at peak the practices in group B had at least twenty employees, although this has reduced in many cases due to market conditions. For comparative purposes it is necessary to compare practices based on pre-recession employment levels.

QS practices within group B displayed greater variation in the characteristics of strategic planning practices and processes and this diversity is evident in both phases of the research. There is an even split between practices with a written strategic plan and those without, similarly for the formality of the process itself, however, the vast majority are ISO certified.

There is a tendency for the strategist to be an analyser type, however yet again there is an even divide between risk takers and those who are risk averse which may go some way to explaining the equal divide between practices pursuing an expansionary and a retrenchment strategy. The following table provides the statistics to support these findings.

Phase 1 Respondents Phase 2 Respondents (Responses July 2009) (Incl. phase 1 June 2010) Company C.D.H.I 8 Practices Link with UK practice No No Established All >20 years 85% > 20 years 2 x Prospector 75% Analyser Strategist 2 x Analyser 2 Taker, 1 Neutral Risk attitude 50% Taker, 50% Averse 1 Averse ISO 3 Yes 1 No 88% Yes **Current Employment** 1=11-20; 3 >20 88%>11; 50%>20 All > 20100% >20: 50% > 50 Peak Employment Macro Environ. analysis Comprehensive Comprehensive (75% important) COmprehensive Industry analysis Comprehensive (75% important) Internal analysis Comprehensive Comprehensive (Internal; R&D; Training) **Competitor Analysis** 1 Comprehensive Cursory (50% consider important) 3 Cursory Corporate Strategy (Past) Expansion 63% note a change (No specifics provided) 50% Expand **Corporate Strategy** 1 Retrench (Present) 2 Stability 50% Retrench 1 Combination 3 Differentiate 63% best cost/value Business strategy 1 Stuck in middle 38% Differentiate Strategic Variation 2 Diversify 63% international 2 International 50% yes Written Strategic Plan 2 Yes 2 No 50% no Formality of process 3 x Informal 50% Formal 1 x Formal 50% Informal Experience with Varied Emerging Strategic Planning Limited Limited Participation in process 3 Planned 75% planned Approach 1 Emergent Communication Formal meetings & IT 50% Formal meetings

Table 9.26Comparative Table Group B

QS practices within group B clearly display varied characteristics, however, they are clearly distinct from group A practices as regards ownership structure, and from strategic group C in terms of the size of the practice.

The findings from phase one are triangulated along ownership structure, varied level of strategic planning formality (existence of a written strategic plan), varied strategic planning characteristics (approach, strategist type, participation, extent of analysis, communication) as well as the diversity of corporate and business level strategy pursued.

Group C

The final group includes smaller QS practices which at peak had fewer than twenty employees. Each practice is independently owned and findings from phase two triangulate results from phase one, however, not in every case. For example the majority of phase two respondents have fewer than five employees, are not ISO certified and a notable proportion (35%) deem strategic planning to be unnecessary.

It should be noted that a number of key findings from phase one are triangulated in phase two along the following dimensions (displayed in table 9.27):

- reactor type strategist
- varied approach to strategy development, with notable absence of planned approach
- risk averse
- company size
- cursory macroeconomic , industry and competitor analysis
- comprehensive internal analysis
- unchanged corporate strategy of stability
- importance of differentiation
- lack of written strategic plan
- informal strategic planning process with limited participation
- informal communication of strategy

	Phase 1 Respondents	Phase 2 Respondents	
	(Responses July 2009)	(Incl. phase 1 June 2010)	
Company	A,B,E,K	45 Practices	
Link with UK practice	No	No	
Established	3 practices >20 years	24% <5 years	
	1 > 50 years	33% > 20 years	
Strategist	2 Defender	39% Reactor	
	2 Reactor	27% Analyser	
Risk attitude	1 Taker, 3 Averse	Predominantly averse	
ISO	3 Yes, 1 No	78% No	
Current Employment	50% 11-20	<u>84% ≤ 5</u>	
	50% < 5	16% 6-10	
Peak Employment	50% 6-10	53% <5	
	50% 11-20	29% 6-10	
		18% 11-20	
Macro Envirn. analysis	Cursory	Cursory	
Industry analysis	Cursory	Deemed important	
Internal analysis	Comprehensive	Comprehensive	
Competitor Analysis	Cursory	Cursory	
Corporate Strategy	Stability	Stability	
(Past)		80 % note no change past	
		to present	
Corporate Strategy	1 Combination	56% Stability	
(Present)	2 Retrench	23% Combination	
	1 Stability		
Business strategy	3 Differentiate	52% Best cost/value	
	1 Stuck in the middle	41% Differentiate	
Strategic Variation	1 International	25% International	
	3 Partnership	18% Partnership	
Written Strategic Plan	1 Yes	82% no	
	3 No		
Formality of process	Informal	Informal	
Experience with	Emerging	43% Emerging	
Strategic Planning		35% Deem not necessary	
Participation in process	Limited	Varied	
Approach	1 Resource	39% Resource based	
	3 Emergent	39% Emergent	
Communication	Informal	61% Informal	

Table 9.27Comparative Table Group C

The findings from phase one are triangulated along ownership structure, strategic planning formality (lack of a written strategic plan), strategic planning characteristics (approach, strategist type, participation, extent of analysis, communication) as well as corporate and business level strategy dimensions.

The preceding analysis demonstrates that findings have been triangulated along a number of dimensions, thus providing evidence of the reliability and validity of the study. Given the exploratory nature of the research this serves as an important contribution to knowledge in the field and furthermore, the research is potentially replicable in other PSF both within and outside of the construction industry.

9.5 Strategic Planning Guidelines for Irish QS Practices

The preceding analysis has established that the type, scope and extent of strategic planning in Irish QS practice varies broadly in line with company size and ownership structure, with limited use of strategic planning tools across the profession. The EBS SPPM has potential to provide a useful framework to assist Irish QS practices in the strategic planning process but the framework may appear daunting for non-strategy experts. When the model was presented to two respondents during the pilot test they both found the diagram confusing. During the course of the interview, both respondents demonstrated that decisions were reached broadly in line with components of the model, however, not necessarily in a systematic or structured way. Respondents thus followed what strategic planning literature purports to be important factors for strategic planning, yet they were unaware that they were doing so.

The downturn in the Irish construction industry presents an opportunity to critically evaluate QS practice management and to consider the future direction of the firm, albeit in an extremely turbulent environment. As previously noted, the extent of strategic planning is varied amongst respondents, however, what is evident is that the training and qualifications of a QS in project and risk management must be transferred from the tactical (project-specific) use to the strategic. What are required are some basic guidelines to assist QS practices in the strategic planning process.

1. Participation from all levels of staff in the process is essential

With excess capacity within QS practices, there is ample opportunity for staff to participate in the strategic planning process thus bringing a wide range of opinion, experience and perspectives to the table which will enhance the process. Some mechanisms by which this is currently achieved amongst QS practices include: staff forums, staff research and presentation of a topic of relevance or interest.

The importance of participation in PSF strategic planning has been reiterated by a number of prominent authors in the field (Maister, 2003; Løwendahl, 2005) and must be the foundation upon which the strategic planning process is laid.

2. Determine Organisational Objectives

It is necessary to define the businesses within which the practice would like to operate, including the sectors serviced and services provided. Maister (2003) suggests four key objectives for a professional service firm (client satisfaction, skill building and dissemination of skills, improved productivity resulting in better business). This is a useful starting point, following which it is necessary to:

- set realistic targets
- establish strategic planning team composition and areas of particular responsibility within the process.
- establish a timeframe within which updates will be required.

3. Environmental Analysis:

Macroeconomic and industry analysis is crucial for a construction firm. As part of the undergraduate degree QS professionals are provided with the necessary skill set to undertake such analysis. Tracking trends in key economic indicators (price inflation, demographics, fiscal policy etc.) as well as in industry (tender price inflation, planning permissions, regulations etc.) is crucial for the success of a QS practice, particularly within the context of a turbulent environment.

Many sources of information are available through government and professional body publications which must be monitored continuously. Furthermore, it is beneficial to assign responsibility for collating and distributing pertinent information in this regard to a member of staff. Macroeconomic and industry analysis is critical in assessing strengths, weaknesses, opportunities and threats facing the organisation.

The research has demonstrated that competitor analysis is not undertaken systematically by QS practices in Ireland primarily due to lack of information in this regard. Word of mouth and the "grapevine" are the two main sources of information on competitors which is unlikely to change significantly in the foreseeable future.

4. Internal Analysis: Staff and Knowledge

PSF's depend on skilled and reliable staff. The construction downturn presents an opportunity to:

- develop staff skills. Broaden range of skills in line with service provision, legislation, sustainable development or legal aspects of the business (this is being undertaken very successfully in a number of QS practices through inhouse training programmes and training days).
- establish processes by which skills are disseminated throughout the office (e.g. if you attend a conference, present summary of findings to colleagues or post them on the intranet).
- document project knowledge and experience (document project experiences, positive and negative, to inform similar projects in the future).
- develop communication systems: establish procedures for the recording of data electronically (eg. develop company cost database).

5. Repeat Business

This is the single most important aspect to QS practices yet the analysis of where, why and how to secure repeat business is often considered to a lesser extant than the pursuit of new business (Maister, 2003). Repeat business and marketing the practice are important and it is necessary to:

- follow up with previous clients. Solicit client feedback, and use it!
- ensure excellent communication with clients by supplying timely information, error-free documentation etc.
- market the practice develop and update company website; advertise in industry journals; publish an article in SCS journal or national press.

6. Strategic Choice

The choice of a generic strategy should be decided on the basis of analysis and diagnosis of the industry, economic, competitive and internal environment. It will be influenced to a large extent by the principal of the practice (particularly for the smaller firms) and by the head office (in the case of subsidiary organisations).

7. Monitoring and Implementation

When the strategic plan has been developed, it is necessary to communicate it to all staff and refer to progress at staff meetings on an ongoing basis. The PSF must recognise that strategic planning is a continuous process that requires monitoring and updating while ensuring it maintains a flexibility to adapt to changing environmental circumstances.

8. Facilitator

The use of a facilitator to aid the strategic planning process may be beneficial (Maister, 2003). This is of particular relevance within the QS profession where the training does not include practice management or strategic planning and if it is the firms first time engaging in the process.

9.6 Conclusion

The analysis above demonstrates the comprehensive nature of the study undertaken within the context of Irish QS practices. Mixed methods have been employed to triangulate findings, which is particularly important with an exploratory study where the primary method used is qualitative.

The findings of the research demonstrate strategic awareness amongst QS practices, however, the formality of the strategic planning process varies with company size and ownership structure. QS practices adhere to EBS SPPM in the main, with the exception of competitor analysis which is divergence from the model. QS practices do not undertake competitor analysis due to the difficulty of obtaining information in this regard. A guide has been provided to assist QS strategists in the use of the EBS SPPM and offers some areas for consideration directly related to the profession.

The following chapter draws final conclusions, identifies limitations in the study as well as recommendations for further study.

10. CONCLUSIONS AND RECOMMENDATIONS

10.1 Introduction

Construction is a strategically important industry in Ireland yet there remains a dearth of empirically tested research undertaken on the strategic management and planning processes of organisations working within it. The existing body of strategic planning process literature has been widely applied to manufacturing, however, there remains a paucity of empirically tested data regarding the process within professional service firms.

This study focused on one profession, namely Quantity Surveying (QS), and the findings have provided considerable insight into this previously under-investigated area. Furthermore, the findings may also usefully be applied to other professional organisations within and outside the construction industry facing similar market and environmental conditions as QS practices.

The following sections summarise the key contributions and limitations of the study and offers suggestions for future research within the area.

10.2 Contribution to knowledge

The study provided a considerable contribution to knowledge, which may be considered along a number of dimensions, seven of which are considered in the next number of sections.

10.2.1 Quantity Surveying Profession in Ireland

Prior to this study a notable paucity of empirical investigation had been undertaken within the QS (and indeed construction industry) in Ireland. This study provides important insight regarding the size and structure of the profession which was analysed along a number of dimensions, including the:

- Number of practices in QS division of SCS
- Ownership structure (independently owned and subsidiaries)
- Number of employees within each practice (2007 and 2010)
- Sectors serviced (2007 and 2010)
- Services provided (2007 and 2010)
- Extent of ISO certification
- Number of practices with international interests

During the course of the investigation, participants noted the lack of demographic information available for the profession, which this study has thus addressed. Furthermore, given the economic circumstances under which the study was conducted it is probable that the demographics of the profession will change in the short and medium term. The findings of this study provide a benchmark against which future studies may compare variations in this regard.

10.2.2 Strategic Planning Process in QS Practices

At the time of writing, no existing empirical investigation regarding the strategic planning process within Irish QS practices was known to exist; an important aim of the research was to bridge this gap. The study has provided new information regarding the type, scope and extent of strategic planning in Irish QS practices, several characteristics of the process as well as variations therein.

Following comprehensive analysis (contained in Chapter 9), a number of conclusions may be drawn regarding the strategic planning process within Irish QS practices:

- Strategic plan formality (existence of a written plan) is positively associated with company size
- Strategic planning process formality is positively associated with company size
- Informal strategic planning is undertaken, often unbeknown to the strategist
- Industry, macroeconomic and internal analysis is prevalent, however competitor analysis is seldom carried out
- A top-down approach to the process is prevalent

- Participation tends to be limited
- Time horizon for the planning process is between three and five years
- Strategic planning tools are not used extensively
- Planning is inclined to be tactical (project-based) rather than strategic in nature
- The need for strategic planning in QS practices is beginning to emerge

The final point above is particularly important for future research within the chosen field. Evidence suggests that strategic planning is being given greater consideration within participating professional practices. Scope now exists for further (longitudinal) research to investigate how strategic planning processes emerge over time in response to the eventual upturn in the construction industry.

Furthermore, the lack of experience in strategic planning evident from the research lends support to the development of mechanisms by which professional practitioners are trained (through CPD and/or engaging a facilitator) in strategic planning. This has practical implications to QS and other professional service firms and provides a commercial opportunity in the provision of such training.

10.2.3 Groups within QS Profession in Ireland

Patterns in firm behaviour result in distinct groups becoming apparent which influence the characteristics of the strategic planning process, serving as a notable contribution to knowledge within the profession.

The first group are subsidiaries of large multinational organisations, consequently the strategic planning process tends to be more formalised, result in a written strategic plan and involve a greater number of participants across organisational levels.

The second group are medium to large indigenous practices which are more varied in the approach to strategic planning. Many have a formal process (similar to that in the first group) while an equal number have an informal approach (similar to that of the third group). The final group are small practices wherein strategic planning is informal and unsystematic and seldom results in a written strategic plan. However, as noted in the previous section, there is a growing realisation of the need to engage in a more organised process.

The approach to the strategic planning process is heavily influenced by the strategist. In the first two groups the dominant strategic type (based on the Miles and Snow, 1978, typologies) is prospector with a planned approach to strategy whereas in the third group a defender type with emergent approach is more apparent. This is a key finding in the context of the study which influences the strategic choices made within the QS profession.

10.2.4 Strategic Choice – Past, Present and Future

The current economic climate within which QS practices are operating has affected the choice of strategy being pursued by QS practices. The study revealed that the overriding strategy at present is survival however this has differed amongst the groups in the past. The primary corporate strategy pursued within the first two groups (noted in the previous section) was an expansionary strategy during the construction boom. Practices within the final group sought to defend their market position, keep the organisation at a certain size and rely on repeat clients to generate business. Stability was thus the primary concern of such practices, even when opportunities were available to expand.

The strategic choices made by QS practices have changed in line with changing circumstances and given the cyclical nature of construction, are likely to continue to evolve over time. This study has provided invaluable insight into the strategic choices of the practices offering the opportunity for future investigations to track and monitor changes using this investigation as a yardstick by which future changes may be analysed.

10.2.5 Strategic Planning Guidelines for Professional Practices

Linked to the conclusions presented in section 10.2.2 above, a number of guidelines have been developed to assist QS professionals in the process of strategic planning. It is evident from the study that QS's have the requisite skills for engaging in complex planning however at present it is evident at a project level (tactical) to a greater extent than strategic. Guidance and direction as to how commence the strategic planning process are offered in Chapter 9, providing another important contribution to knowledge.

10.2.6 Tried and Tested Methodology

The chosen methodology for the research was informed not only by existing literature in the field, through pilot testing of both phases of research and also by using phase one to shape the second phase. Few limitations were encountered with the mixed methods employed, and the results generated provided a considerable contribution to knowledge. It is clear that the chosen methodology has been rigorously tested and proven to yield a significant amount of information across a broad spectrum of variables pertaining to strategic planning.

The thoroughness of approach to the investigation results in the potential for the research instruments to be replicated in future studies of professional service firms, both within and outside the construction industry.

10.2.7 Application to other Professional Service Firm's

As noted in the previous section, the research method has potential to be applied across any PSF, so too do many of the guidelines noted in section 10.2.5.

The literature review contained within this study has noted the prominence of manufacturing within strategic planning literature as well as differences in the process between manufacturing and PSF's. This research has provided a contribution to

research of PSF's, in particular QS practices, however, has potential to be applied, compared and analysed across any professional practice.

10.3 Limitations of Study

The study conformed to the rigours necessary for undertaking empirical research at this level, however, a number of possible limitations may be identified.

Research Sample

The study focused on QS practices specifically, and whereas it was necessary to confine the scope to facilitate an in-depth investigation, there are a number of other professionals operating within the industry for which the research could usefully be extended for comparative purposes.

Research Design

The primary research method employed was qualitative, following which quantitative research was undertaken. The qualitative phase of research was undertaken ahead of the quantitative due to the lack of existing empirically tested research in the area.

The nature of an exploratory study requires flexibility in design (Robson, 2002) and the study demonstrates the importance of such flexibility. The research tool has now been tested thus future studies using the same questionnaire may be administered ahead of indepth interviews

Length of Study

Field work was undertaken over a twelve month period during which industry factors were changing rapidly. Future studies could potentially take a longitudinal approach to monitor changes in strategic planning practices over time, and correlate the extent of change against the business cycle.

Researcher Bias

The extent of researcher bias was limited. The researcher was not a QS, nor a member of the SCS, thus she remained objective throughout. Respondents from phase one were

aware of the researcher's independence which probably resulted in a more open discussion in the absence of her affiliation with any relevant professional body.

10.4 Recommendations for Future Research

A number of opportunities for future research have been noted in section 10.2 above, the most notable of which are considered in the following sections

10.4.1 QS Practice Groups

Considerable research has been undertaken in identifying group behaviour (Desarbo et al. 2009;), much of which concentrates on the stability of groups over time (Cool & Schendel, 1987; Mascarenhas & Aaker, 1989). Evidence from the research at hand confirms the existence of distinct groups within the profession which may be further expanded in the context of strategic group behaviour. The measurement criteria by which strategic planning in QS groups were classified provide a useful framework for future research in this regard.

10.4.2 Performance Relationships

The debate regarding the relationship between strategic planning and organisational performance is ongoing (see section table 2.1). The relationship is outside the scope of the research at hand, however, given the contribution to knowledge the study has provided thus far, potential now exits to extend the scope to incorporate performance measures. New insight has been gained through this investigation regarding the structure of the profession, strategic choice and strategic planning process in QS practices, any of which may be used as a criteria against which performance measured.

Furthermore, some research evidence suggests a link between strategic groups and relative performance (Leask & Parker, 2007; Short et al, 1087) which clearly adds another dimension to the analysis. Once again, a future study may usefully explore this relationship in the context of QS (and other) professional practices.

10.4.3 Construction and other Professional Practices

The existence of a number of professional service firms within the industry lends itself to the study being replicated in other such firms. The environmental conditions facing engineering and architectural practices, for example, are similar to those experienced by QS practices, therefore the strategic management of such PSF's are comparable.

The research method has been derived from the extensive literature base, pilot tested, and used in the main study with few limitations having been encountered. Importantly, the research has been designed such that despite it's being applicable to QS practices it is not specific to such firms.

The research instrument is thus ready to be applied to other PSF's providing an opportunity to explore strategic planning processes in a variety of professions. Clearly scope exits on completion of such empirical testing in other professions, to compare the strategic planning process across a range of PSF firms.

Finally, as noted in section 10.2.2, the relative inexperience in strategic planning amongst professionals coupled with the realisation of the need for strategic planning presents a commercial opportunity in the provision of training and consultancy to professional practices.

REFERENCES

Abraham, Stanley C. (2006) Strategic Planning- A Practical Guide for Competitive Success Mason, Ohio: Thompson/South-Western

Ackelsberg, Robert and Arlow, Peter (1985) Small Businesses Do Plan and It Pays Off, Long Range Planning 18(5), pp. 61-67

Andersen, Torben Juul (2000) Strategic Planning, Autonomous Actions and Corporate Performance, Long Range Planning 33(5) pp.184-200

Ansoff, H. Igor (1957) Strategies for Diversification, Harvard Business Review 35(5), pp. 113-124

Ansoff, H. Igor (1970) Does Planning Pay? The Effect of planning on Success of Acquisitions in American Firms, Long Range Planning, 3(2); pp. 2-7

Ansoff, H. Igor (1987) Corporate Strategy, Revised Edition. London, England: Penguin Books

Armstrong, J. Scott (1982) The Value of Formal Planning for Strategic Decisions: Review of Empirical Research, Strategic Management Journal 3(3); pp.197-211

Ashworth, Allan, Hogg, Keith (2002) Willis's Practice and Procedure for the Quantity Surveyor, 12th edition. Blackwell Science Ltd.

Baden-Fuller, C. and Morgan, Mary S. (2010) Business Models as Models, Long Range Planning, 43(2/3); pp. 156-171

Baker, William H.; Lon Addams, H.; and Davis, Brian (1993) Business Planning in Successful Small Firm, Long Range Planning 26(6), pp 82-88 Bansal, Pratima (2005) Evolving Sustainably: A Longitudinal Study of Corporate Sustainable Development, Strategic Management Journal Vol.26 pp. 197-218.

Barney, Jay (1991) Firm Resources and Sustained Competitive Advantage, Journal of Management 17(1); pp.99-120

Barney, Jay (2001) Resource-based theories of competitive advantage: A ten-year retrospective on the resource-based view, Journal of Management 27(6) pp.643-650

Barney, Jay; Wright, Mike and Ketchen, David J. Jr (2001) The resource-based view of the firm: Ten years after 1991, Journal of Management 27(6); pp.625-641

Barrett, Peter (1993) Profitable Practice, Management London: E & FN Spoon

Bausman, D.C (2002) An Empirical investigation of the relationship between strategic planning and performance of large construction firms, PhD Thesis, Heriot-Watt, 52-1019.

Beer, Michael, Voelpel, Sven C., Leibold, Marius and Tekie, Eden B. (2005) Strategic Management as Organisational Learning: Developing Fit and Alignment through a Disciplined Process, Long Range Planning Vol. 38 (5) pp. 445-465

Benes, J. and Diepeveen, W.J (1985) Flexible planning in construction firms, Construction Management and Economics 3(1); pp. 25-31.

Betts, M.; Cher, Lim; Mathur, K. and Ofori, G (1991) Strategies for the construction sector in the information technology era, Construction Management and Economics 9(6), pp. 509-528

Betts, Margin and Ofori, George (1992) Strategic Planning for Competitive Advantage in Construction, Construction Management and Economics 10(6) pp. 511-532

Betts, Margin and Ofori, George (1994) Strategic Planning for competitive advantage in construction: The institutions, Construction Management and Economics 12(3), pp. 203-217

Boon, John (1996) The Management of Quantity Surveying Practices in a Changing Market, Proceedings of the 1996 RICS COBRA Conference

Boon, John (2001) New Zealand Quantity Surveying Practices – Continuing to Adapt in a Changing Environment, Proceedings of the 2001 RICS COBRA Conference held at Glasgow Caledonian University

Boon, John (2008) Management of New Zealand Quantity Surveying Practices: A Longitudinal Study, Proceedings of the 2008 RICS COBRA Conference held at Dublin Institute of Technology 4-5 September

Boyd, Brian K. and Reuning-Elliott, Elke (1998) A Measurement Model of Strategic Planning, Strategic Management Journal Vol 19(2) pp 181-192

Bracker, Jeffrey S. and Pearson, John N. (1986) Planning and Financial Performance of Small, Mature Firms Strategic Management Journal 7 (6) pp. 503-522.

Brews, Peter and Purohit, Devavrat (2007) Strategic Planning in Unstable Environments, Long Range Planning Vol. (2007) pp. 64-83

Brews, Peter and Hunt, Michelle R. (1999) Learning to Plan and Planning to Learn: Resolving the Planning School/Learning School Debate, Strategic Management Journal 20(10); pp.889-913

Brook, Martin (2008) Estimating and Tendering for Construction Work, 4th Edition, Oxford, Butterworth-Heinemann

Bryman, Alan and Bell, Emma (2007) Business Research Methods, 2nd Edition, Oxford; New York, Oxford University Press Caeldries, Filip and van Dierdonck, Roland (1988) How Belgian Businesses Make Strategic Planning Work, Long Range Planning 21(2), pp. 41-51

Carman, James M. and Langeard, Eric (1980) Growth Strategies for Service Firms, Strategic Management Journal, 1(1) pp.7-22

Cartlidge, Duncan (2006) New Aspects of Quantity Surveying Practice, 2nd edition, Oxford, Butterworth-Heinemann

Chaffee, Ellen Earle (1985) Three Models of Strategy, Academy of Management Review 10(1); pp.89-98

Chandler, A.D (1962) Strategy and structure: Chapters in the history of the American industrial enterprise, Cambridge, MA: MIT Press

Chatterjee, Sayan; Wiseman, Robert M.; Fiegenbaum, Avi and Devers, Cynthia E (2003) Integrating Behavioural and Economic Concepts of Risk into Strategic Management: the Twain Shall Meet, Long Range Planning 36(1), pp. 61-79

Chen, Ming-Jer (1996) Competitor Analysis and Interfirm Rivalry: Toward a Theoretical Integration, Academy of Management Review 21(1); pp.100-134

CIB Priority Theme Revaluing Construction: A W065 Organisation and Management of Construction" Perspective, Publication 313 May 2007

Clark, Delwyn (1997) Strategic management tool usage: a comparative study, Strategic Change 6(7), pp.417-427

Clark, Vicki L. Plano and Creswell, John W. (2008) The Mixed Methods Reader, Los Angeles: Sage Publications 2008 Collier, Nardine, Fishwick, Francis and Floyd, Steven W. (2004) Managerial Involvement and Perceptions of Strategy Process, Long Range Planning Vol. 37 (2004) pp.67-83

Conant, Jeffrey S.; Mokwa, Michael P and Varadarajan, P. Rajan (1990) Strategic Types, Distinctive Marketing Competencies and Organisational Performance: A Multiple Measures-Based Study, Strategic Management Journal 11(5); pp.365-383

Conner, Kathleen R. (1991) A Historical Comparison of Resource-Based Theory and Five Schools of Thought Within Industrial Organization Economics: Do We Have a New Theory of the Firm?, Journal of Management 17 (1) pp. 121-154

Conner, Kathleen R and Prahalad, C. K (1996) A Resource-based Theory of the Firm: Knowledge Versus Opportunism, Organization Science 7(5), pp. 477-501

Cooke, Brian and Williams, Peter (2009) Construction Planning, Programming & Control 3rd Edition, Ames, IA. Blackwell Publishing.

Cooper, Donald R. and Schindler, Pamela S. (2001) Business Research Methods, 6th Edition, Boston, Mass., McGraw Hill

Covin, Jeffrey G and Slevin, Dennis P (1989) Strategic Management of Small Firms in Hostile and Benign Environments, Strategic Management Journal 10(1), pp.75-87

Coxe, Weld; Hartung, Nina F.,Hochberg, Hugh; Lewis, Brian J.; Maister, David H.; Mattox, Robert F. and Piven, Peter, A. (1987) Success Strategies for Design Professionals: Superpositioning for Architecture and Engineering Firms, New York: McGraw-Hill Book Company

Creswell, John W. (2007) Qualitative Inquiry and Research Design: Choosing Among 5 Approaches, 2nd Edition, Thousand Oaks, Sage Publications Dansoh, Ayirebi (2005) Strategic Planning Practice of Construction Firms in Ghana, Construction Management and Economics 23(2) pp 163-168.

Davies, Gary; Chun, Rosa and Kamins, Michael A.(2010) Reputation Gaps and the Performance of Service Organisations, Strategic Management Journal 31(5); pp.530-546

Davig, William; Elbert, Norb and Brown, Steve (2004) Implementing a Strategic Planning Model for Small Manufacturing Firms: An Adaptation of the Balanced Scorecard, SAM Advanced Management Journal 69(1); pp.18-24

De Haan, Job; Voordijk, Hans and Joosten, Geert-Jan (2002) Market strategies and core capabilities in the building industry, Construction Management and Economics 20(2), pp. 109-118

Denscombe, Martyn (2002) Ground Rules for Good Research: a 10 point guide for social researchers, Buckingham, Open University Press

Denscombe, Martyn (2010) Ground Rules for Social Research: Guidelines for Good Practice, 2nd ed. England, McGraw Hill; Open University Press

Denzin, Norman K. and Lincoln, Yvonna S.(2005) The Sage Handbook of Qualitative Research, 3rd Ed. Thousand Oaks, California, Sage Publications

Denzin, Norman K. and Lincoln, Yvonna S. (2008), The Landscape of Qualitative Research, Thousand Oaks, California, Sage Publications

Desarbo, Wayne S.; Di Benedetto, C. Anthony; Song, Michael and Sinha, Indrajit (2005) Revisiting the Miles and Snow Strategic Framework: Uncovering Interrelationships between Strategic Types, Capabilities, Environmental Uncertainty and Firm Performance, Strategic Management Journal 26(1), pp. 47-74

Desarbo, Wayne S.; Grewal, Rajdeep and Wind, Gerry (2006) Who Competes with Whom? A Demand-based Perspective for Identifying and Representing Asymmetric Competition, Strategic Management Journal 27(2), pp. 101-129

Desarbo, Wayne, S.; Grewa, Rajdeep and Wang, Rui (2009) Dynamic Strategic Groups: Deriving Spatial Evolutionary Paths, Strategic Management Journal 30(13), pp. 1420-1439

Dess, Gregory, G (1987) Consensus on Strategy Formulation and Organisational Performance: Competitors in a Fragmented Industry, Strategic Management Journal 8(3); pp.259-277

Dess, Gregory G (1993) Strategic Management, McGraw-Hill

Dess, Gregory G.; Lumpkin, G.T and Covin, J.G (1997) Entrepreneurial Strategy Making and Firm Performance: Tests of Contingency and Configurational Models, Strategic Management Journal 18(9), pp.677-695

Dess, Gregory G.; Newport, Stephanie and Rasheed, Abdul M.A. (1993) Configuration Research in Strategic Management: Key Issues and Suggestions, Journal of Management 19(4); pp. 775-595

Dickson, Pat H and Weaver, K. Mark (1997) Environmental Determinants and Individual-Level Moderators of Alliance Use, Academy of Management Journal 40(2); pp.404-425

Dikmen, Irem; Birgonul, M. Talat and Budayan, Cenk (2009) Strategic Group Analysis in the Construction Industry, Journal of Construction Engineering and Management, 135(4), pp. 288-297

Dillman, Don A. (2000) Mail and Internet Surveys: The Tailored Design Method, 2nd Ed. U.S.A, John Wiley & Sons.

DKM Economic Consultants Construction Industry Indicators, Issue 20, May 2010 (www.dkm.ie)

DKM Economic Consultants Annual Construction Industry Review 2009 and Outlook 2010-2012 (<u>www.dkm.ie</u>)

Dyson, R.G. and Foster, M.J. (1982) The Relationship of Participation and Effectiveness in Strategic Planning, Strategic Management Journal 3(1); pp.77-88

Egan, Sir John (1998) Rethinking Construction: Report of the Construction Task Force, London: Her Majesty's Stationary Office (HMSO)

Eisenhardt, Kathleen M. (1989) Making Fast Strategic Decisions in High-Velocity Environments, Academy of Management Journal 32(3); pp.543-576

Eisenhardt, Kathleen M. and Martin, Jeffrey A. (2000) Dynamic Capabilities: What are they? Strategic Management Journal 21(10-11); pp.1105-1121

Elenkov, Detelin S. (1997) Strategic Uncertainty and Environmental Scanning: The Case for Institutional Influences on Scanning Behaviour, Strategic Management Journal 18(4), pp. 287-302

Fahy, Liam; King, William R. and Narayanan Vadake K. (1981) Environmental Scanning and Forecasting in Strategic Planning – The State of the Art, Long Range Planning 14(1), pp. 32-39

Falshaw, J. Richard; Glaister, Keith W and Tatoglu, Ekrem Evidence on formal strategic planning and company performance, Management Decision, 44(1); pp. 9-30

FAS, National Skills Bulletin 2009

Flick, Uwe (2002) An Introduction to Qualitative Research, London, Sage Publications

Fredrickson, James W (1984) The Comprehensiveness of Strategic Decision Processes: Extension, Observations, Future Directions, Academy of Management Journal 27(3); pp.445-466

Fredrickson, James W. and Mitchell, Terence R (1984) Strategic Decision Processes: Comprehensiveness and Performance in an Industry with an Unstable Environment, Academy of Management Journal 27(2); pp.399-423

Frost, Frederick, A (2003) The use of strategic tools by small and medium-sized enterprises: an Australasian study, Strategic Change, 12(1); pp. 49-62

Green, Stuart D.; Larsen, Graeme D and Chung-Chin, Kao (2008) Competitive strategy revisited: concepts and dynamic capabilities, Construction Management and Economics 26(1) pp. 63-78

Gibbons, Patrick T. and O'Connor, Tony (2005) Influences on Strategic Planning Process among Irish SME's, Journal of Small Business Management 43(2), pp. 170-186

Glaister, Keith W. and Falshaw, Richard (1999) Strategic Planning: Still Going Strong?, Long Range Planning 32(1), pp. 107-116

Grant, Robert M. (1991) Porter's "Competitive Advantage of Nations": An assessment, Strategic Management Journal 12(7); pp.535-538

Grant, Robert M (2003) Strategic Planning in a Turbulent Environment: Evidence from the Oil Majors, Strategic Management Journal 24(6), pp. 491-517

Green, Stuart D., Larsen, Graeme D and Kao, Chung-Chin (2008) Competitive strategy revisited: contested concepts and dynamic capabilities, Construction Management and Economics 26(1), pp.63-78
Greenwald, Royston; Li, Stan X; Prakash, Rajshree and Deephouse, David L. (2005) Reputation, Diversification and Organizational Explanations of Performance in Professional Service Firms, Organization Science 16(6), pp.661-673

Greenwald, Bruce and Kahn, Judd (2005) All Strategy is Local, Harvard Business Review 83 (9), pp.94-104

Griffith, Alan and Watson, Paul (2004) Construction Management Principles and Practice, Basingstoke, Palgrave Macmillan

Grönroos, Christian (2007) Service Management and Marketing: Customer Management in Service Competition, 3rd Edition, West Sussex, England, John Wiley & Sons.

Gubrium, Jaber and Holstein, James A. (2002) Handbook of Interview Research: Context and Method, Thousand Oaks, California, Sage Publications

Gummesson, Evert (1979) The Marketing of Professional Services – An Organisational Dilemma European, Journal of Marketing 13(5), pp. 308-318

Hackley, Chris (2003) Doing Research Projects in Marketing, Management and Consumer Research, London, Routledge,

Hamel, Gary and Prahalad, C.K. (1994) Competing for the Future, Harvard Business School Press

Higgins, Richard B. (1981) Long Range Planning in the Mature Corporation, Strategic Management Journal 2(3), pp. 235-250

Hiles, Dave (1999) Paradigms Lost – Paradigms Regained A summary of conference proceedings, 18th International Human Science Research Conference, Sheffield, UK. July 26-29, 1999

Hillebrandt, Patricia M and Cannon, Jacqueline (1989) The Management of Construction Firms: Aspects of Theory, London, Macmillan Press Ltd.

Hillebrandt, Patricia M and Cannon, Jacqueline (1994) The Modern Construction Firm, 2nd Ed. London, Macmillan Press Ltd.

Hillebrandt, Patricia M; Cannon, Jacqueline and Lansley, Peter (1995) The Construction Company in and out of Recession, London, Macmillan Press Ltd.

Hillebrandt, Patricia (2000) Economic Theory and the Construction Firm, 3rd Ed. Basingstoke, Macmillan Press Ltd.

Hore, Alan V.; O'Kelly, Maria and Scully, Rita (2009) Seeley and Winfield's Building Quantities Explaining Irish Edition, Basingstoke, Palgrave Macmillan

Hoskisson, Robert E.; Hitt, Michael A.; Wan, William P and Yiu, Daphne (1999) Theory and research in strategic management: Swings of a pendulum, Journal of Management 25(3); pp.417-456

Hoxely, Mike; Knight, Andrew and Grada, Mohieddin (2007) Construction Professional Services: A Process Model, Proceedings of the 2007 RICS COBRA Conference held in Georgia Tech., Atlanta USA 6-7 September

Huff, Anne S. and Reger, Rhonda Kay (1987) A Review of Strategic Process Research, Journal of Management 13(2); pp.211-236

Hult, G., Thomas; Ketchen, David J. Jnr. And Arrfelt, Mathias (2007) Strategic Supply Chain Management: Improving Performance Through a Culture of Competitiveness and Knowledge Development, Strategic Management Journal 28(10) pp. 1035-1052

Hutchinson, Colin (1992) Corporate Strategy and the Environment, Long Range Planning 25(4) pp. 4-9

Jennings, Martin J. and Betts, Martin (1996) Competitive strategy for quantity surveying practices: the importance of information technology, Journal of Engineering, Construction and Architectural Management 3(3) pp. 163-186

Johnstone, Steward; Dainty, Andrew and Wilkinson, Adrian (2008) In search of "product-service": evidence from aerospace, construction and engineering, The Service Industries Journal 28(6); pp.861-875

Junnonen, J.M. (1998) Strategy formulation in construction firms Engineering, Construction and Architectural Management 5(2), pp.107-114

Kaplan, Robert S. and Norton, David P. (1996) Using the Balanced Scorecard as a Strategic Management System, Harvard Business Review, 74(1); pp. 75-85

Kaplan, Robert S. and Norton, David P (2005) The Office of Strategy Management, Harvard Business Review 83(10) pp.72-80

King, William R. (1983) Evaluating Strategic Planning Systems, Strategic Management Journal 4(3); pp.263-277

Koufopoulos, Dimitrios N. and Chryssochoidis, George M (2000) The effects of an uncertain country environment upon leadership and strategic planning practices, Strategic Change 9(5), pp.379-395

Kraaijenbrink, Jeroen; Spender, J.C and Grown, Aard J. (2010) The Resource-Based View: A Review and Assessment of its Critiques, Journal of Management 36(1); pp349-372

Lanford, David and Male, Steven (2001) Strategic Management in Construction, 2nd ed. Oxford, Blackwell Science Ltd.

Lanford, D.A. and Retik, A. (1996) The Organization and Management of Construction: Shaping Theory and Practice, London, E&FN Spon Langley, Ann (1988) The Roles of Formal Strategic Planning, Long Range Planning 21(3), pp.40-50

Lansley, Peter (1983) A practical approach to auditing organisational flexibility, Construction Management and Economics 1(2); pp.145-156

Lansley, Peter R. (1987) Corporate Strategy and Survival in the UK Construction Industry, Construction Management and Economics 5(2) pp. 141-155

Larsen, P, Tonge, R. Ito, M. (1998) The strategic planning process in growing companies, Journal of General Management 24(1), pp. 53-68

Larsen, Povl; Tonge, Richard and Ito, Midori (2000) Managing the Strategic Planning Process: a comparative analysis between high-growth medium-sized enterprises and the general business population, Journal of Applied Management Studies 9(2), pp. 275-282

Latham, Sir Michael (1994) Constructing the Team London, Her Majesty's Stationary Office (HMSO)

Lavender, Stephen (1996) Management for the Construction Industry, Harlow, Pearson Education

Leask, Graham and Parker, David (2007) Strategic groups, competitive groups and performance within the UK pharmaceutical industry: improving our understanding of the competitive process, Strategic Management Journal 28(7), pp. 723-745

LeBreton, Preston P. and Henning, Dale A. (1961) Planning Theory New jersey, Prentice Hall

Lockett, Andy and Thompson, Steve (2001) The resource-based view and economics, Journal of Management 27(6); pp.723-754

London, Kerry A. and Kenley, Russell (2001) An industrial organization economic supply chain approach for the construction industry: a review, Construction Management and Economics 19(8) pp. 777-788

Løwendahl, Bente R. (2005) Professional Service Firms, 3rd Ed. Copenhagen: Copenhagen Business School Press

Mahon, John F. and Murray, Edwin A Jr. (1981) Strategic Planning for Regulated Companies, Strategic Management Journal 2(3); pp.251-262

Maister, David H. (2003) Managing the Professional Service Firm, London: Simon & Schuster

Maloney, W.F (1997) Strategic planning for human resources management in construction, Journal of Management in Engineering 13(3); pp. 49-56

Mankins, Michael C. and Steele, Richard (2006) Stop Making Plans Start Making Decisions, Harvard Business Review 84(1) pp.76-84

Markides, Costas (2004) What is strategy and how do you know if you have one?, Business Strategy Review 15(2); pp.5-12

Mathews, Charles H. and Scott, Suzanne G. (1995) Uncertainty and Planning in Small and Entrepreneurial Firms: An Empirical Assessment, Journal of Small Business Management 33(4) pp 34-51

McGee, John and Thomas, Howard (1986) Strategic Management Research: A European Perspective, Chichester John Wiley & Sons.,

Miles, Mathew B. and Huberman, A. Michael (1994) Qualitative Data Analysis 2nd Edition, Thousand Oaks, California, Sage Publications

Miles, Raymond E. and Snow, Charles C (1978) Organisational Strategy Structure and Process, Stanford Business Classics, 2003 edition California, Standford Business Books. Originally printed 1978 New York: McGraw Hill

Miller, Danny (1987) Strategy Making and Structure: Analysis and Implications for Performance, Academy of Management Journal 20(1); pp. 7-32

Miller, Danny; Dröge, Cornelia and Toulouse, Jean-Marie (1988) Strategic Process and Content as Mediators Between Organizational Context and Structure, Academy of Management Journal, 31(3); pp. 544-569

Mintzberg, Henry (1990) The design school: Reconsidering the basic premises of strategic management, Strategic Management Journal 11(3), pp. 171-195

Mintzberg, Henry and Quinn, James Brian (1991) The strategy process: concepts, contexts and cases, Englewood Cliffs, NJ: Prentice-Hall.

Mintzberg, Henry (1994) The Rise and Fall of Strategic Planning, Harlow, Pearson Education Limited

Mintzberg, Henry (1994) The Fall and Rise of Strategic Planning, Harvard Business Review 72(1), pp. 107-114

Mintzberg, Henry; Lampel, Joseph; Quinn, James Brian and Ghoshal, Sumantra (2003) The Strategy Process: Concepts, Contexts, Cases 2nd European Edition, Harlow, Pearson Education

Morden, Tony (1993) Business Strategy and Planning: Text and Cases, London, McGraw Hill

Murugavarothayan (a.k.a Morgan), K.K; John Rowan & Partners and Coffee. Michael (2000) Performance Indicators of Professional Services Used by Clients, Proceedings of

the 2000 RICS COBRA Conference held at University of Greenwich 30th August-1st September

Ng, S.Thomas et. Al (2009) Coping with structural change in construction: experiences gained from advanced economies, Construction Management and Economics 27(2), pp. 165-180

Nkado, Raymond and Meyer, Terry (2001) Competencies of professional quantity surveyors: a South African perspective, Construction Management and Economics 19(5) pp. 481-491

Naoum, Shamil (2001), People and Organizational Management in Construction. London, Thomas Telford.

Ofori, George (1994) Formulating a long-term strategy for developing the construction industry of Singapore, Construction Management and Economics 12(3), pp. 219-231

O'Regan, Nicholas and Ghobadian, Abby (2002) Effective Strategic Planning in Small and Medium Sized Firms, Management Decision 40(7), pp.663-671

O'Regan, Nicholas and Ghobadian, Abby (2007) Formal Strategic Planning; Annual Raindance or Wheel of Success?, Strategic Change, 16(1) pp. 11-22

Papke-Shields, Karen E; Malhotra, Manoj K. and Grover, Varun (2006) Evolution in the strategic manufacturing planning process of organisations, Journal of Operations Management 24(5); pp.421-439

Pearce, John A. II; Robbins, D. Keith and Robinson, Richard B. Jr. (1987) The Impact of Grand Strategy and Planning Formality on Financial Performance, Strategic Management Journal 8(2); pp.125-134

Pearce, John A II and Robinson, Richard B. Jr. (2000) Formulation, Implementation and Control of Competitive Strategy, 7th Ed., Boston, Irwin McGraw-Hill

Peel, Michael J. and Bridge, John (1998) How Planning and Capital Budgeting Improve SME Performance, Long Range Planning 21(6), pp. 848-856

Peters, Tom and Waterman, Robert H. (2004) Jr. In Search of Excellence: Lessons from America's Best-Run Companies, 2nd Edition. London, England: Profile Books Ltd.

Phua, Florence T.T. (2006) Predicting construction firm performance: an empirical assessment of the differential impact between industry and firm-specific factors, Construction Management and Economics 24(3) pp. 309 - 320

Phua, Florence T.T (2007) Does senior executives' perception of environmental uncertainty affect the strategic functions of construction firms?, International Journal of Project Management 25(8); pp.753-761

Pickerell, Simone (2000) Modelling Decision-Making in Building Surveying, Proceedings of the RICS COBRA Conference 2000 hed at University of Greenwich 30th August – 1st September

Plano Clarke, Vicki L. and Creswel, John W (2008) The Mixed Methods Reader, L.A. Sage Publications Inc.

Porter, Michael E. (1979) How Competitive Forces Shape Strategy, Harvard Business Review, 57(2); pp. 137-145

Porter, Michael E. (1996) What is Strategy Harvard Business Review, 74(6); pp.61-78

Porter, Michael E (1998) The Competitive Advantage of Nations: with a new introduction, Basingstoke: Macmillan

Porter, Michael E (2004a) Competitive Strategy: Techniques for Analysing Industries and Competitors, Export Edition: New York, Free Press.

Porter, Michael E. (2004b) Competitive Advantage, Export Edition, New York: Free Press; London: Collier Macmillan

Potts, Keith (2004) Quantity Surveying Tools and Techniques – A Review of Client and Contractor Requirements, Proceedings of the 2004 RICS Cobra Conference held on 7-8 September at Leeds Metropolitan University

Prahalad, C.K and Hamel, Gary (1990) The Core Competence of the Corporation, Harvard Business Review, 68(3); pp.79 -91

Prahalad, C.K. and Hamel, Gary (1994) Strategy as a Field of Study: Why Search for a New Paradigm?, Strategic Management Journal 15(S2); pp.5-16

Prahalad, C.K and Hamel, Gary (1996) Competing for the Future, Boston, Mass.; Harvard Business School Press.

Price, A.D.F; Ganiev, B.V and Newson, E (2003) Changing strategic management practice within the UK construction industry, Strategic Change 12(Nov) pp.347-366

Prince, M.W (1988), Strategic Planning in Construction Firms, Ph.D., Bradford, 38-2993

Rajagopalan, Nandini; Rasheed, Abdul M.A. and Datta, Deepak K. (1993) Strategic Decision Processes: Critical Review and Future Directions, Journal of Management 19(2); pp. 349-384

Rhyne, Lawrence C. (1986), The Relationship of Strategic Planning to Financial Performance, Strategic Management Journal, 7(5) pp.423-436

Robinson, Richard B. and Pearce, John A. II (1983) The Impact of Formalised Strategic Planning on Financial Performance in Small Organisations, Strategic Management Journal 4(3); pp.197-207 Robinson, Richard B. Jnr. and Pearce, John A. II (1984) Research Thrusts in Small Firm Strategic Planning, Academy of Management Review 9(1), pp. 128-137

Robinson, Richard B. Jnr. And Pearce, John A. II (1988) Planned Patterns of Strategic Behaviour and Their Relationship to Business-Unit Performance, Strategic Management Journal 9(1); pp.43-60

Robson, Colin (2002) Real World Research, 2nd Edition Oxford, Blackwell Publishers Ltd.

Rudd, J.M (2005), A multivariate examination of the relationship between strategic planning and organisational performance, Ph.D., Aston, 54-8175

Rugman, Alan M. and Verbeke, Alain (1998) Corporate Strategies and Environmental Regulations: An Organising Framework, Strategic Management Journal 19(4) pp.363-375

Scott, Prof. Alex (2007) Strategic Planning, EBS Course Text, Pearson Education

Scott, Brian W (1965) Long-Range Planning in American Industry, American Management Association Incorporated

Seeley, Ivor H. (1997) Quantity Surveying Practice, 2nd edition, England, Palgrave Macmillan

Segars, Albert H. and Grover, Varun (1999) Profiles of Strategic Information Systems Planning, Information Systems Research 10(3); pp.199-232

Shenhar, Aaron J.; Dvir, Dov; Levy, Ofer and Maltz, Alan C.; (2001) Project Success: A Multidimensional Strategic Concept, Long Range Planning 34(6) pp 699-725. Short, Jeremy, C; Ketchen, David J. Jnr; Palmer, Timothy B. and Hult, G. Tomas M. (2007) Firm, Strategic Group, and Industry Influences on Performance, Strategic Management Journal 28(2), pp. 147-167

Shrader, Charles, B; Taylor, Lew and Dalton, Dan R. (1984) Strategic Planning and Organisational Performance: A Critical Appraisal, Journal of Management 10(2); pp.149-171

Shrivastava, Paul and Grant, John H (1985) Empirically Derived Models of Strategic Decision-making Processes, Strategic Management Journal 6(2); pp.97-113

Shuman, Jeffrey C.; Shaw, John J and Sussman, Gerald (1985) Strategic Planning in Smaller Rapid Growth Companies, Long Range Planning 18(6), pp.48-53

Silverman, David (2010) Doing Qualitative Research, 3rd edition, London, Sage Publications

Silverman, David (2006) Interpreting Qualitative Data: Methods for Analyzing Talk, Text and Interaction, 3rd Edition, London, Sage Publications

Society of Chartered Surveyors "Employment Survey", March 2009

Stalk, Jr., George (2006) Curveball Strategies to Fool the Competition, Harvard Business Review 84(9) pp.114-122

Steiner, G.A (1979) Strategic Planning: What Every Manager Must Know, New York: Free Press

Stonehouse, George and Pemberton, Jonathan (2002) Strategic Planning in SME's – some empirical findings, Management Decision 40(9); pp.853-861

Stubbart, Charles (1985) Why We Need a Revolution in Strategic Planning, Long Range Planning 18(6), pp. 68-76 Teece, David J.; Pisano, Gary and Shuen, Amy (1997) Dynamic Capabilities and Strategic Management, Strategic Management Journal, 18(7), pp. 509-533

Teece, David J. (2000) Strategies for Managing Knowledge Assets: The Role of Firm Structure and Industrial Context, Long Range Planning, 33(1), pp.35-54

Teece, David J. (2010) Business Models, Business Strategy and Innovation, Long Range Planning, 43(2/3); pp. 172-194

Thompson Jnr., Aurthur A; Strickland A.J III and Gamble, John E.(2010) Crafting and Executing Strategy: Text and Readings, 17th edition, Maidenhead, McGraw-Hill.

Tiernan, Siobhan B.; Morley, Michael J. and Foley, Edel (2001) Modern Management: Theory and Practice for Irish Students, 2nd Edition, Dublin, Gill and Macmillan Ltd.

Unni, V.K. (1981) The Role of Strategic Planning in Small Businesses, Long Range Planning14(2), pp.54-58

Van Gils, Anita; Voordeckers, Wim and Van Den Heuvel, Jeroen (2004) Environmental Uncertainty and Strategic Behavior in Belgian Family Firms, European Management Journal 22(5); pp.588-595

Venegas, P.C and Alarcon, L. F (1997) Selecting Long-Term Strategies for Construction Firms, Journal of Construction Engineering and Management, 123(4), pp.388-398

Waalewijn, Phillip and Segaar, Peter (1993) Strategic Management: The Key to Profitability in Small Companies, Long Range Planning 26(2) pp 24-30

Warszawski, Abraham, (1996) Strategic Planning in Construction Companies, Journal of Construction Engineering and Management 122(2), pp. 133-140

Welch, Jonathan B (1984) Strategic Planning Could Improve Your Share Price, Long Range Planning 12(2), pp.144-147

White, Roderick E. (1986) Generic Business Strategies, Organizational Context and Performance: An Empirical Investigation, Strategic Management Journal, 7(3); pp.217-231

Wilkinson, S (2003) Strategy Formulation within large Private QS Practices, PhD Thesis, Salford University

Winch, Graham and Schneider, Eric (1993) The strategic management of architectural practice, Construction Management and Economics 11(6), pp. 467-473

Wright, Patrick M.; Dunford, Benjamin B. and Snell, Scott A. (2001) Human resources and the resource based view of the firm, Journal of Management 27(6); pp.701-721

Website References:

Bruce Shaw Partners www.bruceshaw.ie Central Bank of Ireland www.centralbank.ie Central Statistics Office www.cso.ie Davis Langdon PKS www.dlpks.ie Department of Environment, Heritage and Local Government www.environ.ie Department of Finance www.finance.gov.ie DKM Economic Consultants www.dkm.ie Economic and Social Research Institute www.esri.ie Eurostat http://epp.eurostat.ec.europa.eu/portal/page/portal/eurostat/home/ Procurement (www.constructionprocurement.gov.ie) Appendix A Edinburgh Business School Strategic Planning

Process Model



Appendix B Augmented Process Model

VELLAN ON ON CANNAG ALLAN	Analysis And Diagnoses	Analysis And Diagnoses (cont'd)	Choice	Implementation
Objectives	The general environment	Internal factors	Generic strategy alternatives	Resources and structure
Business definition	Macroeconomic analysis.	Value chain	Corporate and business strategy	Divisional, functional, matrix
Mission	unemployment inflation.	Shareholder value analysis	Stability, expansion,	Managerial style
Shareholder wealth	interest rate, exchange rate	Competence	retrenchment	Critical success factors
Gap analysis	Forecasting	Architecture	Combination	Incentives
Means and ends	Competitive advantage of	Experience curve	Cost leadership	
Ethics	nations	Economies of scale	Differentiation	Resource allocation
Profit maximisation	Environmental scanning	Innovation	Fodus	Opportunity cost
Growth vector	PEST	Economies of scope	Segmentation	Marginal analysis
Stakeholder map	Scenarios	Synergy		Optimisation
Gredible, quantifiable, disag-		Joint production	Strategy variations	Budgets
gregated, economic, financial	The Industry and international environment	Opportunity cost Marginal analysis	Diversification: related and unrelated	Critical success factors
Strategists	Demand and supply, price	Ratios	Vertical integration	Evaluation and control
Prindpal agent	determination, elasticity	Gearing	Mergers and acquisitions	Performance measures
Prospector, analyser, defender,	Barriers to entry	Cash flow	Joint ventures and alliances	Ratios
reactor Dist summary	Forms of competition: perfect,	Benchmarking	Priding: leadership, limit,	Degree of Planning and type of
Team composition	Segmentation	Culture: Dower.role.task.	Antenand	Monitorina systems
Group dynamics	Differentiation	personal	Strategy choice Risk analysis	
	Strategic groups	Competitive position	Managerial perceptions	
		Product life cycle	Net present value	
		Market share	Familiarity	
		Portfolio analysis	Scenarics	
		Perceived differentiation	Break even	
		Strategic groups	Payback	
		Competitive reaction	Sensitivity	
		Hist mover	SWOT	
		Five forces	Game theory	
		Elements of competitive		
		advantage		
		ETOPS		
		ourategic advantage prome		1 - 0 - 4
				Feedback
				Communication Management style Adaptability
				Learning organisation

Appendix C Construction Industry Sectors



Appendix D Strategic Planning in New Zealand QS Practices

Summary of John Boon Work.

	2001	2008		
Market orientation Knowledge base	 75% had expanded services All recorded an increased workload from banking sector 75% had clear value-added strategy Knowledge in minds of the staff 75% had annual reviews 75% had annual reviews 75% had ISO accreditation Firms with international connections accesses expertise when required Mentoring juniors (on a need-to-know basis). 	 Practices that dominate the sector are part of an international group As market improved practices returned to core business High level advice at outset of project including procurement methodology Annual review of staff including training needs All have formal quality assurance systems International connections still important for knowledge transfer Only one ISO accredited More formalised mentoring of juniors 		
Flexibility in cost structure	 Difficulty in recruiting staff Using post contract work as a buffer, use of overtime and transfer of work between offices Fewer temporary employees 	Difficulty obtaining sufficient numbers of skilled professionals		

	and subcontractors available			
	• Less of an issue during this			
	period			
Efficiency and price competitiveness	 All believed they are now more efficient Period not as "cut- throat" as original investigation 	 Competition for larger projects based on competence and price 		
	• All reviewing IT systems			
Financial resources	Not a significant issue during	High workload and		
	uns period	reasonable profits		
Strategy	 Adopted a value-added strategy Not able to find means of further lowering cost-base 	 Half of respondents have strengthened the value added position Half have moved to cost leadership with a lower level of service 		

INTERVIEW PROMPT SHEET

SUMMER 2009

PRIVATE AND CONFIDENTIAL

Company N	ame:					
Interviewee	:					
Position in (Organisation:					
Date of Inte	rview:					
1. GEN	ERAL COMPA	ANY INFORMATION				
• In what y	In what year was your company established?					
• How man	ny people are emp QS	ployed in the company? Admin. Other				
Peak						
Curre	ently					
 How is the organisation structured? 						
• What are the key sectors serviced by your organisation?						
~ Has t	~ Has this changed significantly over the last number of years?					
• What are Project M	• What are the main services which your organisation provides? (Quantity Surveying; Project Management; Health and Safety)					

~ Has this changed significantly over the last number of years?

~ What have been the main drivers of change?

• Describe your organisations' experience over the last 5 years. What has been happening;' how has it affected the firm; what are the ways in which the firm has had to adapt; has it caused firm to change strategy?

2. ORGANISATIONAL OBJECTIVES AND STRATEGIST

- Has the company got a clear set of organisational objectives? Does the company have a strategic plan?
 - ~ Are they written down/formally recorded? (Formality)
- In the last 5/10 years, what has been the company's overall objective? (Growth/Stability/ Retrenchment)

Peak:

Currently:

• How has is sought to achieve objectives? (cost leadership, differentiation, focus)

Peak:

Currently:

- How do you go about reducing cost and improving service to enhance value for the customer? *(Efficiencies)*
- What is the key driver of the planning process? (*profitability, resources allocation, communication, image enhancement*?)
- Where do strategic initiatives come from? (*staff, competitors, clients*?) Flow
- Are objectives primarily short term, long term or on a project basis?
 - ~ Is project planning prioritised over strategic? (strategic versus tactical)
- Are staff involved in the process of objective setting? (participation)
- How are objectives disseminated throughout the organisation (integration)

- ~ Are they incorporated into performance measures?
- ~ Are they apparent within individual project planning?
- ~ Dot they influence resources allocation?
- What are the main difficulties in setting the organisational objectives?
- What are the main hindrances to strategic planning in general?
- Do you consider alternative courses of action?

3. ENVIRONMENTAL ANALYSIS

Macro Environment

- What is the most significant aspect of the macro economy affecting your organisation?
 - ~ Uncertainty: how do you deal with it? Flexibility built in? Contingency?
 - ~ **Information gathering**; regularity with which it is sought? Does somebody have responsibility for it? How and where does information come from?
 - ~ Formal **process** for gathering information? What are the key areas for which data is gathered regularly?
- Have government decision made a considerable impact? (new forma of contract; NDP; Capital spending)
 - ~ What has been the most significant change directly affecting your organisation?

Industry Environment

- How do you envisage the industry changing in the short/medium term?
 - ~ What are the key drivers of change?
 - ~ How has the organisation had to change to meet the changing market conditions?

- ~ What is your expectation of levels of market activity over the next two years?
- How do you envisage the profession changing over the short/medium term?
 - Fee levels: how do you foresee fees changing? Will it be a SR or a LR issue? How will it affect the way you do business?
 - ~ Perceived changes in the ROLE of the QS
 - ~ Related professions
 - SCS/RICS: do you get any benefit from being a member of the SCS? Are there any areas which you feel could be improved upon?

Internal Organisation

- Do you periodically analyse your own business (in terms of identifying key strengths and weakness, profitability, productivity, marketing, quality of service?
- **Repeat business** is an important aspect of a professional service practice, can you briefly describe:
 - ~ Partnering arrangements
 - ~ Communication channels
 - ~ Problems and how they were resolved
 - ~ Do you solicit feedback on services provided?
- **Quality**: in terms of the quality of service provided, it is usually measure in terms of what the clients receives, or how it is delivered?
- Staff:
 - Training and development. Can you outlined the provision currently in place for staff development
 - Involvement in the planning process
- IT
- ~ Does the company utilise IT packages in the delivery of projects
- Finance:
 - Has the company built in financial contingency given the current economic climate?
 - ~ How significant has tender place deflation impacted on your organisation?

Competitive Environment

- How would you describe the competitive environment within which you operate?
 - ~ Peak
 - ~ Currently
- Is there a system/process for monitoring changes in the economic and competitive environment? (formality and comprehensiveness)
- Does the organisation seek information on competitors? How does the organisation obtain such information? (formality and comprehensiveness)
- How do you think your competitors have reacted to the (similar) circumstances facing the industry?
- Is there anything in particular that you believe sets you apart from competitors?

4. STRATEGIC CHOICE

- Looking ahead, what is the company's main objective going forward?
- How is this most likely to be achieved? (cost leadership, differentiation, new markets, alliances/partnerships)
- What do you believe to be the key challenges in achieving the objective? How are you going to address the challenge?
- Are there contingencies built in?

5. IMPLEMENTING STRATEGY

- Given the current situation within the industry are there any issues in relation to staff?
- Are you considering mechanisms to retain and transfer existing knowledge?
- To what extent will training/retraining be necessary?

- Overall, what areas could present opportunity for the profession?
- Overall, what are the key threats/hindrances to the profession/industry?

Appendix F Article for the SCS Magazine

Strategic Planning in Irish Quantity Surveying Practices By Róisín Murphy BA(hons); PDES; MBS; MSc.

Strategic planning is an important function of business which has been analysed in a variety of industry settings. There remains a notable lack of investigation of the construction industry particularly regarding QS practices.

Numerous reports have been published which rightly highlight the importance of construction to the economy as a whole, but comparatively few investigate how, at a micro-level, these firms are managed. What size are practices? What strategies are being pursued? What are the objectives for the future? How are they seeking to achieve them? In order to address these questions a survey was conducted of QS practices at chief executive, managing director or managing partner level.

Profile of QS Practices

Prior to investigating the strategic planning in QS practices, it is useful to establish a context of the professions structure. The following provides a snapshot of the QS profession in Ireland:

- 85% of practices are independently owned. The remainder are part of an international organisation
- 79% in business more than 5 years; with 66% in business over 10 years
- 39% are ISO certified
- 66% have five or fewer employees, with only 1 respondent having greater than 100
 - ~ These figures were 41% and 2 respectively at peak (2007)
- Services currently being provided:
 - 97% offer traditional QS services, 76% offer project management; 55% bank monitoring; 31% construction management; 23% Sustainability-related services
 - ~ 71% confirm that the range of services provided has not changed in last five years
- Evidence of concentration on private non-residential and social infrastructure sectors at present

Characteristics of Strategic Planning in QS Practices

The process of strategic planning may be formal or informal, planned or emergent, and involves many or few members of the organisation in its development. Characteristics of strategic planning in Irish QS practices include:

- 30% of practices have a written plan. These practices tend to be larger practices, many of which have links to a global construction cost consultancy firm.
- Regardless of the formality of the process or the existence of a written strategic plan, it is evident that principals of QS practices consider similar issues as those undertaking a formal approach to the process.
- Critical factors for the success of QS practices include quality staff, repeat business and maintaining relationships with other industry professionals.
- Industry and economic analysis are prevalent however a notable lack of analysis of competitors is undertaken.
- In general, the strategic planning process tends to be unstructured and undocumented. Outcomes of the process are communicated informally amongst staff.
- Participation is limited to senior management.

QS practices are thinking and acting strategically however not necessarily in a systematic way. It is inconceivable to think of undertaking a construction project without a plan yet entire practices are being run (and often quite successfully) without one! Crucially however, over 40% of respondents noted that strategic planning processes are beginning to emerge in their practice, with only 24% believing it to be unnecessary (of these 24%, 85% have fewer than 5 employees).

What strategies are QS practices pursuing?

The turbulent environment within which QS practices are operating has presented a number of challenges hence survival is the overriding concern. It is interesting to note however, that many strategies currently being pursued have not changed between the peak and trough of the construction cycle.

The following summary is indicative of the responses received:

- 50% of respondents are seeking stability in a rapidly changing environment
- 27% are rationalising or downsizing their operations (this figure may be low due to previous rationalising having taken place)
- The most common mechanisms by which strategy is being pursued (which for 75% of respondents have not changed in the last five years) include:
 - ~ to provide the best value compared to rivals (59%)

- ~ to differentiate their service from competitors (39%)
- to compete based on providing lower overall cost than rivals (15%) an interesting finding in light of current trends in tender price inflation
- Approximately 33% QS practices are seeking opportunities abroad. The two most common places cited as being the UK and Middle East.

A diverse range of strategies are being implemented as a means to survive the current downturn. Patterns in the strategic planning process result in the existence of groups within the profession. Three distinct groupings are evident, centred on size and ownership structure which influences the type, scope, formality of the process and to a lesser extent the current strategy being pursued.

Group 1

Practices forming part of a larger international group tend to have a formal, structured process for strategy development, involving participants from several levels within the organisation. The strategic planning process closely mirrors that of head office which is often the driving force behind the whole process. These practices, although currently seeking stability, expanded considerably during the construction boom and have the capacity to undertake large projects and offer a diverse range of services.

Group 2

The next group are also reasonably large (greater than 20 employees) indigenous firms. Although comprehensive macroeconomic and industry analysis is undertaken by this group, the strategic planning process is often informal. Only half of the practices have a written strategic plan, and where one exists, the extent of participation of staff is limited. Similarly to the first group, these practices expanded rapidly through the construction boom and many are now seeking stability in the domestic market while exploring opportunities abroad.

Group 3

The final group consists of small to medium sized QS practices, ranging from sole traders to twenty employees, and represent over fifty percent of survey respondents. The majority of this group have no formal strategic plan or planning system in place, however three quarters confirm that strategic planning processes are beginning to emerge. QS practices within this group did not expand rapidly during the construction boom and consequently have faced a proportionately smaller reduction in staff numbers in the downturn. Corporate strategy has not changed significantly over the last number

of years and differentiation of service provision is reliant on the provision of a personal service to clients.

Where to from here?

The organisational benefits of strategic planning are well documented. The recognition of the need for strategic planning in the profession is now becoming increasingly apparent. The multi-dimensional nature of strategic planning results in a complex process which is exacerbated in the context of a turbulent environment. The process does not however need to be cumbersome or rigid, nor does it require the recruitment of external consultants. Strategic planning must be systematic, participative, ongoing and realistic yet remain flexible to allow for changes within the industry and competitive environment. Participation from staff is a crucial component for idea generation as well as instilling a sense of ownership of the strategy. This is particularly important for professional service practices whereby staff interaction with clients fosters the reputation upon which repeat business depends.

The education, training and knowledge of the QS in project and risk management must be transferred from the tactical to the strategic in order to better position QS practices for the recovery. Appendix G Pilot Questionnaire Feedback Form

Pilot Study	-	Strategic	Planning	in	0S	Practices
a contraction of the states and					-	

4. Feedback on the Questionnaire

Thank you very much for taking the time to participate in the pilot study of my research.

Your feedback on the survey will assist me in making improvements prior to undertaking the main study. To that end, the next number of questions seek feedback on various aspects of the questionnaire and invite you to provide your own comments.

Once again, your assistance in this regard is much appreciated.

Should you have any queries or require further information, please do not hesitate to contact me at roisin.murphy@dit.ie

Kind regards

Roisin

35. How long did it take you to complete the questionnaire?

36. What is your opinion of the questionnaire length?

O Too long

- O Too short
- O About right

Comments

37. What is your opinion of the clarity of questions?

- O Poor
- Satisfactory
- Very Good
- C Excellent

Comments
ilot Study -	Strategic Planning in QS Practices
38. What is yo	our opinion of the format and structure of the questionnaire?
O Poor	
🔵 Satisfacto	ry .
O Very Good	
O Excellent	
Comment	
-	
39. Does the s invasive?	survey contain any questions that you believe to be inappropriate or too
O Yes	
O No	
If yes please s	specify
40. Dunn 11m	annu mit a los issue en ellos de tanis d'anim in Os atits Suemaine
practices?	anvey official key issue regarding scracegic planning in Quanticy Surveying
() Yes	
O No	
If yes please s	specify
41 Didawa ta	and and difficulty in completing the superficult in 2
() Yes	we any unitency in completing the questionnane?
O No.	
Unio	
Il yes please s	pecny
_	

Appendix H Questionnaire

pronses to the questionnaire will be treated with strict confidentiality. hysis will be undertaken based on aggregates and neither individual non- npany names will be specifically identified at any stage. c. Please solect the position you currently hold within your organisation. Managing Director/Partner Director/Partner Other (please specify) c. Please indicate the number of years your organisation has been in business. 0-5 years 5-10 years 21-50 years 21-50 years 21-50 years 21 he ownership structure that best describes your organisation is Independently owned Wholly evened subsidiary Other (please specify) c. Dees your organisation currently hold 1SO Certification? Yes Nu	Gen	eral Company Information
	spo alys npa	nses to the questionnaire will be treated with strict confidentiality. is will be undertaken based on aggregates and neither individual no ny names will be specifically identified at any stage.
Managing Director/Partner Director/Partner Associate Director Other (please specify) P. Please indicate the number of years your organisation has been in business. 0-5 years 5-10 years 5-10 years 21-50 years 21-50 years 21-50 years 11 20 years 21-50 yea	1.0	ease select the position you currently hold within your eraphication
 Director/Partner Associate Director Other (please specify) Please indicate the number of years your organisation has been in business. 0-5 years 5-10 years 21-50 years 21-50 years 21-50 years 350 years 8. The ownership structure that best describes your organisation is Independently owned Wholly owned subsidiary Other (please specify) 4. Does your organisation currently hold 150 Certification? Yes No 	0	Managing Director/Partner
 Associate Director Other (please specify) Please indicate the number of years your organisation has been in business. 0-5 years 6-10 years 11 20 years 21-50 years >50 years Independently owned Wholly owned subsidiary Other (please specify) Absorver organisation currently hold 1SO Certification? Yes No 	õ	Director/Partner
 Other (please specify) Please indicate the number of years your organisation has been in business. 0-5 years 6-10 years 21-50 years 21-50 years >50 years Independently owned Wholly owned subsidiary Other (please specify) Loes your organisation currently hold 1SO Certification? Yes Nu 	ň	Associate Director
	0	Other folgase cherily)
 Please indicate the number of years your organisation has been in business. 0-5 years 6-10 years 21-50 years >50 years 21 the ownership structure that best describes your organisation is Independently owned Wholly owned subsidiary Other (please specify) 4. Does your organisation currently hold 150 Certification? Yes No 	0	ormer (prease specify)
 0-5 years 5-10 years 21-50 years >50 years 3. The ownership structure that best describes your organisation is Independently owned Wholly owned subsidiary Other (please specify) 4. Does your organisation currently hold ISO Certification? Yes No 	2. P	ease indicate the number of years your organisation has been in business.
 6-10 years 11 20 years 21-50 years 350 years 350 years 36 the ownership structure that best describes your organisation is, Independently owned Wholly owned subsidiary Other (please specify) 4. Does your organisation currently hold 1SO Certification? Yes No 	0	0-5 years
 11 20 years 21-50 years >50 years 3. The ownership structure that best describes your organisation is Independently owned Wholly owned subsidiary Other (please specify) t. Does your organisation currently hold ISO Certification? Yes No 	õ	6-10 years
 21-50 years >50 years 3. The ownership structure that best describes your organisation is Independently owned Wholly owned subsidiary Other (please specify) Does your organisation currently hold 1SO Certification? Yes No 	õ	11 20 years
 >50 years S. The ownership structure that best describes your organisation is Independently owned Wholly owned subsidiary Other (please specify) 4. Does your organisation currently hold ISO Certification? Yes No 	õ	21-50 years
 3. The ownership structure that best describes your organisation is A. Independently owned Wholly owned subsidiary Other (please specify) 4. Does your organisation currently hold ISO Certification? Yes No 	õ	>50 years
 Independently owned Wholly owned subsidiary Other (please specify) Loes your organisation currently hold ISO Certification? Yes No 	3 1	e ownership structure that best describes your organisation is
 Wholly owned subsidiary Other (please specify) Loes your organisation currently hold ISO Certification? Yes No 	0	Independently owned
 Other (please specify) 4. Does your organisation currently hold ISO Certification? Yes Nu 	ŏ	Wholly owned subsidiary
Concrete (product objective)	õ	Other (mease specify)
4. Does your organisation currently hold ISO Certification? Yes No	9	ener (pease spear)
4. Does your organisation currently hold ISO Certification7 O Yes No		
O Yes ○ Nu	4. U	oes your organisation currently hold ISO Certification?
Ο Nu	0	Yes
	0	No

Strategic Planning in Irish Quanti	ty Surveying Practices
5. Please indicate the number of people CURF	RENTLY employed in your organisation
0-5	
0 5-10	
O 11-20	
O 20 50	
O 50 99	
○ >100	
6. Please indicate the maximum number of p the PEAK of the construction boom (c. 2007)	cople your organisation employed during
0-5	
0 6-10	
O 11 20	
O 20-50	
O 50-99	
() >100	
7. Please indicate the key services CURRENT	Y provided by your organisation.
(Note: select as many services as appropriat	e - multiple answers will be accented.)
Traditional Quantity Surveying Services	Facilities Management
Construction Management	Bank Monitoring
Project Management	Sustainability
Health and Safety Management	Other
If other services provided, please specify	-
8. Has the range of services provided by you	r organisation changed in the last 5 years?
() No	
O Yes	
If "yes", please specify	

	I None at	2 5mal	3 Moderate	4 Large
Residential (private or public)		proportion	proportion	proportio
Private non-residential (office, retail etc.)	ŏ	ŏ	ŏ	ŏ
Productive infrastructure (civils, roads etc.)	ŏ	ŏ	ŏ	ŏ
Social infrastructure (healthcare, education etc.)	ŏ	ŏ	ŏ	ŏ
10. Has the proportion of work undertaken will last 5 years?	Urin each s	sector servi	ced changed	l in the
\bigcap No (if selecting this option please then click t skip the following question)	he "next" t	outton at the	end of the p	page to
O Yes				
11. Please identify the proportion of your over sectors during the PEAK of the construction bo	all worklo. oom (c.200	ad undertal 06).	en within e	ach of th
	1 None at	2 5mail	3 Moderate	4 Large
Description () () () () () () () () () (proportion	proportion	proportio
Private pop-residential (office, retail etc.)	X	X	X	X
Dracha tive intractructure (rivile manualtri)	X	X	X	X
Social infrastructure (healthcare, education etc.)	X	X	X	ŏ
Strategic Management Practices				
e following section seeks information reg actices within your organisation. 12. Organisational Strategy	arding st	trategic m	anagemen	t
Please select the statement that best describe	es vour on	anisation's	s strategic i	Itention
The organisation is primarily concerned with t and the provision of new services and innovations	he identific	ation of new	market oppo	ortunities
The organisation seeks to defend market shar new developments in others following extensive n	e in some s narket anal	sectors while ysis	pursting pro	mising
Maintaining the current market position of the developing new market opportunities are less important.	e organsatio ortant.	on is of critic	a importanc	e while
Decisions are reached based on circumstance	s as they a	inse,		

13. Approach to strategy formation.					
Please indicate the approach to strategy for organisation.	rmation w	hich is do	ominant wi	ithin you	r
The bosiness environment is analysed and plan is worked out by senior management which	objectives this then p	are deter assed dov	mined, tollo vir for împle	wing whi ementatio	chia ni.
Strategy is not planned before the event b have little structure.	ut emerge	s over tim	ie, therefor	e may ap	pear to
Competitive advantage rests within our co- drives the strategy formation process.	mpany's in	ternal res	ources/capa	abilities w	hich
14. Risk attitude.					
Please indicate the extent of your agreeme	nt with ea	ch state	ment.		
	1 Strongly disagree	2 Disagree	3 Neither agree nor disagree	4 Agree	5 Strongl agree
The organisation has a stong tendancy for low risk projects		E		Ε	
Top managers in the organisation believe it is best to explore opportunities gradually via incremental behaviour.					
The organisation typically adopts a cautious "wait-and-see" posture.					
15. Please indicate the degree of importanc compete successfully.	e of each	issue bel	ow for you	ır organi	sation t
Un	1 : Importantin	2 OI little noortance	3 Moderately	4 Importan	5 Very timporta
Analysis of competitors	0	0	O	0	0
Construction industry analysis	Q	Q	õ	Q	Q
Macroeconomic analysis	8	Q.	Q	Q	Q
Internal organisation analysis	8	X	S	S	2
Repeat business	8	X	8	X	X
Relations with other industry professionals	X	X	S	20	X
Long range planning	X	X	X	S	8
starr graining and development	X	X	X	X	8
Research and development	X	X	X	X	X
	Q	0	Ų	0	0
other (please specify)					

16. Corporate	Strategy
Please select I	the strategy that best describes your CURRENT corporate
strategy/busi	ness objective.
Maintain c	iment market position in the face of capidly changing competitive environme
Expansion	into new growth sectors/markets
Rationalisa	tion/downsizing of current operations in pursuit of greater efficiency
Combinatio	on of any two above
If combination,	, please specify the combination
17. Has the ov	verriding corporate strategy changed in the last 5 years?
O NO	
() Yes	
If "yes" please	specify
18. Business I	evel Strategy
18. Business I Please state ti	evel Strategy ne primary mechanism by which your CURRENT strategy is most likely
18. Business I Please state ti achieved.	evel Strategy ne primary mechanism by which your CURRENT strategy is most likely
18. Business I Please state fl achieved.	evel Strategy he primary mechanism by which your CURRENT strategy is most likely chieve lower overall cost/price than rivals.
18. Business I Please state fl achieved. Strive to a Seek to dif	evel Strategy to primary mechanism by which your CURRENT strategy is most likely chieve lower overall cost/price than rivals. ferentiate our service from rivals to appeal to a broad spectrum of clients
18. Business I Please state th achieved. Strive to a Seek to dif	evel Strategy be primary mechanism by which your CURRENT strategy is most likely chieve lower overall cost/price than rivals. ferentiate our service from rivals to appeal to a broad spectrum of clients a best value compared to rivals offering services with similar attributes
18. Business I Please state fl achieved. Strive to a Seek to dif Provide the Concentrat cost/price	evel Strategy ne primary mechanism by which your CURRENT strategy is most likely chieve lower overall cost/price than rivals. ferentiate our service from rivals to appeal to a broad spectrum of clients to best value compared to rivals offering services with similar attributes te on narrow market segment and compete with rivals on basis of lower
18. Business I Please state fl achieved. Strive to a Seek to dif Provide the Concentrat cost/price	evel Strategy ne primary mechanism by which your CURRENT strategy is most likely chieve lower overall cost/price than rivals. ferentiate our service from rivals to appeal to a broad spectrum of clients to best value compared to rivals offering services with similar attributes are on narrow market segment and compete with rivals on basis of lower te on narrow market segment and compete on basis of customised attributes
18. Business I Please state fl achieved. Strive to a Seek to dif Previde the Concentrat cost/price Concentrat	evel Strategy the primary mechanism by which your CURRENT strategy is most likely chieve lower overall cost/price than rivals. ferentiate our service from rivals to appeal to a broad spectrum of clients to best value compared to rivals offering services with similar attributes are on narrow market segment and compete with rivals on basis of lower the on narrow market segment and compete on basis of customised attribute terriding business level strategy changed in the last 5 years?
18. Business I Please state fl achieved. Strive to a Seek to dif Previde the Concentrat cost/price Concentrat 19. Has the ov	evel Strategy be primary mechanism by which your CURRENT strategy is most likely chieve lower overall cost/price than rivals. ferentiate our service from rivals to appeal to a broad spectrum of clients a best value compared to rivals offering services with similar attributes to on narrow market segment and compete with rivals on basis of lower are on narrow market segment and compete on basis of customised attributes the on narrow market segment and compete on basis of customised attributes the on narrow market segment and compete on basis of customised attributes the on narrow market segment and compete on basis of customised attributes
18. Business I Please state fl achieved. Strive to a Seek to dif Previde the Concentrat cost/price Concentrat 19. Has the ov No Yes	evel Strategy the primary mechanism by which your CURRENT strategy is most likely chieve lower overall cost/price than rivals. Ferentiate our service from rivals to appeal to a broad spectrum of clients a best value compared to rivals offering services with similar attributes to on narrow market segment and compete with rivals on basis of lower to on narrow market segment and compete on basis of customised attribute terriding business level strategy changed in the last 5 years?
18. Business I Please state fl achieved. Strive to a Seek to dif Previde the Concentrat cost/orice Concentrat 19. Has the ov No Yes 1. "yes" observed	evel Strategy the primary mechanism by which your CURRENT strategy is most likely chieve lower overall cost/price than rivals. ferentiate our service from rivals to appeal to a broad spectrum of clients to best value compared to rivals offering services with similar attributes to on narrow market segment and compete with rivals on basis of lower to on narrow market segment and compete on basis of customised attribute perriding business level strategy changed in the last 5 years?
18. Business I Please state fl achieved. Strive to a Seek to dif Previde the Concentrat cost/price Concentrat 19. Has the ov No Yes It "yes" please	evel Strategy be primary mechanism by which your CURRENT strategy is most likely chieve lower overall cost/price than rivals. ferentiate our service from rivals to appeal to a broad spectrum of clients a best value compared to rivals offering services with similar attributes to on narrow market segment and compete with rivals on basis of lower be on narrow market segment and compete on basis of customised attribute perriding business level strategy changed in the last 5 years?

	egic Planning in Irish Quantity Surveying Practices
20.	Please indicate if any of the variations below are currently being
	Sidered/pursued.
H	Partnership with the practiceter
	Acquision of a co-precise
	merger with QS processo
	Operate in a foreign marker (prease indicate general location below)
	Uther (please specify below)
	None of the appyc
Fair	ther details may be provided in the textbox below
21.	Please indicate how your organisations strategy is communicated to firm members.
	Informal discussion/word of mouth, in a broad, general fashion
	Formal moetings, where firm strategy is communicated verbally
	Through the distribution of documents explicitly containing firm strategy
	Other (please specify)
>>.	To what extent have personal lifestyle choices influenced your company's strategic
dec	isions?
0	No influence at all
0	Little influence
0	Same influence
0	Significant influence
~	

Strategic Planning in Irish Quantity Surveying Practices

23. Please indicate your extent of agreement/disagreement with each statement below.

In our company....

	Strongly disagree	2 Disagree	3 Neutral	4 Agree	Strongly
Strategic/business planning process is very structured	Ô	0	0	0	0
Written guidelines exist to structure strategic/business planning	0	C	0	0	0
Outputs of strategic/business planning are formally documented	0	0	0	0	0
We are exhaustive in gathering relevant information	0	0	0	0	0
We determine optimal courses of action from alternatives	0	0	0	0	0
We delay decisions until all alternatives have been evaluated	0	0	0	0	0
The planning flow is "top down" Our process of strategic/business planning involves numerous participants at all levels of the organisation	00	00	00	00	00
Participation of (external) specialists in strategic planning is high	0	0	0	0	0
Line managers and staff are involved in the strategic/business planning process	0	0	0	0	0
We constantly evaluate and review strategic plans	0	0	0	0	0
We frequently adjust strategic plans to adapt them to changing conditions	0	0	0	0	0
Strategic planning is a continuous process	0	C	0	0	0
24. Which one of the following best character planning?	erises you	r linns ex	operience	withstr	ategic
 The organisation has no experience of stra- necessary. 	tegic plann	ing at pre	sent, and	it is not d	leemed
O Procedures and policies for conducting plan	ning are ju	st beginni	ng to eme	rge.	
O Planning activities have been formally developed	bire bagol	lested.			

Long history of strategic planning activities.

25. Does your organisation currently have a formal (written) strategic plan?

No (if selecting this option, please then click the "next" button at the bottom of the page to skip the next number of questions)

O Yes

Mission/vision statement Corporat Organisation objectives Organisation Qrganisation structure Marketin Sales targets Productor Profit targets Quality 2 Industry analysis (construction) Staff application Industry analysis (construction) Staff application Competitor analysis Competitor analysis Additional Items (please specify) Staff application for your amount of time into the future the plan covers) 1-3 years 1-3 years	e appraisal (e.g. SWO1 and tional core competencies g strategy in/operation process ssurance process raisal ning and development
Organisation objectives Organisation Organisation structure Marketin Sales targets Production Profit targets Quality 3 Industry analysis (construction) Staff application Industry analysis (QS profession) Staff application Competitor analysis Staff up Additional Items (please specify) Staff up Production Staff up Industry analysis Staff up Industry analysis Staff up Operation analysis Staff up Operation analysis Staff up Operation analysis Staff up Industry analysis Staff up Operation analysis Staff up Operation analysis Staff up Additional Items (please specify) Staff up Production of time into the future the plan covers) Staff up 1-3 years Staff up	tional core competencies g strategy on/operation process ssurance process raisa) ning and development
Organisation structure Marketin Sales targets Products Profit targets Quality 3 Industry analysis (construction) Staff apr Industry analysis (QS profession) Staff tra Competitor analysis Additional Items (please specify) 7. Please indicate the usual planning horizon/duration for your mount of time into the future the plan covers) 1-3 years	g strategy n/operation process ssurance process raisal ning and development
Sales targets Products Profit targets Quality / Industry analysis (construction) Staff application Industry analysis (QS profession) Staff training Competitor analysis Additional Items (please specify) 7. Please indicate the usual planning horizon/duration for your mount of time into the future the plan covers) 1-3 years	n/operation process ssurance process raisal ning and development
Profit targets Quality 3 Quality 3 Industry analysis (construction) Staff ap Industry analysis (QS profession) Staff tra Competitor analysis Additional Items (please specify)	ssurance process raisal ning and development
 Industry analysis (construction) Staff approximation industry analysis (QS profession) Competitor analysis Additional Items (please specify) Additional Items (please specify) Professe indicate the usual planning horizon/duration for your mount of time into the future the plan covers) 1-3 years 	raisal ning and development
 Industry analysis (QS profession) Competitor analysis Additional Items (please specify) Additional Items (please specify) Provide the usual planning horizon/duration for your mount of time into the future the plan covers) 1-3 years 	ning and development
Competitor analysis Additional Items (please specify) 7. Please indicate the usual planning horizon/duration for your mount of time into the future the plan covers) () 1-3 years	
Additional Items (please specify)	
27. Please indicate the usual planning horizon/duration for your amount of time into the future the plan covers)	
 ○ 8-15 years ○ >15 years ○ Other 	
If you selected "other" please specify	

1 Never 2 Racely J 4 5 Ve 1 Never 2 Racely Occasionally requently/reque Time three maints 0 0 0 Market uncertainty 0 0 0 0 I ack of capertise 0 0 0 0 0 Builget constraints 0 0 0 0 0 0 Resistance to change (by staff) 0	Please indicate the frequency with which organisation.	h cardi has	peren ext	erienced)	by your	
Time torestraints O O O Market uncertainty O O O O Tack of expertise O O O O O Budget constraints O O O O O O O Budget constraints O </th <th>organisation.</th> <th>1 Nanar</th> <th>2. Duralu</th> <th>j</th> <th>4</th> <th>5 Ve</th>	organisation.	1 Nanar	2. Duralu	j	4	5 Ve
Innections 0 0 0 0 0 0 Market uncertainty 0 <t< th=""><th>-</th><th>1 Never</th><th>2 50.00</th><th>Decasiona</th><th>y requestly</th><th>freque</th></t<>	-	1 Never	2 50.00	Decasiona	y requestly	freque
Market incertainty 0	Turne three counts	8	X	X	8	2
Budget constraints 0	Market incertainty	8	X	X	S	2
Resistance to mange (hy staff) 0 <	Pudeat constraints	X	X	X	X	Z
Resistance to change (by management)	Bunger constraints Resistance to change (by staff)	X	X	X	X	č
Other obstacles (please specify) 29. Please select the use(s) of your organisation's strategic plan. (Note: select as many as are appropriate, as multiple answers will be accepted) To culce company operations To measure company performance To establish incentives for managers To establish incentives for managers To improve allocation or resources To establish in attracting major customers To gain 1SO certification Other uses (please specify)	Resistance to change (by management)	ŏ	ŏ	ŏ	ŏ	č
 29. Please select the use(s) of your organisation's strategic plan. (Note: select as many as are appropriate, as multiple answers will be accepted) To culce company operations To measure company performance To restablish incentives for managets To improve allocation or resources To obtain funding To assist in attracting major customers To gain ISO certification Other uses (please specify) 	Other obstacles (please specify)	\sim	0	0	0	~
29. Please select the use(s) of your organisation's strategic plan. (Note: select as many as are appropriate, as multiple answers will be accepted) To pulce company operations To establish incentives for managers To improve allocation of resources To improve allocation of resources To assist in attracting major customers To gain ISO certification Other uses (please specify)	contra universitative shiranyy					
	(Note: select as many as are appropriate To culce company operations To measure company performance To establish incentives for managers To improve allocation of resources	nisation's o _r as multip	strategir de answo	plan. ers will be	accepted))
	 (Note: solect as many as are appropriate To culce company operations To measure company performance To establish incentives for managers To improve allocation or resources To assist in attracting major customers To gain ISO certification Other uses (please specify) 	nisation's c _r as multip	strategir de answo	plan. Yes will be	accepted))
	(Note: select as many as are appropriate To oulde company operations To measure company performance To establish incentives for managers To improve allocation of resources To obtain funding To assist in attracting major customers To gain 150 certification Other uses (please specify)	nisation's e, as multip	strategir de answo	plan. Ses will be	accepted))
	(Note: select as many as are appropriate To oulde company operations To measure company performance To establish incentives for managers To improve allocation of resources To obtain funding To assist in attracting major customers To oain ISO certification Other uses (please specify)	nisation's	strategir de answo	plan. Ses will be	accepted))
	(Note: select as many as are appropriate To oulde company operations To measure company performance To establish incentives for managers To improve allocation or resources To obtain funding To assist in attracting major customers To gain 1SO certification Other uses (please specify)	nisation's e _t as multip	strategir de answo	plan. Ses will be	accepted))

IMPLEMENTATION of your strategic plan?	1 Never	2 Ranely	3	4	5 Ver
Day to day operations distracted attention from implementation	0	0			
Implementation took longer than anticipated	0	0	0	0	0
The stated objectives were unachievable	Õ	Õ	Õ	ŏ	Õ
Employee's capabilities were insufficient	Õ	Õ	Õ	Õ	Õ
Communication was inadequate	\bigcirc	0	0	0	0
Overall goals of strategy were not well enough understood by staff	0	0	0	0	0
External factors impacted on implementation	0	0	0	0	0
Co-ordination of implementation not effective enough	0	0	0	0	0
Unanticipated problems arose	0	0	0	0	0

Appendix I Correspondence with Survey Respondents

A. E-mail seeking participation from Phase 1 Respondents

Message Delivery Schedule Delivery completed on May 5, 2010 11:47. Message Recipients The message mailed to **11** recipient(s). View Message Preview Below is a preview of your message based on the first recipient in your list ([Email]).

To: [Email]

From: roisin.murphy@dit.ie

Subject: Strategic Planning in Irish Quantity Surveying Practices - Phase 2

Body: Dear [FirstName]

Hope you are well.

Further to my previous correspondence, you will be aware that on behalf of DIT and the SCS I am undertaking a survey of every QS practice in Ireland.

Having participated in the first phase of my doctoral research, you have been extremely generous with your time and willingness to provide invaluable insight into the strategic management of your own QS practice. In order to validate my findings from phase one, and to reflect any changes since our meeting, I would once again be grateful for your participation in this final phase of my research.

The questionnaire contains 31 questions and takes approximately 15 minutes to complete. As always, responses will be treated with strict confidentiality and neither individual nor company names will be identified in subsequent analysis. Results will be aggregated across all participating practices prior to being analysed and published.

The survey can be accessed by clicking on the following link: http://www.surveymonkey.com/s.aspx

However, should you experience difficulty in accessing it through the link, please copy and paste it into your internet browser.

I would like to take this opportunity to thank you very much once again for your participation.

Should you have any queries or require further information, please do not hesitate to contact me at roisin.murphy@dit.ie .

Yours sincerely Róisín Murphy BA(hons); PDES; MBS; MSc. Lecturer School of Real Estate & Construction Economics Dublin Institute of Technology Bolton Street Dublin 1

NOTE: If you would rather not participate in the survey, please click on the following link to be excluded from future correspondence. http://www.surveymonkey.com/optout.aspx

B. E-mail seeking participation from wider sample

Message Delivery Schedule Delivery completed on May 5, 2010 11:11. Message Recipients The message mailed to **156** recipient(s). View Message Preview Below is a preview of your message based on the first recipient in your list ([Email]).

To: [Email]

From: roisin.murphy@dit.ie

Subject: DIT/SCS research: Strategic Management of Irish QS Practices

Body: Dear Sir/Madam,

As part of my doctoral research and on behalf of Dublin Institute of Technology and the Society of Chartered Surveyors, I am undertaking a survey of every Quantity Surveying (QS) practice in Ireland. The purpose of the study is to examine the management process of professional QS practices and in particular reflect changes therein.

To date, very few high level studies have been undertaken in this field (particularly in an Irish context) therefore your participation will provide invaluable insight into an area of increasing importance.

The survey contains 31 questions and takes approximately 15 minutes to complete. Answers will be treated with strict confidentiality, and neither individual nor company names will be identified at any stage in the analysis. Results will be aggregated prior to publishing through the SCS.

The survey can be accessed by clicking on the following link: http://www.surveymonkey.com/s.aspx

If you have difficulty in accessing the survey please copy and paste the link into your internet browser.

I would like to take this opportunity to thank you very much for taking the time to complete the questionnaire, it is much appreciated.

Should you have any queries regarding the research or require further information, please do not hesitate to contact me at roisin.murphy@dit.ie .

Yours sincerely Róisín Murphy BA(hons); PDES; MBS; MSc. Lecturer School of Real Estate & Construction Economics Dublin Institute of Technology Bolton Street Dublin 1

NOTE: If you would rather not participate in the survey, please click on the following link to be excluded from future correspondence. http://www.surveymonkey.com/optout.aspx

C. Second wave e-mail

Message Delivery Schedule Delivery completed on May 19, 2010 12:08. Message Recipients The message mailed to **122** recipient(s). View Message Preview Below is a preview of your message based on the first recipient in your list ([Email]).

To: [Email]

From: roisin.murphy@dit.ie

Subject: Strategic Planning in Irish QS Practices

Body: Dear Sir/Madam,

Further to my recent correspondence, you are aware that as part of my PhD I am currently investigating the strategic management of QS practices in Ireland.

The research requires participants to complete an online survey with approximately 30 questions. The survey can be accessed through the following link: http://www.surveymonkey.com/s.aspx

The key to the success of the research lies with achieving a high participation rate. The survey will remain "live" for the next two weeks, so if you would please take the time to complete the survey it would be greatly appreciated. All responses will be treated with strict confidentiality, and under no circumstances will individual nor company names be identified during analysis.

Should you have any queries please do not hesitate to contact me at roisin.murphy@dit.ie

Yours sincerely

Roisin Murphy BA(hons); PDES; MBS;MSc. Lecturer DIT, School of Real Estate & Construction Economics Bolton Street Dublin 1

NOTE: If you do not wish to participate, please click on the following link: http://www.surveymonkey.com/optout.aspx

D. Third wave e-mail

Message Delivery Schedule Delivery completed on June 9, 2010 10:40. Message Recipients The message mailed to **104** recipient(s). View Message Preview Below is a preview of your message based on the first recipient in your list ([Email]).

To: [Email]

From: roisin.murphy@dit.ie

Subject: Reminder - Strategic Planning in Irish QS Practices Survey

Body: Dear Sir/Madam,

Further to my recent correspondence you will be aware that I am currently undertaking research on the strategic management of QS practices in Ireland.

Data is currently being collected by way of online survey containing 31 questions, which takes approximately 15 minutes to complete. Responses will be treated with strict confidentiality and under no circumstances will individual nor company names be identified at an stage in analysis. The survey can be accessed through the following link:

http://www.surveymonkey.com/s.aspx

The survey will be closing within the next two weeks therefore your participation would be much appreciated.

Should you have any queries, please do not hesitate to contact me at roisin.murphy@dit.ie.

Yours sincerely

Roisin Murphy BA(hons); PDES; MBS; MSc. Lecturer DIT, Bolton Street Dublin 1

NOTE: If you do not wish to participate, please click on the following link: http://www.surveymonkey.com/optout.aspx Appendix J

Excerpt from Interview Transcript Exemplar

Company A

...

Interviewer:

In terms of the objectives themselves. Do they tend to be more short run, are they broader long-term objectives? And does it differ? Does it filter down onto a project basis really or...?

Director A

Case by case, There's certain projects have a certain client and you might see an opportunity.. We have particular clients who we take a very long view at. We've done, there's one, that paper that was brought in that I signed there...that's for a client for whom we done 4 substantial projects. He has an interest in two huge projects. Now, they're on hold at the moment in the current climate but we would have a long term understanding with him. We give him a quality of service, our fee is very, very, very competitive, but if he has any quantity surveying, he comes to us. We make sure he is very well taken care of and that approach would mirror to a number of the public bodies and housing trusts clients we always deal with. You're maintaining relationships, you're building on it. I mean the return on these jobs; we've had to reflect the market. You have to be seen to adjust your fees. Still give them quality of service but give them perception of value for money that costs are coming down and so on that level there's an ongoing view there with particular clients but on jobs generally you always, at every job you do, there is an old say...it's a Hackney phrase, you're only as good as your last job but it's very very true. It only takes one stinker of a job to follow you around. Keep doing your work well and you'll get more.

Interviewer:

Well, with repeat business I think it's something we'll come back to in a couple of minutes but the, one thing you've identified there...how do you go about reducing the cost or even reducing the level of fees in the current environment? But yet still improve and maintain the quality that your reputation depends upon?

Director A

At this juncture, you try to do everything as efficiently as possible. But, to be perfectly honest with you, in the current market, you're having to subsidise the quality and the service out of your margin. You're having to sacrifice these margins to maintain

relationships and if you like, if you would ask the simple overriding strategy at the moment it is survival. And that really is it. And I think in being...if we can ride this out, still provide quality of service, that the people that we're dealing with, if we're all still standing at the end, then we'll all move on to better times, you know.

Interviewer:

What are the main difficulties in terms of setting organisational objectives? What are the main hindrances to planning, in general, generally speaking?

Director A

The biggest problem for me is translating what I intuitively or basically do, into a formalised...you have a sense of a lot of these things, without ever having the formal step back and put it into boxes, this is what we do. But for us, I mean the biggest issue for a company this size is, most professional...actually I would say of all professional practices we're the flotsam and jetsam. We're, you know, we're price takers, not price makers. And so, the biggest thing is to keep yourself as informed as you can do. You know, so that none of us can foretell the future, but at the same time, there are certain horses you don't want to back. Or in terms of individuals, projects are also types of work. That's just a general awareness. A general awareness of the climate, the environment you're in. A general awareness of your own business but the hardest thing is translating those general intuitive understandings into a formalised set of goals or objectives. And often they just spell out, for me, if I'm out for a run or in my bed at night, kind of, oh must if we go that way, or if we address that, that might help with that. It's not; there isn't a formal process of planning and management. It's an ongoing, semi-ad-hoc process. It's taking the overall understanding issues that arise and fitting them into an overall framework.

Interviewer:

So, one of the difficult, not difficulties but perhaps a hindrance to strategic planning is that the process itself is probably a little more ad-hoc than formal.

Director A

There isn't a process. There isn't a formal process.

Interviewer:

There isn't a formal process.

Director A

No.

Interviewer:

And do you think that is something that will change over the coming years or...

Director A

It is changing very much. There's a lot more time goes in to sitting looking at your market, your business, your costs. I mean, another Hackney phrase, no shortage of Hackney phrases, is a rising tide lifts our boats. So and there's a lot of organisations in the construction industry fail to understand this. Revenues are lifting and lifting and lifting, and everybody to a degree took their eye of the costs and the structure, you know, they might have adapted the structure for a set of revenues that wasn't sustainable, which you got away with it on the up, because the jobs just kept coming. We would have been guilty of it but to a much lesser degree than most. We were always being highly conservative so our cost base never expanded hugely and our revenues just marched on. Now, it's a different set of circumstances. Revenues are contracting at a rate of knots and like, our turnover this year will be 50% of what it was 3 years ago. At the moment, I can't even see next year's turnover and I'm not being dramatic. Next year's turnover...the projected turnover for next year at the moment is zero. I know it won't be zero, but at the moment it may not be far off. You know.

And, following those circumstances and the change reality means you have to look...you can control your revenue to a great degree. You can negotiate your fees, but then the public bodies for instance, will impose an 8% cut in your fee, unilaterally. There's certain bodies resisting that, but agin we're price takers not price makers. I do a lot of work for Dublin City Council and other public...it not behove me to be giving them two figures and saying well feck you and your 8%. I would rather focus on having 92% and glad of it. So, and your revenue, you can adjust your fees and you can offer value and be seen to be competitive, without slitting your throat, so your clients

appreciate that your trying to reflect the market and they'll still do business with you. But the one area where you'll still do business and spend time is on your costs, because they're the only places you can truly control.

Interviewer:

So, in some ways perhaps creating efficiencies?

Director A

Yeah.

. . .

Interviewer:

In terms of staff, you mentioned earlier on that there's very little staff turnover? What would you put that down to?

Director A

The same underlying thing. I mean we take everything we do has a medium to long term. It's not spectacular, it's certainly not glamorous but there's a continuum...we try to take care of the people that work with us and they work with us, not for us. While all we ask in return is that they take care of us. So, when we're making a few bob we make sure that everyone is making a few bob. They see if that there's jobs, we put in the hours. We don't sit down and say that has to be done; you have to stay at your desk until it's done. But, if you like, to a degree, we lead by example. You know, most of the people here we would have recruited them because we felt that they were the right type of people. As Quantity Surveyors, no matter what anybody tells you... You don't have to be a genius to be a Quantity Surveyor. There's a lot more about attitude and aptitude and the ability to deal with people and most of the people here, they're decent skins and they understand what we're about. So, we never took on, like there was times when we were really busy. We could have done with another 4 bodies, but you couldn't find four suitable bodies. And you'd rather we were 7 days than bring in people who were unsuitable because they just, they cause unrest.

Interviewer:

And is there provision for training and development of staff?

Director A

Yeah. We afford everybody...if anybody has a genuine interest that can contribute in any way to the underlying business; it's encouraged and paid for. In recruitment, one of the things we found is a great success, would be we think, Quantity Surveyors from Bolton Street, but who've done the CTD and would recruit them on the technical ability that they've learned there and we would encourage them, support them, pay for them, give them their time, give them plenty of study leave to do the degree course in construction economic and management degree.

• • • •

Interviewer:

Internally as well, repeat business. If we could talk about that for a little while. Repeat business is very important to your organisation, so you've said. In terms of repeat business, do you ever solicit feedback from your clients and have they identified to you what the strengths of your organisation and why they've come back to you?

Director A

Yep. Again, it's part of the QA process. We have a customer feedback form, which at a milestone, either by telephone or face...ideally face to face, we would ask them to fill out a simple enough questionnaire, I can give you an example of it. It's very useful but what comes back time and time again is things like responsiveness, quality of service, and one thing I suppose, we feel is that if someone asks us for something, we're whores for deadlines. Deliver. Now, on realistic, if somebody asks for an unrealistic deadline, you're not doing anybody a favour by saying, oh yeah, I'll do that. You say, you're not going to get it Wednesday....you'll certainly get off by the end of the week. But when you say you'll get it for them at the end of the week, come hell or high water, they get it at the end of the week and they get it to the standard they require. And that comes back again and again. That responsiveness and the delivery of outcome is a big issue for them. You know?

Interviewer:

Have any clients identified problems or difficulties in those feedback forms.

Director A

Sometimes. I mean, it has been known occasionally that another strength, they always like dealing with the fact that they knew if they were dealing with X or I, they were dealing with the principals. But it was noted on the form that it is only the two of you, and you know, you can only spread yourself so thin and while I've had a period of grace since X, a few people have noted to me already, well, how are you going to manage, you know. It's difficult enough to spread two people thinly; it's next to impossible to spread one.

An issue would be to make sure...for people to see that there's a structure coming through and that the people who are here are the next generation, if you like. But it was always an issue. I said at the start, we were always, there was only two of us, you could only do so much. It is...it's a strength in that they know they're always dealing with you. But when they're dealing with you enough, they'll realise that you're spread thin, they've never been let down, but they would note it.

•••

Interviewer:

And have you had any experience of the new form of government contract?

Director A

Those projects, they would be tendered on it, but we would only have one small one on site with it.

Interviewer:

And have you completed any projects under it yet?

Director A

We've one small one of half a million, just finished. Because the builder doesn't want to misbehave and anyway, if he does, the new form of contract is such that you just...it's like Wiley coyote...they just drop on a rock on the road. It's a horrible contract and contractors are tendering on it, savagely, without reflecting the risks inherent in it. Like the social housing project I was talking about. The budget for that is 6 million, the tender is 4.7 and that's on the new government form of contract. Lump sum, fixed price, there you go, don't even come back to us kind of thing...

Interviewer:

Yeah. Again, with regard to the macro economy, do you keep and how do you keep abreast of gathering information? Is there a formal process for example, to keep an eye on the economic variables, does anyone have responsibility for it or is it expected as part of the daily...

Director A

I do it, at the simplest; it's gone through the business section in The Times. It's reading, you know, I mean I'm, I read Bruce Shaw, I make sure we always have a copy of Bruce Shaw, the PKS reports; even though they can be by their nature historic by the time they're out. Things like the DOE, Outlook and Review – they tend to be historic but I would say the most current information you get is the news. The print media, the radio, the television news, and you would be amazed at how many people you deal with never turn on the news, and not in a good way.

Interviewer:

Do you think it could possibly be an area that could be developed in terms of staff development or do you think it's really needed to that extent.

Director A

No, I think, there has to be an awareness of it and it has to inform everything that you do. But the thing that you can actually react, you know if formally react or formally influence these things, for an organisation of this size would be very difficult, you know. You have to be aware of it, you have to cut your cloth accordingly, you know you have to shift your emphasis as you see fit but they wouldn't, I mean, the size of the organisation, everybody's a hands on surveyor. I would expect them all to have the wherewithal to keep themselves abreast of current affairs, economic...issues that impinge on the business. And we chat about it, you know, and I may inform...well, see such and such has gone down such a route, but for the most part it's not a formal thing.

Interviewer:

In terms of the industry itself, obviously it's dependent to some extent on the market, you know, how there's a huge amount of uncertainty. If you had to hazard a guess as such, how do you think the industry is likely to change over the next number of years as a consequence of what's currently happening?

Director A

Well, a lot of it will be good. It's going to be painful but there's a purge, there's a huge amount of deadwood and inefficiencies. I mean, put it this way, if guys can tender at the level they're at now – albeit some of them are at, or below cost, it goes to show that the cost level would...you know...goes way below the market level. Now, what was really inefficient in that, if you ever look at the actual profits that these guys returned, their margins weren't, they weren't what you would call super normal or excessive. A lot of the big contractors would be making large profits in volume, in terms of an absolute amount of money. But if you put it back against the volume of the turnover needed to generate that -4, 5 percent, at a time when you look at tenders and they were relevant to where the market was, which would suggest that there is a 20 to 30 percent swing in prices. So, that would suggest that, you know, there is 20 percent between cost and price. So, there was a huge amount of inefficiencies in money not...subcontractors, suppliers, delivery, execution of the works. There was an awful lot of money that just disappeared into the ether. Now, if I went home and whacker the block layer or Joey the carpenter and he bought this 4x4, but even those are a nonsense. So we're going to have to get back to professional practices and contractors are managing the resources, efficiencies, getting back to running a business on a tight margin if any margin and getting back to good principles of cost management. The other thing, naturally, I mean, the Irish market, like a lot of the firms, the firms, the constituent firms within the Irish market, both in terms of contractors and professional practices, grew to such a size based on a value of work in the market that's completely unsustainable now. So, on one level you have massive layoffs in engineering and architecture and surveying, equivalent to 40 to 50% redundancy, particularly on the junior end. Contractors...you know, you can let go the labourers and a lot of the operatives, but you still have organisations there that are a lot bigger now than they were in a market, even if it comes

back, is not going to sustain them. So, a lot of the contractors are looking abroad and the professional practices similarly.

• • •

Interviewer:

And that actually brings up an interesting point. Do you market yourselves, apart from the SCS; do you do anything by way of marketing?

Director A

It goes back to the core principle. We get work, we do work well, we try to tell people this is what we do. We get ourselves out there in person, as much as we can. Not the social events, not the networking, not to dinners or drinks or sporting events. Getting work done and delivered.

Interviewer:

And where would you go to try and solicit the jobs?

Director A

You deal with people, and we take care of them and we know certain people who know certain other people and you say...listen, if he is doing something, for all means ask them, often we've looked at stuff speculatively. You know, we say we'll have a look at it and let them see what it is we do. And that's the best marketing we have. Rather than tell people, promise we'll do A, B and C, let them see that we do it. You know?

•••

Interviewer:

So, am I right by saying that it's your understanding that some of the competition have cut fees to an unsustainable level and that this organisation has chosen under those circumstances not to compete on that basis, for certain projects?

Director A

Some of the organisations that have cut it to unsustainable...are newer arrivals in the market or they are firms who grew very, very significantly. There are a number of firms who would have been similar size to ourselves 10 odd years ago, are multiples of ourselves today. So, they have different cost structures, they'd have different overhead obligations...you know, they'd have different liabilities. I mean, you can reduce the staff numbers to a degree, but I know somebody who would have two offices or larger premises, etc, to serve it. And a lot of professional practices as well as contractors are carrying debt.

They might have invested in projects, they might have other obligations and that will have...you know, they have other obligations and liabilities that we don't have. SO, we're trying to be mindful – these are our costs, this is our overhead, we've squeezed them down as tight as we can, we know what we need just to wipe our face and that's what we're trying to do.

Interviewer:

And does this organisation seek information on competitors? Do you actively seek information?

Director A More intuitively.

Interviewer: So, it's intuitive?

Director A

And we'll seldom get into a Dutch option, I mean you're not meant to anyway, but you're not. I know there's competition but I wouldn't be interested. If I had a client who said, I have so and so he says he'll do it for that, will we do it for less. Well, when does he then take my price and go to someone else. I don't want to be dealing with that. Because back to personal relationships and the people we do business with, we try, we do reflect the market in our fee and I mean, off our own bat, we're saying things are cheaper now, we've cut out costs, and we're looking to offer them....you're only fooling yourself if you're putting in a flat fee, which is not reflective of the market.

You're better off to tell them that your fees are reflecting the much more competitive climate, than have them come to the realisation, or God forbid, have someone else knock on the door and say, "Jesus, you're man is saying he'll do it for that, I'll do it for half of that." And it'll happen. I'm sure it does happen.

Interviewer:

But you don't necessarily go seek information...

Director A

No.

Interviewer:

... On the competition? So, to some extent you don't really undertake some kind of competitive analysis, as such?

Director A

No. Too busy....

Interviewer:

Competitor analysis, I should say.

Director A

No, too busy worrying about ourselves. What I do know, I mean, again you see what certain people are doing. How they do it, the quality of what they produc

Appendix K Pertinent Points & Observations for Exemplar

Company Background	Company A
Established	1948
No. of Employees	10 – 9 QS & 1 administrator
Sectors serviced	Mainly public sector
Services provided	QS
Other salient issues	Embarking on international alliance

Strategic Planning		
Flow (where do initiatives come from?)	Market driven; Resource based approach	
Formality (formal or informal)	Informal. Written business plan (ISO require)	
Comprehensiveness (depth of analysis)	Not in depth; industry reports and national press	
Intensity (frequency of SP)	Ad hoc mostly. ISO 3 year requirement	
Participation (communication/feedback)	Client feedback forms; monthly staff meeting	
Integration (<i>budget, performance</i> ?)	Financial ; ISO requirements	

Objectives	
Strategist (analyse, defender, prospector)	Defender
Mission	None
Risk (taker, averse, neutral)	Averse

Analysis & Diagnosis		
Macro Environment:		
Economy (national & international)	Interest rates (+); securing finance (-); demographics(-); income (-); Demand (-)	
Government	Capital expenditure (-); 8% fee cut(-).	
Industry Environment		
Profession (changes)	Fees reduced;	
Related Professions	Has relationship with related practices (architect)	
Professional body	Ineffectual	
Segmentation	Apparent; not affected larger practices	
Internal Organisation		
Staff	High retention; training and development	
IT	External consultant; improvements being made	
Marketing	No marketing plan. Reliant on word of mouth	
Finance	€1.6m turnover 2008(peak) forecast €1m in 2009	
Competitive Position		
Competitor analysis	Cursory.	
Intensity of Competition	Very intense; prices/fees falling	

Choice	
Generic Strategy (<i>stability, expand, retrench</i>) (cost leader; differentiation; focus; segment)	Survival; Combination – differentiation (quality nd cost

Strategic variation (diversify alliance, merge)	Alliance; international market
Strategy Choice (scenarios; SWOT; risk,)	Do formal use SP tools; aware of SWOT

Implementation		
Constraints (on SP and implementation)	Uncertainty; economic environment;	
Monitoring/Feedback (communication)	Monthly meeting with staff; client feedback form	

COMPANY A Pertinent Points

- 1. **Increasing focus on strategic planning** respondent mentioned that in recent past and into the future strategic thinking would become more focused. This may be due to more time to consider issues, the necessity to think strategically in the current environment.Currently it is done in an ad hoc fashion.
- 2. Alliance with Northern Ireland QS Practice key change in the strategy of the company. This had been mentioned during the meeting in January. It is now at early stages (there is an office in Bahrain) and seen as a contingency plan while focusing on domestic market. Had previously worked with the practice on another project.
- 3. **Repeat Business** absolutely essential. No formal marketing and projects awarded based on reputation of company. Taking long term view with some clients in terms of payment for work completed
- 4. **Staff considered very highly**. Care taken in their recruitment; training and development available to all; on the job training; internal promotion; rewarded when company in a position to do so. Open communication and monthly meeting
- 5. **Professional Body** highly critical of usefulness of SCS. Compared and contrasted to IEI (Institute of Engineers of Ireland).
- 6. Attitude towards the future although the threats to the industry were apparent the respondent was still reasonably positive about the companies future. It was not doom and gloom and respondent seemed to be confident that if they looked after their existing client base well and kept "in" with some clients that the loyalty shown would be rewarded in the future
- ISO Certification documentation exists as a requirement for ISO certification. Objectives have been set, however remain reasonably general and have not altered significantly in the last number of years despite the changing economic and industry environment.
- 8. **Management style and communication** conducted on an informal basis and through monthly staff meetings and client feedback questionnaires which are routinely distributed.
- 9. **Buzz words** no formal SP tools used, however respondent regularly used strategic terminology (e.g. SWOT; price takers; supernormal profit, barriers to entry etc)
- 10. Key areas for the profession in areas related to sustainability, value management and M&E (mechanical and electrical) areas.

CONTACT SUMMARY SHEET

Company Name:	А
Position of Respondent:	Director
Date of Interview:	17 th June 2009

1. COMPANY INFORMATION

- Established in 1948. Currently 10 Employees (9 QS's and 1 administrator).
- Provide core QS services.
- Sectors serviced primarily non-residential and smaller public sector projects.

2. SUMMARY OF KEY ASPECTS OF EBS SPPM WDTDW:

- Directors set objectives. Plan for ISO purposes.
- Conscious decision to limit company to max of 20 people. Objective was to increase turnover in serviced markets. Caution with residential sector – did a number of mixed use.
- Repeat business is essential. Good relationships with client & other professional practices
- Informal structure. Management do not work in separate office "do as I do". On the job mentoring
- Written business plan in place (ISO requirement). Resource-based approach to strategic planning. Informal process

Analysis & Diagnosis:

- Macro environment: recession uncertainty; projects not going ahead; construction price deflation; interest rates low however availability of finance still a problem; 8% drop in public sector fees.
 ~ Information comes through Bruce Shaw handbook. DLPKS reports and national press
- Competitive environment: no formal consideration of competitors. No formal marketing. Competition driving prices down - unsustainable. Market segmentation apparent.
 - Competitive position: strategic groups apparent. Not in same league as large firms. Not trend setters ("sheep"); align to what is happening in market place; price takers not makers; Hear things through grapevine and through clients
- Internal:
 - Staff: good staff essential; Invest in training. Outsourcing some of the specialised aspects (M&E and IT). Good staff retention in company – promote from within (3 move to senior QS status). Staff apprised of what is happening during monthly meetings. Look after them and they will look after you.
 - Efficiency: now a chance to examine document management system (IT). ISO documentation.
 - Finance: turnover 2004: €1m; 2005 €1.5m; 2006 €1.9m; 2007: €1.8m; 2008: €1.6m; 2009e
 €1m
- Industry
 - SCS absolutely useless. Does nothing for them. Compared to IEI

Choice

- Generic Strategy: Survival. Stability is the corporate level strategy
 - ~ **Business Level:** differentiation. Rely on quality of work undertaken.
- Strategy Variations:
 - Differentiation (offer quality service to client; personal approach); Currently low cost competition driving fees down.
 - ~ Alliance with Northern Ireland QS practice gain foothold in Bahrain market. Very early stages. Seen as a contingency and not intended to take over from core business in Ireland.
- Risk attitude: risk averse. Started to undertake "ugly" work before downturn started. Taking LR view with some clients if they having difficulty paying for work completed.

Implementation:

- **Constraints:** obtaining finance; uncertainty; difficulty in translating what is known implicitly into strategic plan. Do not use SP tools such as SWOT.
- Monitoring: Member of the 25 club and members of the SCS; client feedback forms.

3. GENERAL OBSERVATIONS

- More focus on strategic planning since Director died in May 2008.
- Opportunities: work on efficiencies; staff training; improve document management system;
- Threats: interest rates (+), income (-); demographics (-); declining demand (-); oversupply (-); intense competition (-)
- Buzz words: sustainability; value for money; M&S services

Appendix L NVivo 8 Coding Strategy
Nvivo 8 Coding Strategy

The following sections provide detail of the coding strategy adopted for the project. As is evident, the coding structure is broadly based on elements of the EBS SPPM.

<u>Broad Coding Categories</u> General Company Information Objectives Analysis and Diagnosis Strategic Choice Implementation Strategic Planning

Coding On

The broad headings of the EBS SPPM were expanded to include a number of subheadings as follows:

General Company Information

- Sectors serviced
- Services provided
- Number of people employed
- Year of establishment
- Organisation structure

Objectives

- Risk
- Strategist

Analysis and Diagnosis

- Internal organisation
- Competitive environment
- Macro environment
- Industry environment

Strategic Choice

Corporate level

- Business level
- Strategic variations
- SWOT

Implementation

- Flexibility
- Constraints
- Communication
- Monitoring and feedback

Strategic Planning

- Flow
- Formality of process
- Comprehensiveness
- Intensity
- Integration
- Planning tools
- Participation
- Approaches to strategy

Further Coding On

The coding structure was further expanded to include the following

General Company Information

- Sectors serviced
- Services provided
- Number of people employed
 - ~ Decision to restrict employees
- Year of establishment
- Organisation structure

Objectives

- Risk
 - ~ Risk Neutral
 - ~ Risk Management
 - ~ Risk Averse
 - ~ Risk Taker
- Strategist
 - ~ Strategic Type
 - o Reactor
 - o Defender
 - Prospector
 - o Analyser
 - ~ Lifestyle influences
 - ~ Management Style

Analysis and Diagnosis

- Macro environment
 - ~ Economy
 - ~ Government
- Internal organisation
 - ~ Organisation meetings
 - ~ Culture
 - ~ Quality
 - Quality of service
 - o ISO
 - ~ Core competence
 - ~ Finance
 - o Financial Management
 - Efficiency
 - o Revenue
 - Cost base
 - ~ Information Technology
 - $\circ \quad QS \text{ function and } IT$
 - o Reliance on IT
 - Office Management and IT
 - ~ Human Resources

- Work practices
- o Training and development
- o External training
- o In-house training
- o Salary
- o Recruitment and retention
- ~ Marketing
 - o Marketing plan
 - Marketing the profession
 - o Reputation
 - Repeat business
 - \circ Word of mouth
 - Public sector projects and repeat business
- Competitive environment
 - ~ Fees
 - ~ Market Segmentation
 - Competitive Reaction
 - Competitive Position
 - o Competitor Analysis
 - No analysis
 - Formal analysis
 - Informal analysis

Industry environment

- ~ Construction Industry
 - Project uncertainty
 - Change in construction industry
- ~ Quantity Surveying Profession
 - Change in profession
 - o Profession structure
 - $\circ \quad \mbox{Future role of the QS} \\$
 - o Related Professional
 - Professional Body (SCS)
 - Involvement in the Society
 - Areas for improvement
 - Limitations of SCS

• Strengths of SCS

Strategic Choice

- ~ Corporate level
 - o Stability
 - Survival
 - Expansion
 - o Retrenchment
 - \circ Combination
- ~ Business level
 - Cost leadership
 - o Differentiation
 - o Focus
 - Stuck in the middle
- ~ Strategic variations
 - o Diversification
 - International Markets
 - Alliance or partnership
 - Mergers and acquisitions
- ~ SWOT
 - o Strengths
 - o Weaknesses
 - Opportunities
 - o Threats

Strategic Planning

- ~ Approaches to strategy
 - o Emergent
 - Resource-based
 - o Planned
- ~ Flow
- ~ Formality of process
 - o Formal
 - \circ Informal
- ~ Comprehensiveness
 - o Cursory

- Comprehensive
- ~ Intensity
- ~ Current plan on hold
- ~ Integration
- ~ Planning tools
- ~ Participation
 - Director participation only
 - Director plus senior staff participation
 - Company-wide participation
- Implementation
 - ~ Flexibility
 - ~ Constraints
 - ~ Communication
 - ~ Bottom-up communication
 - ~ Monitoring and feedback
 - Feedback from staff
 - Feedback from clients

Appendix M Examples of Exemplar Case Coding in NVivo8

Exemplar Case Coding in Nvivo 8

The following screenshots have been taken from the project created in Nvivo 8 for the purposes of the thesis. The case chosen as an exemplar is Company A.

Each heading represents an element against which all participants were analysed, and the screenshots display an example of coding from interview transcript under a number of headings for Company A.

Strategist (Management Style):



Risk Attitude (Averse)

🚯 Strategic Planning in Ir	ish Quantity Surveying Practices.nvp - I	Wiva	
Elle Edit View Go Project	t Links <u>C</u> ode <u>T</u> ools <u>Wi</u> ndow <u>H</u> elp		
New • 😸 🖻 🖬	■ ■ ****	- 5 9 Q 🔲 E A A 1 7 4 M * 👘 💡	
- Anal	- 10 A-BZUZ	▲ 百百百百百百百百百百百百百百百百百百百百百百百百百百百百百百百百百百百百	
> D D0 00 00 00 0	10 10 10 mil		
E Codealt Mame -		e todes in a construction in the state in th	
Nodes	Look for:	Search In Tree Nodes Find Now Clear	Options 👻 🗙
Free Nodes	Tree Nodes	Risk Averse	×
Cases Relationships Mathices Search Folders	Name Name Name	Sintermals/Interviews/Dompany A2- & 3 references coded [D.67% Coverage] Reference 1 - D.12% Coverage We were always being highly conservative so our cost base never expanded hugely and our revenues just marched on. Reference 2 - D.34% Coverage Take the long term view. And, it goes back to everything we do is the long term view. You make a killing tomorow, or last week, the year before, but when you've made it, it's gone. Whereas, this business has been around since 1948. I'd rather it was going to be around for another 6D years after I'm gone you know Reference 3 - D.20% Coverage And I know that sounds really fuddy duddy and old school but we would be more old school. We make sure that no one gets responsibility to do things until they've learned to do it properly.	Summery Releases Tex
Sources		<internals\interviews\company c="" transcript=""> - § 1 reference coded [0.57% Coverage]</internals\interviews\company>	
Nodes		Reference 1 - 0.57% Coverage	
Sets		we don't have a huge resource that we can just dispatch them to Poland and say, "go hustle over	
Queries		there". So to be honest with you, and it's from a position of prejudice maybe and not a lot of knowledge, that I feel that to go at it now, and chase it now, it's going to take a bune time overhead.	
Models		and resource overhead. That, maybe we would be better off just keeping our powder dry, you know.	
Links		<a>statemalsUnterviews\Company D Transcent> - 61 reference coded (0.36% Coverage)	
Classifications			
Folders		Reference 1 - U.do % Coverage	
	*	in terms of the risk attitude of the company. would you say in general risk takers or risk averse.	<u>×</u>
ARM 159 Items Sources: 7 F	References: 15 😻 Unlikered IAIN FOLDER 😗 Strategic Planning in I	Downerst - Nerssof	EN 🖮 🔍 🔊 🖏 🍓 🧐 20:46

Strategic Type (Defender)

B Strategic Planning in I	rish Quantity Surveying Practices.nvp - N	Vivo	
Ele Edit View Go Proje	rct Links ⊆ode Iools <u>Wi</u> ndow Help		
New - 😸 🖻 🖬	10 10 11 - 1 1 1 = 1 = 1 = 1 = 1 = 1	- # 9 🕲 🔲 🗉 - A. A. A. F 📥 🖬 - 🦢 👘	
= Anal	- 10 A-BIU Z-2	A FFEE BALLA, DO UNA,	
5 D BN 40 D0 ND 1	0.0 0.00 00.0		
E Gode M Hughe 👘 🖛	- In The Mark	Notes the second of the second s	
Nodes	Look for:	Search In 🔫 Tree Nodes Find Now Clear	Options 👻 🗙
7 Free Nodes	Tree Nodes	Defender	×
Alexionations Antices Search Folders	Phase 1 Broad Coding Phase 2 Coding On Phase 2 Coding Phase 2 Coding On Phase 2 Coding On Phase 2 Coding On	stritemakulnetwisewiCompany A2- S7 references coded [5/03% Coverage] Reference 1: 181% Coverage But, most of what we do is fairly intrins in terms of everybody understands it. We get a job, our aim is to the do the job well. most of our work is repeat work. And if it's not repeat work directly with the person, it's more often than not word of mouth from one to another. So, the underlying principle is, we look to provide quality service, take care of things and be afforded you know, afforded opportunities out of other opportunities. Thereafer, it's the usual things. It's generate the turnover, get a return on your turnover. We made a conscious decision, even at the height of the thing that 15 people was our upper limit. That we didn't want to exactually, this office can accommodate 20 people, but we never envisaged going near that because we with Director X and myself, even managing it at 14 people, you needed a whole new layer of management and we weren't set. You know, we were changing fundamentally you change the structure of the organisation and if you go back to be underlying principle, of quality control and that, the quality control and the message dissipates. You know, you start to delegate out more and more, you lose we're control freaks, we're not great delegators. There is an element of everything that goes through here, goes through myself or either Director X or the other 3 now, understand the message, they take responsibility for their stuff.	Sumaiy Releated Teel
Sources Nodes	Strategic Choice	So, I suppose there is, without ever consciously getting it, we're a small to medium size enterprise. There's a niche for us there, we're comfortable in what we do, we're big enough to take on decent size projects and we're small enough to roll into a ball when we have to.	
Sets	E 👷 Strategic Planning	Reference 2 - 0.76% Coverage	
Queries			
Models		This is the way that we do things. Therefore, everybody in the place understands the message. They understand the way we do things and they do the things the way they should be done. When	
Links		looking to recruit people, we're very slow. We're not slow to recruit, but we like to be careful enough	
Classifications		good people and we'd have had the sense of them, most of them before they came here. And most	
Folders	2	of them understand what we're about. They're here for the long haul. There's very little staff turnover, there's never really beenwe've never really had people coming in demanding money for this or the	
9	•		<u> </u>
A HM 159 Items Sources: 2	References (1 W/ Unulered	Telesconce and the second	
Start DBA	MALLY POLOCIK B Strategic Planning in I	a coconence mercent	

Number of Employees (including decision to restrict size)

🚯 Strategic Planning in I	rish Quantity Surveying Practices.nvp - N	Vivo	E 🗗 🔀
Ele Edit View Go Projec	t Links ⊆ode Tools Window Help		
New - 😁 🖻 🖬	10 10 M + 2 = + = +	- 🍠 🔊 🗐 🔲 🗛 🏭 🛱 📥 🗤 👘 👘 👘	
1 I	<u>A-BIU</u>	2-三三三三三日に日 4 1 2 5 3 3 3 4 3 2	
D IN 10 10 10 10	D0 60 as -		
E Code M +	4 10		
Nodes	Look for:	Search In 🔻 Tree Nodes Find Now Clear	Options 👻 🗴
Free Nodes	Tree Nodes	General Company Information	×
Cares Relationships Matices Search Földers	Name Name Phase 1 Broad Coding Phase 2 Coding On Phase 2 Coding On Phase 2 Coding On Phase 3 Concern Information Phase 3 Concern Phase 3 Conce	Sintemalalinterviews0Company A2- & 4 references coded [0.62% Coverage] Reference 1 - 0.14% Coverage It hasn't changed honestly, Currently, there is nine OS's and one Admin. At the peak there were thirteen OS's and one admin. Reference 2 - 0.04% Coverage We would have outside support like Trevor Reference 3 - 0.07% Coverage An I.T. guy who we use a lot. Like Trevor's in it quite a bit. Reference 4 - 0.37%. Coverage	Summary [Reference] Tert
	Sy Objectives Strategic Choice	We made a conscious decision, even at the height of the thing that 16 people was our upper limit. That we didn't want to exactually, this office can accommodate 20 people, but we never envisaged, going near that because we with Director X and myself, even managing it at 14 people, you needed a	
Sources	e ar suategic Hanning	whole new layer of management and we weren't set.	
Fals		<internalsunterviews\company b="" transcript=""> - § 9 references coded [1.89% Coverage]</internalsunterviews\company>	
Quaviar		Reference 1 - 0.05% Coverence	
Models		Neleience 1 - 0.05 te coverage	
Linke		At peak we would have had 17	
Classifications		Reference 2 - 0.43% Coverage	
Folders	»	Now you would have had 2 secretarial, they would be admin, and the rest would have been QSs. Even our directors are QSs because our directors don't just play golf, they do QS'ing as well as everything elso. Take of quantities, the lot	*
🔏 RM 159 Items 🍯 Linked	Sources: 11 References: 59 🥳 Unlifered		
🚺 start 🛛 🔁 DBA1	MAIN FOLDER 🚯 Strategic Planning in I	Document1 - Microsof	EN 🖮 🤇 🖉 🖏 🌄 20:49

Sectors Serviced

8 Strategic Planning in Irish Qu	antity Surveying Practices.nvp - N	Vivo	🗐 🗗 🗙
Ele Edit View Go Project Links	<u>C</u> ode <u>T</u> ools <u>W</u> indow <u>H</u> elp		
	n	- 5 9 G G E A A & F & M · · · · · · · · · · · · · · · · · ·	
-	- A-BIU 2-3	TETTTE BALLON UN N.	
5 D BW M M M M 00 0	10 M 10 - 1		
Code M 1	- m 1a	1.86 NT-55.	
Nodes	Look for:	Search In 🔹 Tree Nodes Find Now Clear	Options 👻 🗙
7 Free Nodes	Tree Nodes	Organisation Structure	×
Cases Relationships Matrices Search Folders All Nodes	Name Phase 2 Broad Coding Phase 2 Coding Un Phase 2 Coding Un Phas	StritemalalinterviewalCompany A2+ 6 11 references coded [4 40% Coverage] Reference 1 - 1.01% Coverage It would have changed. Traditionally, we actually always do a bit of averything and I don't want to make that sound too random. We would have always had a decent dollop of private work, both commercial, by commercial I mean office, retail, industrial. But a lot of public. We do a lot, we have a very long standing relationship with a large housing association in town. The hy Trust. So that relationship has been running for 40 years. So, anything they do, we do. We have done, you know, social housing for various local authonties. Alot of schools over the years. A significant amount of religious orders also. The nuns, the priest hemselves, a couple of the schools like St. Mary's college, Holy Chosts, Alexandra's College, The Nuns, we're involved up in Southyou know, we do a bit like that. Then, obviously at the height, when the residential was ploughing on, we actually rarely had the one private residential. C Reference 2 - 0.95% Coverage I said that traditionally we were never involved inwell, traditionally; most people were never involved in private residential because, at the time, it was low density, housing estates. Housing	Surmary (Polience) Ter
Sources Nodes Sets Queries Models		estates, if you're going to get a OS involvement, you'll involve them in the site development. The roads, the drains, the digging. The houses themselves, most builders can do it, nule of thumb, but when the market changed to a lot more high rises, higher density apartments and the likes, there was a market there. So we were doing quite a bit of that. Not as much as some offices but had started to back away from it, hwo, three years ago, because people we were directly involved with were backing away from it themselves. So, when the private residential slowed way down, we had yary little private residential on. A lot of private commercial and a number of those are on hold.	
Links		Reference 3 - 0.41 % Coverage	
Classifications Folders *	í	But, at the moment, our old, reliable, public sector, and public sector like, HSE, we've had HSE have put a stop on all capital projects at the moment. So, we have 2, 3, a primary care centre and two little hospital units on hold at the moment, and, but we've a couple of little 4 classroom extensions. It little school looks and little social housing iobs and houses in side aratems.	~
RM 159 Items A Linkert Source	s 11 References 60 👹 Unilland		
Start DEA MAIN FO	LDER 🚯 Strategic Planning in I	Documenti - Niccosof	EN 🖮 🤆 🔊 🕼 🛰 🛃 😕 20:50

Services Provided

B Strategic Planning in Ir	ish Quantity Surveying Practices.nvp - N	Ϋίνο	
Ele Edit View Go Project	t Links Gode Iools <u>Wi</u> ndow Help		
New • 😸 🖻 🖬	10 10 M - 2 11	- 3 9 9 🔲 🗉 🗛 🕹 🖬 👘 👘 👘	
-	A-BIU Z-2	2- E E E E E E E E E E E E E E E E E E E	
	0 0 M as - 1		
E Codeal I	- m 1a		
Nodes	Look for	Search In 🔻 Tree Nodes Find Now Cléar	Options 👻 🗙
Tree Nodes	Tree Nodes	Services Provided	×
Cares Relationships Matrices Search Folders	Name	Stritemals/Interviews/Company A>+ § 6 references coded [1.93% Coverage] Reference 1 - 0.05% Coverage we're a traditional Quantity Surveying Services. Reference 2 - 0.43% Coverage At various stages we do provide a project management option, but to be perfectly honest with you, it's not worth the hassle. I have an opnion that when one steps into the PM's shoes, everybody else switches of their brains and thinks, "ah, he'll Soit to ut". "Is to time consuming. You see, the size of the organisation is such that it, even at the peak, everybody is a hands on surveyor. Reference 3 - 0.08% Coverage	Summary [Relievence] Text
Sources Nodes Sets Queries Models Links Classifications Folders	B & Dipeteres B & Statege Draining B & Statege Planning	So, there's no room, there's very little room for a standalone PM function Reference 4 - 0.59% Coverage we would offer as part of the service inherently, you know, we have a good, we'd have a good grasp of property values and development appraisals. We have good, you know, reasonable management and co-ordination skills. We'd have a reasonable grasp on M&E services. We don't offer them as a bespoke service but we'd say that, intrinsic in the GS Service, we have an awareness or, we have a, you know, there's elements of the service we can offer to add value on those kind of headings, but they're not, they're not explicit stand alone services. Reference 5 - 0.58% Coverage We're doingdabbling in a little of legal and accountants work, assisting, you know doing a few expert winess ytype things. Small stuff, but its a few bob here and there. It doesn't register in the great scheme a tithe moment. It's like the house in the side garden, everything is a little bit of revenue, it's a little bit of turnover. It contitutes to the wage bill and the overhead and then, I suppose outside of thatthat's in the domestic marketwe're looking to see if we can bick u work	*
📥 RM 159 Items 🥥 Linked	Sources: 11 References: 56 🤯 Unlikered IAIN FOLDER 😗 Strategic Planning in I	🗃 Documenti - Nicrosof	EN 🖮 (\$)\$" 🛱 🌭 🧱 😕 2051.

Macroeconomic Analysis

B Strategic Planning in Iris	sh Quantity Surveying Practices.	ivp - NVivo	
Ele Edit View Go Project	Links <u>C</u> ode <u>L</u> ools <u>Wi</u> ndow <u>H</u> elp		
New - 8 8 8 4	0.00.00	·배·경영영 🔲 등 유규 회부를 배· 👘 👘	
= Anal	-10 A-B / U	∠·△·●亨書集 導進任任 A (19 10 10 10 10 10 10 10 10 10 10 10 10 10	
5 10 10 40 10 10	10 10 M at		
Code M Marie -		le Rue Nodes ····································	
Nodes	Look for:	Search In Tree Nodes Find Now Clear	Options 👻 🗙
7 Free Nodes	Tree Nodes	Concorny Sovernment	×
Calaes Relationships ∰ Matrices Search Folders ⊯ All Nodes	Name Name Name	sinternalsUnterviews/Company Ab - 96 references coded [3 41% Coverage] Reference 1 - 0.44% Coverage You can negotiate your fees, but then the public bodies for instance, will impose an 8% cut in your fee, unitaterally. There's certain bodies resisting that, but again we're price takers not price makers, not al of dwrk for Dublic Cty Coucil and other publicth rot behow me to be gring them two figures and saving well feck you and your 8%. I would rather focus on having 92% and glad of it. Reference 2 = 0.39% Coverage I's hard enough to generate the work, but then you have to be mindful of, I mean, you know, am I generating something that I'm going to get paid for. And that goes for public bedies as well. There's been a slow up on, you know, getting your money in and, as I've said, the public sector, the Department of Finance out a circular 8% off all lees, unilaterally. Reference 3 = 0.99% Coverage	Surving Petersce Text
Sources Nodes Sets Queries Models Links Classifications Folders	a 24 steels rand.	Intere solviously the tigmening of the capital purse. Overvously, the HSE amouniced all capital projects are on hold. So, we have time projects that are stopped dead, the Department of Environment have again, the forms for social housing have dired up. The, so, on the macro environment, the public purse, the public and private purse are servery restricted; if not, don't exist. But, the other side of it is, you can, like we've certain public, schools, religious orders and that. These are great bodies, they have funds and now is the best time to do a capital project because there's more, you can get more bang for your buck. So, we have a number of clients, the less glamorous variety, the runs, the brothers, the old Franciscan frans, the school. You can get a lot of value for your money. The contractors are cheaper. Then there's a little bit of innovative thinking on some of the social housing. Reference 4 - 0.69% Coverage If you go for your interest rates are down in basement level, if you can get access to finance. So, a	
	*	housing association, for instance, whose balance sheet has a huge amount of fixed assets, what	×
🛃 RM 159 Items 🏮 Linked S	Sources: 11 References: 119 😻 Unliker	d d	
🛃 start 👘 🏉 Hotmail	l - murphyrcis 📴 Appendices	🚯 Strategic Planning in I 👹 Exemplar Case A Ivi 💾 EPSON SX510W Serie	EN 🛲 🔇 🍓 🗞 🕼 🔊 11:09

Industry Analysis

Strategic Planning in Iris	h Quantity Surveying Practices.nvp - NV	iva	E 8 🗙
Eile Edit View Go Project	Links Code Tools Window Help		
New • 😸 🖻 🖬 🖷	0 - 0	398 🔲 등 유유 회부출 🖬 · · · · · · · · · · · · · · · · · ·	
-	- A-BIU Z-2	FEERE BEEL A. DO UU N.	
D D DH 44 DA 500 00	00 10 -		
E Codeal/			
Nodes	Look for.	Search In 🔹 Tree Nodes Find Now Clear	Options 👻 🗙
Free Nodes	Tree Nodes	Construction Industry	×
Caree Relationships Monice Search Folders	Name Phase Broad Doing Phase Broad Doing Phase Coding On Phase 2 Coding On Phase 2 Coding On Phase Dispositive Environment Phase Dispositive Environment	InternalsWnterviewsUCompany, A2 - \$5 references.coded [1:57% Coverage] Reference 1 - 0.05% Coverage You see, we work, we work always a year in advance Reference 2 - 0.34% Coverage Before a project is ever going to get near a site, it comes to us. So, at the moment, because of uncertainty about in the public purse, difficulties in both the private sector both in terms of you know, commercial status, but also access to finance, people aren't anticipating and planning for their projects next year Reference 3 - 0.82% Coverage like anything I'm working on today, I've been working on, at least if it started this year, I've been working on toince 12 months ago. I don't have the next 12 months work because of the underlying uncertainty. Then beyond that, the biggest issues would be because of the constrictive business climate, is getting, you know, getting, work and, yetting work and future work. Then more your morks	Summary References Text
Sources Nodes Sets Queries Models Links Classifications	ia 🖗 Stategic Praning	grand, because the other side of it is, is resources and supply. You know, staff issues and resources. Like, it's every construction site. You can get builders to da anything now because they're much more amenable. The fellows who wouldn't lift up a bag of sweets now will carry bags of plaster up five stories. But the biggest uncertainty is getting paid Reference 4 - 0.20% Coverage And we're trying to maintain relationships and talk to people about fees, rather than sticking it in an envelope, kind of thing, because the bid in the envelope has got ridiculous you know Reference 5 - 0.15% Coverage	
TODETS	ě	there's an oversupply in the market, demand in the market is down near zero, both public and private. Usually, one would subsidise the other	<u>.</u>
👌 RM 159 Items 🏮 Linked Sc	ources: 9 References: 33 😻 Unfiltered		
🛃 start 🥢 🌈 Hotmail	- murphyrcis 📄 Appendices	😟 Strategic Planning in J 📑 Exemplar Case A Nvi	EN 🖮 🤇 📎 🕼 🕬 11:12

Internal Analysis (Repeat Business)

Ele Edit View Go Project Li	iks <u>C</u> ode <u>T</u> ools <u>Wi</u> ndow <u>H</u> elp		
	- A-B/U /- A		
5 m 101 04 10 101 (0 0	D 10 at -		
Code M	-1 m 14	1.04 ST-55.	
Nodes	Look for:	Search In + Tree Nodes Find Now Clear	Options 👻 🗙
🔂 Free Nodes 🔂 Tree Nodes	Tree Nodes	Construction Industry	×
Relationships Seric Alders Seric Folders ≩ Al Nodes	Name Name Name	EnternalsWnferriewsUCompany Ab - 9's references coded [1.57% Coverage] Reference 1 - 0.05% Coverage You see, we work, we work always a year in advance Reference 2 - 0.34% Coverage Before a project is ever going to get near a site, it comes to us. So, at the moment, because of uncertainty about in the public purse, difficulties in both the private sector both in terms of you know, commercial status, but also access to finance, people aren't anticipating and planning for their projects next year Reference 3 - 0.82% Coverage like anything I'm working on today, I've been working on, at least if it started this year, I've been working on it since 12 months ago. I don't have the next 12 months work because of the underlying uncertainty. Then beyond that, the biggest issues would be because of the constinctive business climate, is getting, you know, getting work and future work. Planning for twork. Then, the other works	Sumary [Reference] Teat
Sources Nodes Sets Queries Models Links	ia) தூ Strategic Phonice ia) தூ Strategic Phanning	grand, because the other side of it is, is resources and supply. You know, staff issues and resources. Like, if sever construction site. You can get builders to do anything now because they're much more amenable. The fellows who wouldn't lift up a bag of sweets now will carry bags of plaster up five stories. But he biggest uncertainty is getting paid Reference 4 - 0.20% Coverage And we're trying to maintain relationships and talk to people about fees, rather than sticking it in an erwelope, kind of thing, because the bid in the envelope has got ridiculous you know	
Classifications Folders		Reference 5 - 0 15% Coverage	
1	*	there's an oversupply in the market, demand in the market is down hear zero, both public and private. Usually, one would subsidise the other	<u>×</u>

Internal Analysis (Human Resources)

Ele Ede yew Go Broter, Links Code Tools Window Belo Bew - B D D D D D D D D D D D D D D D D D D	× x
Bew Bew <th>× x</th>	× x
- Brid - Dr Dr - Dr - Dr - Dr	x - x
Image: Second and Second an	s • x
Contrast Manuel Hudies End End End End Diptore Nodes Look for: Search In Tree Nodes Find Now Clear Diptore	× ×
Nodes Look for Search In + Tree Nodes Find Now Clear Option	s • x
Free Nodes	×
Tree Nodes Generation Content of the Nodes Content	~
 Name <li< td=""><td>Summary Reference Text</td></li<>	Summary Reference Text
And I know that sound's really fuddy duddy and old school but we would be more old school. We make sure that no one gets responsibility to do things until they're learned to do it properly.	
Subjects Barbarborn Connegy Deference 4 _ 11 21% Coversing	
Sets Set	
Queries Querie	
Models By Macro Environment medium term	
Unks Gereat Coropary Information Reference 5 - 0.25% Coverage	
Classifications	
Folders Brakegic Choice get them qualified in M&E Engineer. Get them qualified as somebody who understands the technical and financial issues of M&E.	-
	<u> </u>
A HM Tostlens, sources 3 Heferences 51 W Unified State A Homel-muchaniss, A Adventices (1) State Decembin 1, (2) Exemplar Case A Nu., (3) A G	⁰⁰ 11:15

Competitor Analysis (Cursory)

🔁 Strategic Planning in Ir	rish Quantity Surveying Practices.nvp - NVi	vo	
Ele Edit View Go Projec	t Links ⊆ode Iools <u>Wi</u> ndow Help		
New - 😁 🖻 🖬 🖬	No Maria States and No.	ク 9 9 回日 みお 1 手当 M - 1	
	A-B/U 2-2	- 三百百百 (1995) 日本 (1995) 1000) 1000) 日本 (1995) 1000) 1000)	
> III BN 00 10 101 0	0.01		
Code M -			
Nodes	Look for:	Search In 🔻 Tree Nodes Find Now Clear	Options • X
7 Free Nodes	Tree Nodes	Competitive Environment	×
Cares Relationships Matrices Search Folders	Name Name Phase 1 Broad Coding Phase 1 Broad Coding Phase 2 Coding Dn Phase 2 Coding Dn Phase 3 Competitive Position Competitive Position Competitive Position Phase 8 Competitiv	KinternalsUnterviews(Company A2- §8 references coded [246% Coverage] Reference 1 - 0.32% Coverage But for us, I mean the biggest issue for a company this size is, most professional actually I would say of all professional practices we're the flotsam and jetsam. We're, you know, we're price takers, not price makers. And so, the biggest thing is to keep yourself as informed as you can do. Reference 2 - 0.05% Coverage I said that, we're all factors of the market. Reference 3 - 0.16% Coverage But the number of participants in the market has also reduced and even if there's guys, there's firms who are painfully hanging on by their fingernails Reference 4 - 0.41% Coverage Beference 4 - 0.41% Coverage	Survivay [Federation] Text
Sources Nodes Sets Queries Nodels Links Classifications Folders	Contrast Configuration Contrast Contrast Configuration Contrast Contrast Contrast Contrast Contrast Contrast Contrast Contrast Contrast Contrast Contrast Contrast Contrast Contrast Contrast Contrast Contrast Contrast Contrast Contrast Contrast Contrast Contrecontrast Contrast Contrast Contrast Contrast Contrast Contrast Con	But what I've seen is, generally, you wouldn't get a builder in to do your house if he told you thought he was on the verge of collapsing. We're still being favoured with Sware still being favoured with Sware still being the substant is a business relationship and keep working, although it's obviously on a much tighter basis. Reference 5 - 0.38% Coverage The level of competition if you go public, public, public sector work, the level of competition is such that you don't want the work at that price. It really has got, I mean, it's almost like the perfect competition, except I'm a 100% certain that guys are having to take work way below cost. They're buying turnover. Naw, you cannot sustein that. Reference 6 - 0.65% Coverage	*
🙎 RM 159 Items 🏮 Linked	Sources 10 References: 40 🤯 Unlillered		
🚺 Start 🛛 🌔 Hotma	ail - murphyrcis 🔁 Appendices	😟 Strategic Planning in I 🔛 Exemplar Case A Nvi	EN 🖮 🔇 🗞 🚮 🕬 11:21

Competitor Analysis (contd.)

Strategic Planning in Irish	Quantity Surveying Practices.nvp - NViv	o de la companya de l	E 🕫 🔀
Ele Edit View Go Project Li	inks <u>C</u> ode <u>T</u> ools <u>Wi</u> ndow <u>H</u> elp		
New - 18 2 2 4 10	5 m - 2 = - = - 1 + - 1 + 1	2 2 2 2 2 2 2 7 4 W	
- Anal	-10 A-B Z U Z-24-	三百三里 道徳田田 林山 3 か びせん。	
D D DH W N NJ (C)	02.50		
Code M Harne (+)	- In Is Free No		
Nodes	Look for	Search In 🔻 Tree Nodes Find Now Clear	Options 🔹 🗙
Free Nodes	Tree Nodes	Photormal Competitor Analysis	×
Cares Cares Matices Seach Folders	Nome Prove Prove 2 Coding Un Prove 2 Coding Un Prove 2 Coding Un Prove 2 Coding Un Prove 2 Competitive Environment Proventive Prostion Proventive Rescion Proventive Rescin Proventive Rescin Proventive Rescin Proventive Res	Anternals/Interviews/Company, A2 = § 10 references coded [3,38% Coverage] Reference 1 - 0.52% Coverage is "a reing tide lifts all boats". So and there's a lot of organisations in the construction industry fail to understand this. Revenues are lifting and lifting, and everybody to a degree took their eye of the costs and the structure, you know, they might have adapted the structure for a set of revenues that waan't sustainable, which you got away with it on the up, because the jobs just kept coming. We would have been guilty of it but to a much lesser degree than most Reference 2 - 0.63% Coverage at the peak there was no shortage of fellas going around. There was no shortage of work. But what is certainly seen at the peak is the quality of work has suffared. I mean, we would ease, don't get me wrong, 'm not trying to sound holier than thou, I ve seen blins of guantities and the like from other, big practices in town, very, very poor. But, the margins were being out to get work. Particularly with the prakte developers. They would cut you to the bone but there was a value of work there to sustain it. Now, there was a lot of bodies around but there was a value of work.	Sunnary Reference Text
Sources		Reference 3 - 0.51 % Coverage	
Nodes	Macro Environment	I here are a number of tirms who would have been similar size to ourselves 10 odd years ago, are multiples of ourselves today. So, they have different cost structures, they'd have different overhead	
Duariar	General Company Information General Company Information	obligations you know, they'd have different liabilities. I mean, you can reduce the staff numbers to	
Models	E 😥 Objectives	a lot of professional practices as well as contractors are carrying debt.	
Links	Strategic Choice	Reference 4 - 0.07% Coverage	
Classifications	Strategic Planning		
Folders		Do you actively seek information?	
	*	Director A More intuitively	*
A RM 159 Items Sources: 8 Refer	rences: 44 👿 Unfiltered		
🛃 start 🦳 🏉 Hotmail - n	nurphyrois 📴 Appendices	🕲 Strategic Planning in J 🔄 💆 Exemplar Case & Nvi	EN 🛲 🔦 🗞 🚮 💵 11:22

Corporate Strategy (Stability/Survival)

Strategic Planning in Irish	Quantity Surveying Practices.nvp - NVi	vo	
Ele Edit View Go Project Li	inks <u>C</u> ode <u>T</u> ools <u>Wi</u> ndow <u>H</u> elp		
🛛 New 📲 🕾 🖻 🗸 🖷	18 M - 21 E + E - F - 18 + 16 +	🧷 🔊 🕲 🔲 🖪 🗛 🕹 🔫 👘 👘 👘 👘	
1 1	A-B / U Z-2	- 三百百里 (1995) E A (1995) 199 (1995) 10 (1995)	
	10. 61 42 - 7		
Calculat 1		A CONTRACTOR OF A REPORT OF A	
Nodes	Look for:	Search In 🔻 Tree Nodes Find Now Clear	Options 👻 🗙
Tree Nodes	Tree Nodes	Corporate Level Strategy	×
C Cares Careloicombps Matrices Search Folders C All Nodes	Name Name Phase 1 Broad Coding Phase 2 Coding On Phase 2 Coding On Phase 2 Coding On Phase 3 Broad Disposis Phase 1 Bro	KinternalsUnterviews/Company A2 = § 18 references coded [5:32% Coverage] Reference 1 - 0.37% Coverage We made a conscious decision, even at the height of the thing that 16 people was our upper limit. That we didn't want to ex actually, this office can accommodate 20 people, but we never envisaged going near that because we with Director X and myself, even managing it at 14 people, you needed a whole new layer of management and we weren't set. Reference 2 - 0.27% Coverage without ever consciously getting it, we're a small to medium size enterprise. There's a niche for us there, we're consciously getting it, we're a small to medium size enterprise. There's a niche for us there, we're consciously getting it, we're a small to medium size enterprise. There's a niche for us there, we're consciously getting it, we're a small to medium size enterprise. There's a niche for us there, we're consciously getting it, we're a small to medium size enterprise. There's a niche for us there, we're in the simple overriding strategy at the moment it is survival. And that really is it. And I think in beingf. we can rule this out, still provide quality of service, that the people that we're	Summery Relevence) Text
Sources		dealing with, if we're all still standing at the end, then we'll all move on to better times	
Nodes		Reference 4 - 0.39% Coverage	
Sets		your revenue, you can adjust your fees and you can offer value and be seen to be competitive,	
Queries		without slitting your throat, so your clients appreciate that your trying to reflect the market and they'll still do business with you. But the one area where you'll still do business and spend time is on your	
Models		costs, because they're the only places you can truly control.	
Links		Reference 5 - 0.33% Coverage	
Classifications	_	Complete data a balance in the first of the first of the second second of the second sec	
Folders	2	talk to him and sayfor tarker be dealing with you when we all come out of the fast, him isster). Unlocation of a little to the side and we'll all do αcood business together. Then and in the future. Than make it like more difficult for people.	
🙎 RM 159 Items 🎒 Linked Sour	rces: 11 References: 77 👿 Unliftered		
🛃 start 👘 🏉 Hotmail - m	rurphyrais 🚞 Appendices	🔞 Strategic Planning in I 🔯 Exemplar Case A Nvi	EN 🖮 🔦 🗞 🕼 🔊 11/23

Business Level Strategy (Differentiation)

10 Strategic Planning in Iris	sh Quantity Surveying Practices.nvp - NVi	vo	E 8 🛛			
Ele Edit Yew Go Project	Links <u>C</u> ode Tools <u>Window</u> Help	and the second is body as a second				
New • 😸 🖸 🖬 🛛	o o n - Carlon Maria	🧿 🕲 🔲 🗉 🖓 🖓 🕹 🖬 🖞 🔚 👖				
	<u> </u>	- 臣吾君臣 遠律田臣 Ag うか びせ ね。				
	D2 50 act - 1					
E Code AF		·····································				
Nodes	Look for.	Search In 👻 Tree Nodes Find Now Clear	Options 🔹 🗙			
Tree Nodes	Tree Nodes	Differentiation				
Cases Relationships	Name Phase 1 Broad Coding Phase 2 Coding On	<pre><internalsunterviews\company_a>= § 5 references coded [2.60% Coverage]</internalsunterviews\company_a></pre>	Summ			
E Search Folders	Analysis and Diagnosis	Reference 1 - U.84 % Coverage	ary			
🛐 All Nodes	 So General Computing So General Computing So Inglementation So Objectives So Objectives So Stategic Choice So Business Level Stategy So Cott Leadership 	There's certain projects have a certain client and you might see an opportunity. We have particular clients who we take a very long view at. We've done, there's one, that paper that was brought in that I signed therethat's for a client for whom we done 4 substantial projects. He has an interest in two huge projects. Now, the y're on hold at the moment in the current climate but we would have a long term understanding with him. We give him a quality of service, our fee is very, very, very, corr, corrompetitive, but if he has any quantity surveying, he comes to us. We make sure he is very well taken care of and that approach would mirror to a number of the public bodies and housing trusts clients we always deal with. You're maintaining relationships, you're building on it.	Federance Text			
	Differentiation Defension Defen	Reference 2 - 0.20% Coverage if we can ride this out, still provide quality of service, that the people that we're dealing with, if we're all still standing at the end, then we'll all move on to better times, you know				
	🛞 🤬 Strategic Variation	Reference 3 - 0.51 % Coverage				
Sources	🕀 😥 SWOT	I can hide my time. They've enand hune money on this site, it's the old thing _ survival, role yourself				
Nodes	🗑 🥪 Strategic Planning	into a ball, get out the far side, hope that they get out the far side and then, in a better market, that				
Sets		development will progress. They will remember the fact that, they'll be mindful of the fact that you were patient with them and we've been asked to be patient, when things were tight and that you'll still				
Queries		be in the game. So, we've a couple of them like that				
Models		Reference 4 - 0.19% Coverage				
Links		like we're here since 1948 our core market is here. We're hest nlaced to do what it is we do here				
Classifications		and offer quality of, you know, to deliver the product we've always lived here				
Folders	2	Reference 5 - 0.86% Coverage	-			
ARM 159 Items Sources: 9 R	eferences: 39 😿 Unfiltered		4			
🔧 start 👘 🏉 Hotmail	l-murphyrcis 🚞 Appendices	🕲 Strategic Planning in I 🧃 Exemplar Gase A Nvi	EN 🛲 🔇 🗞 🛱 🖓 11:25			

Strategic Variation (International Market)



Approach to Strategic Planning (Emergent)



Strategic Planning Process Formality (Informal)



Participation in Process (Director-driven)

Strategic Plannin	ig in Irish Quantity Surveying Practices.nvp - NViv		
File Edit View Go	Project Links Code Taols Window Help		
New - H B	Baaan g≓r≣r⊟r⊠r Nrd	199 UE 44 174 W	
+ Ala	- // ■ <u>A</u> B / U <u>∠</u> <u>A</u>	■ 革 事 書 ほ ほ 日 日 A () 3 7 (5 7 8)	
◎ □ Mi 4i No	- CP 06 10 - 00 00		
Hole At Name	 Industry Environment 0¹ Tree Nor 	des (Priese 2 Control Onliver 🔹 🗃 🚎 👘 🏝 🖷 🗮 👘 🦉	
Vodes	Look for: Search In	Tree Nodes Find Now Clear	Options - X
Free Nodes	Tree Nodes	Participation	
Cases	Name	<pre><internals\interviews\company a="">- § 1 reference coded [0.51% Coverage]</internals\interviews\company></pre>	N 9
Matrices	🚊 🥁 Phase 2 Coding On	Reference 1 - 0.51% Coverage	
Search Folders	Analysis and Diagnosis Analysis and Diagnosis Analysis and Diagnosis	Do they get involved in terms of setting objectives or is there a form for staff to suggest things they	
	P Implementation	may think themselves to yourself?	
	E P Objectives	Director A No heretafers there have there but noticularly gines Director Y, Juppeld make a point of sitting	
	Strategic Choice Strategic Planning	them down once a month, everybody, and just communicating. This is where we are, this is where	
	Approachs to Strategic Planning	we want to be, this is what we need to do, this is what we need to be carefuland these are the threats and opportunities.	
	🗄 🚱 Comprehensiveness of strategic planning pr	distantial all interviewed Company (D. Transmitter) & E. rafarannea coded (2) 579(Converse)	
	Flow Engality of Process	sinternalsonterviews.company o manscripts - 9 5 references coded (2.5% % coverage)	
	P Integration	Reference 1 - U.96 % Coverage	
	🗄 🤗 Intensity	The 2 founding directors manage the company from an overall view but the 6 new directors manage to the optimized of the sectors manage to the sectors mana	
ources	Participation	strategy and overall running of the company. They wouldn't be managing projects or managing	
odes	Director participation only	cashflow, finance, future clients, like that. The other directors are managing the projects and teams.	
ets	Director plus senior staff participation	Reference 2 - 0.64% Coverage	
ueries	Planning Tools	You mentioned an external consultant that has provided services, is that in terms of providing	
iodels		consultancy services regarding market trends and the economy or is that consultancy services as regards putting together your strategic plan?	
inks		Director D' Both	
lassifications			
olders		Reference 3 - U.75% Ceverage	
*		All the directors would all have a hand, senior staff, there's probably 11 people who have been here	~
RM 159 Items 🧕	Linked Sources: 6 References: 25 😻 Unfiltered		
🔧 start 🔰 🖌	🖻 Hotmail – murphyros 🛛 😇 Appendices	🚯 Strategic Planning in I 📓 Exemplar Case A IVV	EN 📖 🔦 👧 🔊 11/4

Following the coding of interview transcripts, each company (case) was categorised based on a number of attributes providing the opportunity for cross-case analysis as well as queries based on a chosen attribute.

🚯 Strategic Planni	ng in Irish Quantity Su	aveying Practices m	rp = NViva				- 🗗 🔀
Eile Edit Yiew Go	Project Links Code I	ools Window Help					
New - 🗑 🖹	0200-		10-370 DE 22	1字画 加・1	a _		
-	-	ABIU		E M . 39 39	10 L		
D D DN 44 00	M M	DA DO 45-7-			1		
Lode A	+	+]	-		-		
Nodes	Cases				-		
Free Nodes	Name	Sources	References	Created On	Created By	Modified On	Modified By
7 Tree Nodes	🕒 Company Á	3		22/06/2009 11-24	DM	17/07/2010 16:32	BM
Cases	Company B	2	Case Properties		2 🔀	17/07/2010 16:32	BM
😡 Relationships	Company C	2				24/07/2010 12:27	BM
Matrices	Company D	2	General Attribute Values			11/10/2010 21:01	BM
🖻 岁 Search Folders	Company E	2	Attribute	Value	~	17/07/2010 16:33	BM
All Nodes	Company F	2	Approach to Strategic Planning	Emergent Strategy	~	17/07/2010 16:33	BM
1.1.1.1	🕘 Company G	2	Business Level Strategy	Differentiation		17/07/2010 16:33	BM
	Company H	2	Corporate Level Strategy - Past	Stabilitu		12/10/2010 13:42	BM
	Company I	2	Corporate evel Strategu - Present	Combination		17/07/2010 16:34	BM
	Company J	2	ISO Cottified	ISO partitied		12/10/2010 13:55	RM
	Company K	2	Lish with UK assesses	Mat folia dite UK assesso		17/07/2010 16:34	BM
			Link with OK company	Not inked to DK company	×		
			Number of Employees - July 2009	6-10	*		
			Number of Employees - Peak	11-20	×		
			Participation in Strategic Planning Proc	Senior Management only	~		
			Risk Attitude	Risk Averse	×		
			Francessian Internet	I nadata anti nata e atas			
Sources				Apply 01	Cancel		
Nodes							
Sets							
Queries							
Models							
Links							
Classifications							
Folders							
÷							
A RM 11 Items							
1 start	Hotmail - murphyrois	C Appendices	Strategic Planning in I	🖞 Exemplar Case A Nvi 🦉	Nvivo 8 Coding Strat		EN - C & R P 12:25

An example of Company A attributes are displayed in the screenshots below:

New + H A	B 2 5 1 1 1			1 P 4 ht -			
		A-B Z U					
the state and the	AND 400 1				8		
In This die Ab	MAT OF	No MI NZ .					
Code of	*	The second	n 👻 .				
lodes	Cases						
Free Nodes	Name	Sources	References	Created On	Created By	Modified On	Modified By
Tree Nodes	Company A	3	a contraction of the second second	22/06/2009 11-24		17/07/2010 16:32	RM
Cases	Company B	2	Case Properties			17/07/2010 16:32	BM
Helationships	Company C	2	C Attribute Values			24/07/2010 12:27	HM
Matrices Search Folders	Company D	2	General Autobule Taules		T	11/10/2010 21:01	HM
All Nodes	Lompany E	2	Attribute	Value		17/07/2010 16:33	HM
- A HOUDS	Company F	2	Bisk Attitude	Risk Averse	~	17/07/2010 16:33	HM
	Company G	2	Sectors Serviced - July 2009	Predominantly Public Sector	~	17/07/2010 16:33	RM DM
	Company H	2	Sectors Serviced - Peak	Predominantly Public Sector	~	12/10/2010 13.42	DM
	Company I	2	Services Provided	Core QS, PM plus bank monitoring	~	12/10/2010 13:55	PM PM
	Company K	2	Strategic Plan Process Formality	Informal	~	17/07/2010 16:34	BM
	O company is	-	Strategic Planning Comprehensiveness	Cursory	~	111011201010.01	
			Strategist	Defender	× -		
			Written strategic plan	No	~		
			Year of Establishment	1940's	~		
				4	~		
ources			-	Anely OK	Cancel		
des							
ts							
ieries	17						
odels							
nks							
assifications							
Iders							
***	1						
•	11						
nim 11 items		C					